WAIKATO Research Commons

http://researchcommons.waikato.ac.nz/

Research Commons at the University of Waikato

Copyright Statement:

Te Whare Wānanga o Waikato

The digital copy of this thesis is protected by the Copyright Act 1994 (New Zealand).

The thesis may be consulted by you, provided you comply with the provisions of the Act and the following conditions of use:

- Any use you make of these documents or images must be for research or private study purposes only, and you may not make them available to any other person.
- Authors control the copyright of their thesis. You will recognise the author's right to be identified as the author of the thesis, and due acknowledgement will be made to the author where appropriate.
- You will obtain the author's permission before publishing any material from the thesis.

Sustainable livelihood development: A case study measuring social capital and its role in sustainable livelihood development in Zoucheng, China

A thesis

submitted in partial fulfillment

of the requirements for the degree

of

Doctor of Philosophy in [Economics]

at

The University of Waikato

by

Shaoyong Liu





2017

ACKNOWLEDGEMENTS

In keeping with the topic of this thesis, many social networks helped and supported me to complete my "livelihood choice." Without their generous support, this thesis would not have been completed. I would like to acknowledge the support and trust of my three supervisors: Dr Steven Lim, Dr Michael Cameron, and Dr Jenny Cave. Dr Steven Lim, my chief supervisor, provided invaluable help in many areas, including the structure of the thesis, the research framework, relevant analytical tools, simulation model development, and key literature.

During the thesis writing, I experienced unanticipated family problems. However, I received emotional support, understanding, patience, and encouragement from Dr Lim and my other two supervisors. Words cannot express my thanks. To Dr Michael Cameron, a big thank you for your support with corrections and suggestions on my research design, data analysis, and writing style. All of these suggestions considerably improved the quality of the research, the analytical results, and the final form of this thesis. To Dr Jenny Cave, the other member of my panel of supervisors, many thanks for your helpful and valuable suggestions concerning tourism, cultural values, social networks, and writing.

I would like to thank the Department of Economics for office space and other research facilities during my PhD course, with special thanks to Ms Maria Fitzgerald and Mrs Leonie Pope who organized the provision of the relevant facilities. Thanks also to my colleagues in the department, especially Tran Quang Tuyen, who discussed and shared valuable research questionnaires with me.

Very special thanks to Dr Dan Marsh and Professor Mark Holmes – without their unfailing encouragement this thesis would not have been possible. I also thank Judy McDonald for her thorough English correction.

I am grateful to many people and organizations that helped me to conduct my survey in China. Many thanks to all those who kindly agreed to participate in my survey. To many friends, some of whom have grown up with me or studied with me, your support has been invaluable. I also thank the staff of relevant industries in local mining in Zoucheng, the local Statistical Bureau and Tourism Bureau, and the Zoucheng Government for their support and openness to discussions and provision of data, permission to conduct the survey and provision of much other relevant information.

To my family, I give my heartfelt thanks to my parents, especially my mother, who has taken care of the whole family during this time. Without their support and encouragement during the research and study periods, I would never have completed this thesis. To my cousins, thank you for providing valuable suggestions to support this study. Finally, I thank my wife for her forbearance and invaluable sacrifices.

ABSTRACT

This thesis is to discover whether the strength of social capital can alter household livelihoods and their sustainability. The intention of this thesis is to answer two questions: (1) Can different types of social ties¹ (kin and non-kin ties) lead to differential access to social capital in a small community in China? (2) Does the strength of social capital alter local livelihood patterns by diversifying household livelihoods, or producing new livelihoods in an industry that has potential for development?

This thesis, using a quantitative methodology, focuses on social capital measures. It begins with data collection on the social capital status and livelihoods of mining households. The household survey used a random recruitment for sample selection, which involved 228 participants in mines and related working departments (e.g. electricity, and the corporate headquarters of the local biggest mining company) in Zoucheng city, Shandong Province, China. OLS regression models used the household survey to generate associations of variables, such as human capital resources for respondents, length and location of residency, access to social networks, etc. with the composite measure of access to social capital. With these analyses and background information about Zoucheng, I use a simulation model to project the alternative livelihoods for local rural households based on local resources (e.g. farmland, tourism, and mineral resources). The aim of the simulation model was to better understand the role that social capital might play in possible transitions to alternative livelihoods (namely, tourism and farming).

The research results argue that both kin ties and weak ties (such as friendships) are important factors associated with social capital for mining sector residents in Zoucheng (Chapter Six). The advantages of strengthened social capital are presented in the simulation model (Chapter Eight). As social capital was enhanced, both incomes and number of rural tourism operators increased. These results have implications for governmental policy regarding livelihood strategy diversification and poverty alleviation. An understanding of these concepts may play a role in assisting emerging economies to be resilient at local levels and better equipped to withstand exogenous shocks in certain cultural contexts (e.g. Confucian society).

¹ In general, social capital refers to resources embedded in networks of relationships, trust, and social norms that will bring reciprocal benefits to those involved (Sobel, 2002; Lin, 1999; Coleman, 1990; Bourdieu, 1986; Putnam, 2000). Social ties are individual actors' relationships in hierarchical structures and other networks, which provide potential access to other actors in the social networks (Lin, 2001:2004). These network resources embedded in individual actors potentially become people's social capital (Lin, 2001:2004).

TABLE OF CONTENTS

ACKNOWLEDGEMENTS	ii
ABSTRACT	. iii
CHAPTER ONE: INTRODUCTION	1
1.1 Introduction	1
1.2 Statement of the problem	2
1.3 Review of the literature	5
1.4 Literature gaps	7
1.5 Research objectives and questions	8
1.6 Outline of the thesis	.10
CHAPTER TWO: Research Context	.13
2.1 Description of the study area	.13
2.2 Distribution of livelihood alternatives: mining, tourism, and agriculture	.15
2.2.1 Mining	10
2.2.1.1 Resource dependence	19
2.2.1.2 Economic growthing to industrial output value	20
2.2.1.4 Industry structure	. 21
2.2.1.5 Residents	. 23
2.2.2 Tourism	. 24
2.2.2.1 Tourism resources	. 25
a) Diverse categories	. 25
b) Potential tourist attractions	. 26
c) Geographical position	. 29
d) Market demand	. 30
2.2.2.2 Characteristics of tourism operations	. 31
a) Government role and private business	. 31
b) Lack of destination identity	. 35
2.2.3 Agriculture	. 37
a) Rural population & labourers' demand for work	. 37
b) Owning & using farmland	. 39
c) Market demand	41
d) Eating habits	. 42
2.2.4 Comparing livelihood alternatives	. 43
a) Opportunities for rural tourism in Zoucheng city	. 44
b) Rural tourism operators	. 45
2.2.5 Concluding summary	. 47

CHAPTER THREE: CONCEPTUAL AND THEORETICAL FRAMEWORK	.49
3.1 Introduction	.49
3.2 Sustainable development theory as related to livelihood3.2.1 Vulnerability context	.52 .54
3.2.1.1 Natural resource dependency-issues for mining activities3.2.2 Institutional influences, transforming structures and processes	. 54 . 56
3.2.3 Livelihood strategies and outcomes3.2.3.1 Over-reliance one particular economic activity versus diverse livelihood strategies.3.2.3.2 Alternative livelihood strategy	. 57 . 60 . 61
 3.3 Social capital	.66 .66 .68
3.3.4 Some economics of social capital	. 71
3.4 Role of social capital3.4.1 Role of social capital in the sustainable livelihood framework3.4.2 Applications of social capital: social networks and social resources	.72 .72 .74
CHAPTER FOUR: RESEARCH METHODS	.77
4.1 Research Design	.77
 4.2 Data collection for household survey 4.2.1 Sampling frame	.78 .78 .79
 4.2.3 Questionnaire design	. 80 . 80 . 80
 b) Selected measures of social capital. 4.2.3.2 Other information	. 83 . 85 . 85
 4.3 Analytical framework design 4.3.1 Issues, risks and shocks relevant to the local community 4.3.2 Measures of social capital 4.3.3 Simulation model and analysis 	. 86 . 88 . 88 . 89
4.4 Translation, data entry and consistency checks	.90
4.5 Calculation of composite social capital value	.90
4.6 Definitions and summary of variables	.93
4.7 Applications to research questions	.94
4.7.1 Regression model	.94

95
00
99
99
03
07
10
10
10
14
17
19
19
20
23
25
27
32
36
38
41
42
42
42
43
44
44
47
47
40
40 49
50
52
53
55

a) Labour allocation between agriculture and tourism	156
b) Tourism sector	157
c) Agricultural sector	160
8.3 The potential impact of increased social capital	165
8.4 Livelihoods development strategies-current livelihoods and alternati	ve livelihoods 172
8.5 Summary	179
CHAPTER NINE: SUGGESTED POLICY IMPLICATIONS AND CO STATEMENT	NCLUDING 185
9.1 Discussion and Implications	
9.2 Conclusion	
9.3 Limitations	
9.4 Possibilities for future research	
REFERENCES	194
Appendices	234
Appendix A: Simulation model evaluation and testing	234
A.1 Structure verification test	234
A.2 Dimensional consistency	234
A.3 Data used for assessment	
A.4 Boundary adequacy test	235
A.5 Sensitivity test	235
Appendix B: LOOKUP	238
Appendix C: Summarized data for simulation model	239
Appendix D: Household survey	240

List of Figures

FIGURE 2.1. THE LOCATION OF ZOUCHENG COUNTY IN SHANDONG FROVINCE, CHINA	14
FIGURE 2.2: MINING AND TOURISM DISTRIBUTION IN ZOUCHENG CITY	15
FIGURE 2.3: COAL PRODUCTIONS (MILLION TONS)	18
FIGURE 2.4: THE CONTRIBUTION OF LOCAL MINING INDUSTRIES TO TOTAL INDUSTRIAL OUTPUT VAL	JE,
2000-2013 (UNIT: %)	21
Figure 2.5: The industrial structure distribution of local GDP in Zoucheng (% GDP)	22
FIGURE 2.6: NET INCOMES PER CAPITA FOR MINING AND NON-MINING AREAS, 1999-2013	24
FIGURE 2.7: THE MAIN CULTURAL AND SCENIC AREAS IN ZOUCHENG	29
FIGURE 2.8: TOURISM DEVELOPMENT IN ZOUCHENG, 2000-2013	31
FIGURE 2.9: THE DISTRIBUTION OF PRIVATE AND PUBLIC OWNERSHIPS IN LOCAL TOURISM	33
FIGURE 2.10: THE PERCENTAGE OF RURAL POPULATION IN ZOUCHENG CITY	38
FIGURE 2.11 ARABLE LAND PER HOUSEHOLD IN ZOUCHENG CITY 2000-2013 (MU/HOUSEHOLD)	41
FIGURE 2.12: RECRUITMENT ADVERTISEMENTS IN A LOCAL RESTAURANT	44
FIGURE 2.13: RURAL TOURISM ACTIVITIES IN ZOUCHENG	47
FIGURE 3.1: THEORETICAL FRAMEWORK FOR ANALYSIS OF ZOUCHENG HOUSEHOLD LIVELIHOODS	51
FIGURE 3.2. ASSETS, SOCIAL CAPITAL, AND NETWORKS	73
FIGURE 4.1: RESEARCH FRAMEWORK	86
FIGURE 5.1: COAL PRODUCTION AND INDUSTRIAL DUST DISCHARGED IN ZOUCHENG FROM 2000 TO 2	2013
	105
FIGURE 5.2: NUMBER OF POLLUTION ISSUES IN THE ZOUCHENG LOCAL ADMINISTRATIVE PETITIO	N
SYSTEM. 2000-2013	106
FIGURE 7.1: ANALYTICAL FRAMEWORK SHOWING THE INTERACTIONS BETWEEN COMPONENTS OF TH	Е
SYSTEM	121
FIGURE 7.2: THE PROPORTION OF CITY MAINTENANCE EXPENDITURE IN TERMS OF LOCAL GDP FROM	I
FIGURE 7.2: THE PROPORTION OF CITY MAINTENANCE EXPENDITURE IN TERMS OF LOCAL GDP FROM 2000 TO 2010	
FIGURE 7.2: THE PROPORTION OF CITY MAINTENANCE EXPENDITURE IN TERMS OF LOCAL GDP FROM 2000 TO 2010 FIGURE 8.1: TOTAL COAL OUTPUT (C) AND FARMLAND QUALITY (Q _F)	
 FIGURE 7.2: THE PROPORTION OF CITY MAINTENANCE EXPENDITURE IN TERMS OF LOCAL GDP FROM 2000 TO 2010 FIGURE 8.1: TOTAL COAL OUTPUT (C) AND FARMLAND QUALITY (Q_F) FIGURE 8.2: INDUSTRIAL DUST DISCHARGED AND THE INITIAL TOURIST NUMBERS 	146 154 155
 FIGURE 7.2: THE PROPORTION OF CITY MAINTENANCE EXPENDITURE IN TERMS OF LOCAL GDP FROM 2000 TO 2010. FIGURE 8.1: TOTAL COAL OUTPUT (C) AND FARMLAND QUALITY (Q_F). FIGURE 8.2: INDUSTRIAL DUST DISCHARGED AND THE INITIAL TOURIST NUMBERS FIGURE 8.3: WORKERS ALLOCATION BETWEEN AGRICULTURE AND TOURISM. 	146 154 155 156
 FIGURE 7.2: THE PROPORTION OF CITY MAINTENANCE EXPENDITURE IN TERMS OF LOCAL GDP FROM 2000 TO 2010. FIGURE 8.1: TOTAL COAL OUTPUT (C) AND FARMLAND QUALITY (Q_F). FIGURE 8.2: INDUSTRIAL DUST DISCHARGED AND THE INITIAL TOURIST NUMBERS FIGURE 8.3: WORKERS ALLOCATION BETWEEN AGRICULTURE AND TOURISM. FIGURE 8.4: NUMBER OF TOURISTS 	146 154 155 156 157
 FIGURE 7.2: THE PROPORTION OF CITY MAINTENANCE EXPENDITURE IN TERMS OF LOCAL GDP FROM 2000 TO 2010. FIGURE 8.1: TOTAL COAL OUTPUT (C) AND FARMLAND QUALITY (Q_F). FIGURE 8.2: INDUSTRIAL DUST DISCHARGED AND THE INITIAL TOURIST NUMBERS	146 154 155 156 157 157
 FIGURE 7.2: THE PROPORTION OF CITY MAINTENANCE EXPENDITURE IN TERMS OF LOCAL GDP FROM 2000 TO 2010. FIGURE 8.1: TOTAL COAL OUTPUT (C) AND FARMLAND QUALITY (Q_F). FIGURE 8.2: INDUSTRIAL DUST DISCHARGED AND THE INITIAL TOURIST NUMBERS	146 154 155 156 157 157 160
 FIGURE 7.2: THE PROPORTION OF CITY MAINTENANCE EXPENDITURE IN TERMS OF LOCAL GDP FROM 2000 TO 2010. FIGURE 8.1: TOTAL COAL OUTPUT (C) AND FARMLAND QUALITY (Q_F). FIGURE 8.2: INDUSTRIAL DUST DISCHARGED AND THE INITIAL TOURIST NUMBERS	146 154 155 156 157 157 160 161
 FIGURE 7.2: THE PROPORTION OF CITY MAINTENANCE EXPENDITURE IN TERMS OF LOCAL GDP FROM 2000 TO 2010. FIGURE 8.1: TOTAL COAL OUTPUT (C) AND FARMLAND QUALITY (Q_F). FIGURE 8.2: INDUSTRIAL DUST DISCHARGED AND THE INITIAL TOURIST NUMBERS	146 154 155 156 157 157 160 161 162
 FIGURE 7.2: THE PROPORTION OF CITY MAINTENANCE EXPENDITURE IN TERMS OF LOCAL GDP FROM 2000 TO 2010. FIGURE 8.1: TOTAL COAL OUTPUT (C) AND FARMLAND QUALITY (Q_F). FIGURE 8.2: INDUSTRIAL DUST DISCHARGED AND THE INITIAL TOURIST NUMBERS	1 146 154 155 156 157 157 160 161 162 162
 FIGURE 7.2: THE PROPORTION OF CITY MAINTENANCE EXPENDITURE IN TERMS OF LOCAL GDP FROM 2000 TO 2010 FIGURE 8.1: TOTAL COAL OUTPUT (C) AND FARMLAND QUALITY (Q_F)	1 146 154 155 156 157 160 161 162 162 163
 FIGURE 7.2: THE PROPORTION OF CITY MAINTENANCE EXPENDITURE IN TERMS OF LOCAL GDP FROM 2000 TO 2010. FIGURE 8.1: TOTAL COAL OUTPUT (C) AND FARMLAND QUALITY (Q_F). FIGURE 8.2: INDUSTRIAL DUST DISCHARGED AND THE INITIAL TOURIST NUMBERS	1 146 154 155 157 157 160 161 162 162 163 163
 FIGURE 7.2: THE PROPORTION OF CITY MAINTENANCE EXPENDITURE IN TERMS OF LOCAL GDP FROM 2000 TO 2010. FIGURE 8.1: TOTAL COAL OUTPUT (C) AND FARMLAND QUALITY (Q_F). FIGURE 8.2: INDUSTRIAL DUST DISCHARGED AND THE INITIAL TOURIST NUMBERS	1 146 154 155 156 157 157 160 161 162 163 163 164
 FIGURE 7.2: THE PROPORTION OF CITY MAINTENANCE EXPENDITURE IN TERMS OF LOCAL GDP FROM 2000 TO 2010 FIGURE 8.1: TOTAL COAL OUTPUT (C) AND FARMLAND QUALITY (QF) FIGURE 8.2: INDUSTRIAL DUST DISCHARGED AND THE INITIAL TOURIST NUMBERS FIGURE 8.3: WORKERS ALLOCATION BETWEEN AGRICULTURE AND TOURIST NUMBERS FIGURE 8.4: NUMBER OF TOURISTS	1 146 154 155 157 157 160 161 162 163 163 164 164
 FIGURE 7.2: THE PROPORTION OF CITY MAINTENANCE EXPENDITURE IN TERMS OF LOCAL GDP FROM 2000 TO 2010. FIGURE 8.1: TOTAL COAL OUTPUT (C) AND FARMLAND QUALITY (Q_F). FIGURE 8.2: INDUSTRIAL DUST DISCHARGED AND THE INITIAL TOURIST NUMBERS	1 146 154 155 157 157 160 161 162 163 163 164 164

FIGURE 8.14: COMPARING NUMBER OF RURAL TOURISM OPERATORS WITH AND WITHOUT SOCIAL	
CAPITAL16	6
FIGURE 8.15: THE DIFFERENT STRENGTHS OF SOCIAL CAPITAL	7
FIGURE 8.16: CHANGE IN EXPECTED TOURISM INCOMES (YUAN/MONTH)16	8
FIGURE 8.17: CHANGE IN NUMBER OF RURAL TOURISM WORKERS	8
FIGURE 8.18: CHANGE IN GROWTH RATE OF TOURIST ARRIVALS	9
FIGURE 8.19: CHANGE IN NUMBER OF TOURIST ARRIVALS	9
FIGURE 8.20: CHANGE IN PRICE OF AGRICULTURAL OUTPUT	0
FIGURE 8.21: CHANGE IN CROP PROFITS PER HOUSEHOLD LABOURER IN FARMING	0
FIGURE 8.22: CHANGE IN NUMBER OF FARMERS	1
FIGURE 8.23: CHANGE IN GROWTH RATE OF FARM PRODUCTS	1
FIGURE 8.24: COMPARING LIVELIHOOD DEVELOPMENT OPTIONS FOR RURAL HOUSEHOLDS	3
FIGURE 8.25: AGRICULTURAL OUTPUT PER HOUSEHOLD & GROWTH RATE OF FARM PRODUCTS (PIT	
CLOSURES)	5
FIGURE 8.26: THE PRICE OF AGRICULTURAL OUTPUT (PIT CLOSURES)	5
FIGURE 8.27: CROP PROFITS PER HOUSEHOLD (PIT CLOSURES)	6
FIGURE 8.28: THE GROWTH RATE OF TOURIST ARRIVALS (PIT CLOSURES)	7
FIGURE 8.29: NUMBER OF TOURIST ARRIVALS (PIT CLOSURES)	8
FIGURE 8.30: EXPECTED TOURISM INCOME (PIT CLOSURES)	8
FIGURE 8.31: THE EFFECT OF GOVERNMENT INVESTMENT ON TOURIST GROWTH	9
FIGURE A.1: SENSITIVITY ANALYSIS ON SIMULATION MODEL	7
FIGURE B.1: COAL MINING POLLUTION ON FARMLAND QUALITY (INPUT: TOTAL COAL OUTPUT)	8
FIGURE B.2: TOTAL COAL OUTPUT AND INDUSTRIAL DUST DISCHARGED (INPUT: TOTAL COAL OUTPUT)	
	8

List of Tables

TABLE 2.1: THE STRUCTURE OF LOCAL TOURISM RESOURCES IN ZOUCHENG 26
TABLE 2.2: NATURE OF OWNERSHIP (PUBLICLY OWNED) OF THE MAIN TOURIST ATTRACTIONS
TABLE 2.3: COMPARISON BETWEEN ZOUCHENG TOURISM RESOURCES AND TOURISM RESOURCES IN
NEARBY CITIES
TABLE 4.1: PARTICIPANT DISTRIBUTION AMONG LOCAL RELEVANT MINING COMPANIES 79
TABLE 4.2: PRESTIGE RATING FOR JOB POSITIONS IN CHINA 84
TABLE 4.3: FACTOR STRUCTURES OF ACCESS TO SOCIAL CAPITAL 92
TABLE 4.4: DEFINITIONS AND SUMMARY OF VARIABLES FOR SOCIAL CAPITAL MODELS (N=228)
TABLE 5.1: SUBSIDENCE AREAS IN ZOUCHENG CITY 100
TABLE 5.2: CHANGES IN RATE OF LAND USE IN VILLAGE TOWNS FROM 1990 TO 2005 (%)101
TABLE 6.1 COMPARISON OF ACCESS TO SOCIAL CAPITAL BY KIN AND NON-KIN TIES BETWEEN RURAL
AND URBAN AREAS (N=218)113
TABLE 6.2: COLLINEARITY DIAGNOSTICS FOR VARIABLES USED IN THE SOCIAL CAPITAL MODEL (OLS)
TABLE 6.3: FACTORS ASSOCIATED WITH ACCESS TO SOCIAL CAPITAL (PARTIAL REGRESSION
COEFFICIENTS, N=191)116
TABLE 8.1: SUMMARY OF THE SIMULATED RESULTS OF RELEVANT VARIABLES FOR EACH SCENARIO 181
TABLE C.1: COAL PRODUCTION AND EMPLOYMENT 239
TABLE C.2: GRAIN YIELD (2004-2013)
TABLE C.3: NUMBER OF TOURISTS IN MENG TEMPLE AND YI MOUNTAIN FROM 2003 TO 2010

CHAPTER ONE: INTRODUCTION

1.1 Introduction

In general, social capital refers to resources embedded in people's connections or group memberships that will bring reciprocal benefits to those involved (Sobel, 2002; Lin, 1999; Coleman, 1990; Bourdieu 1986). Although the concept of social capital was originally constructed in the context of western societies, it has recently drawn the interest of researchers from different cultural backgrounds, particularly as it relates to Chinese culture. Traditional Chinese culture emphasizes the importance of interpersonal relationships and personal networks. Influenced by these traditional cultural values, Chinese society is normally recognized as an exchange-based economy that possesses high social capital, characterized by 'guanxi'. 'Guanxi' is defined as "a dyadic, particular and sentimental tie that has the potential of facilitating favor exchanges between the parties connected by the tie" (Bian, 2006, p.312), which has played an extremely important role in social and economic life in Chinese society (Lu et al., 2013). Therefore, social capital that constitutes guanxi is pervasive in economic and non-economic life in Chinese society. These features are a concrete manifestation of social capital in China. With the rapid development of the economy and improvements in living standards, it is particularly useful to analyze the role of social capital in Chinese society.

Over the years, social capital has been shown to act as an important bridge between academic disciplines in the development of economic theory (Woolcock & Naryan, 2000, Cordes et al, 2003, Yen, Barnes, & Wang, 2011). Furthermore, social capital has an important role to play in government, regarding the investment of resources into poverty alleviation or income generation. With a clearer sense of livelihood assets such as social capital, policy-makers can identify the most useful sorts of public investment for different people in different places (Bebbington, 1999). The focus of this thesis is on the role of social capital during the transition to sustainable livelihoods in a small city (i.e., a county-level $city^2$) in China. It may be generalizable to at least some other counties that have similar features to Zoucheng, China.

1.2 Statement of the problem

Natural resources are normally considered the bedrock for most developing economies (Sovacool, 2015; Conceicao et al., 2011). However, serious resource depletion can lead to environmental contamination and degradation. The main problem is that there are regions that are heavily natural resource-dependent. This is a problem for future development if those regions fail to develop some alternative sources of economic growth. This issue involves the sustainability of natural resources. One important criterion for sustainability of natural resources emphasizes that the rate of consumption is equal to or less than the rate of recovery, which could be applied to both renewable and non-renewable resources (Asafu-Adjaye, 2005; Richards, 2006). Non-sustainable development of natural resources can generate environmental degradation, resource depletion, increasing income disparity, poverty, and marginalization (Raskin, 2000). Examples of natural resource-dependent and relevant issues can be found in many literature sources (e.g. Li et al., 2015; Ferrol-Schulte et al., 2015; Habibah et al., 2010; Nguyen, 2015; Ruffini, 2006; Bury, 2004; Amijaya & Littke, 2005; Pegg, 2006). Therefore, heavy natural resource-dependency is an issue of concern to the internationally.

This thesis uses Zoucheng County as a case study, because this city is heavily reliant on mining. Although mining companies contribute positively to the social and economical components of Zoucheng's development by generating employment and

² In this thesis, the concept of a small city is only in terms of its administrative system (i.e. county-level city). The formal administrative hierarchies in mainland China are: 1. Central government; 2. Provincial-level units such as Province, Autonomous Region, and Cities under Central Administration; 3. Prefectural-level units such as Prefecture, Autonomous Prefecture, and City; 4. County-level units such as Small City, County, and Urban District (Chan & Hu, 2003). A county-level city is often considered as an alternative word for 'small city', and the lowest level that central government reaches. In the strictest sense of the word, county-level cities are not "cities" since they contain rural areas (e.g. villages and farmland) many times the size of their urban, built-up areas.

wealth, they still contribute negatively to the local environment and create miningassociated risks for the community. Thus, this section focuses on mining issues in China so as to understand the general background related to mineral resource dependence.

China is one of the world's largest consumers of energy and the largest producer of CO₂ emissions (Li, 2010; Hu & Lee, 2008). Because of the resource characteristics of China, coal has made a great contribution to economic development so that the coal-based energy consumption structure is not likely to change in a short period of time (Ju et al., 2010; Shi & Liao, 2012; Shi, 2015). China consumes about 40% of the world's total coal production and its coal production accounts for about 39% of the worldwide total (Bian, Inyang, Daniels, Otto, & Struthers, 2010). Some experts and institutions estimate that coal is one of the most plentiful energy resources underpinning the economic and social development of the world (World Coal Institute, 2002; The World Bank Group, 1999).

In general, the coal energy production system includes coal mining, preparation or processing and energy generation. Every stage of the coal energy system may have environmental impacts (Bian, et al., 2010). In many European countries, coal is recognized as a dirty source of energy and has been rendered obsolete. For instance, the French government closed all coal mines in 2004 (Bian, et al., 2010), in contrast to the growing coal production and consumption in China. The impact of mining on the environment is becoming an increasingly prominent problem.

Coal mining is also a major source of atmospheric pollution in China. In general, fugitive dust and gases including sulfur dioxide, oxides of nitrogen and methane from coal mining operations cause air pollution (Bian, et al., 2010; Xu, et al., 2000). Wind erosion from coal stockpiles is another source of fugitive dust (Ghose & Majee, 2000).

Spontaneous heating of coal in waste dumps, mine fires, and methane leaking from the coal strata and coal seams release substantial amounts of CO, CO_2 , NO_x , and SO_x (Ghose & Majee, 2000; Bian, et al., 2010; Xu et al., 2000). As particulate matter and

gases are released into the mine air to be eventually discharged into the atmosphere, the environment can be damaged. For example, in China, around 87% of SO₂ and 60% of dusts in the air are from coal combustion (Xu et al., 2000). The huge emissions of SO₂ can affect the pH of rainwater (e.g. acid rain), which has created much economic and social damage (Xu et al., 2000). Ye et al., (2001) indicated that burnt coal residues can increase the pH of the soil. Excessive coal fly ash (a coal combustion residue) incorporation results in heavy metal pollution, high soil pH, lack of microbial activity and high soluble salts, which all degrade the quality of the land and reduce agricultural productivity (Pandey & Singh, 2010; Gupta, et al., 2002; Haynes, 2009; Lyer & Scott, 2001). In terms of human health, long-term living in a poor environment can increase lung cancer mortality (Xu, et al., 2000; Hendryx et al., 2008).

Mining has possibly precipitated more disputes over land use than any other industries (Hilson 2002). Chen (2007) found that construction accounted for 21% of China's total arable land loss in the period from 1986 to 1995 and 20.8% from 1997 to 2003. Construction was defined as "conversion of arable land to urban and rural settlements, industrial and mining sites, communications and agricultural facilities" (Cheng, 2007 p.4). Deng et al. (2006) and Chen (2006) found that farmland loss has accelerated after 2000.

In China, approximately 95% of coal production was from underground mines in 2007 (Bian et al., 2010). However, one important adverse impact of mining on the environment is land subsidence over underground mines. Li et al. (2009) pointed out that the process of mining coal resources gradually breaks the original stress equilibrium in strata, which causes arable land decline in mining areas. In China, the cumulative total of collapsed coal mining areas is more than 700,000 ha, and about 1.86 ha per capita (based on the population in mining areas) in mined-out regions (as cited in Li, Liu, Du, & Chen 2009). On average, mining about 10,000 tons of raw coal will cause around 0.2 hectares of land to subside (Bian et al., 2010).

In addition, mining activity also affects farmland quality and reduces agricultural productivity, which can subsequently influence rural households' incomes. Long et al. (2007) indicated a strong negative relationship between crop yield and industrialization by analysing a city in China. Coal exploitation on arable land can cause land collapse or degradation, and reductions in crop production (He & Su 2002; Bian et al., 2010).

When local economic development is heavily natural resource-dependent (e.g. mining), economic activities can impact adversely on the environment and other sectors. In recent years, the Chinese government has had to consider the development of measures to meet the requirements for sustainable development (Xu, Yao, & Mo, 2011). For instance, the Chinese government announced that China aims to reduce the intensity of carbon per unit of GDP by 40-45% by 2020, based on 2005 levels (Xu, Yao, & Mo, 2011). In addition, the Chinese government is placing greater emphasis on advancing economic restructuring, economic development modes, and encouraging consumption and production methods that help conserve resources (NCNA, 2008). Other countries and regions have also developed ways to deal with natural resource dependence, which are discussed in the next section.

1.3 Review of the literature

In dealing with the depletion of natural resources and environmental degradation, sustainability and economic transition solutions are important. There is a vast academic literature dealing with natural resource dependence. Some studies (e.g. Wang et al., 2013; Ferrol-Schulte et al., 2015) support governmental and institutional approaches in mitigating the underlying causes of vulnerability of communities and in facilitating sustainable livelihood strategies. Some other studies (Li et al., 2015; "Technologies, preferences", 2011; Hentschel et al., 2002) emphasize the role of technology and innovation in energy efficiency improvements to enable the sustainable transition of resource-dependent cities or industries. Some natural resource-dependent activities (e.g. shrimp aquaculture) are highly vulnerable to natural disasters such as floods and cyclones. Paul et al. (2013) suggested increasing the capacity of community

participation to cope with uncertain phenomena in ways that make livelihoods sustainable. In general, the efficient use of natural resources and wider livelihood options are approaches to dealing with natural resource-dependence (Ahmed et al., 2010; Ursache, 2014). Erzurumlu & Erzurumlu (2015) investigated involving diverse stakeholders and local communities to develop more sustainable livelihood strategies.

The effects of natural resource-dependence on households cannot be ignored. Such dependence can generate over-reliance on one particular economic activity and reduce the diversity of livelihoods available to households (Soriano, 2007). Bury (2004) and Allison & Horemans (2006) evaluated how current livelihoods (e.g. mining operations, fishing) altered household access to resources (e.g. human capital, financial capital, social capital). Asset, activity and income diversification are important attributes of rural livelihoods in developing countries, which might reduce the risk of livelihood failure (Barrett, et al., 2001; Allison & Ellis, 2001). Other studies (Horsley et al., 2015; Ferrary, 2003; and Mayoux, 2001) focus on capital assets and their interrelationships in households.

As one of the important capital assets³ (i.e., natural, human, social, financial, and physical capitals) in households, social capital has often attracted the interest of researchers at the community level (Dagupta, 2000; Knack & Keefer, 1997). The economics literature has concentrated on modelling social capital in areas such as economic growth and development (Putman et al. 1993; Rupasingha et al, 2000; Whiteley, 2002), health and well-being (Zhang et al, 2006; Yip et al 2007), labour markets (Rebick 2000; Knight & Yueh 2008), and inter-firm relationships (Batijargal 2007; Peng & Quan, 2009; Gao, Xu, & Yang 2008).

In the household livelihood development area, Nguyen et al. (2015) studied the role of social capital in environmental resource dependence. They suggested that higher levels of social networks affect the choice of livelihood strategies and reduce income

³ Further explanations are presented in Chapter Three.

dependence on the extraction of environmental resources. Habibah et al. (2010) found that social capital supported the sustainability of local ecotourism and produced better livelihoods. Some social network scholars (Lin 2001a & 1999; Lu, et al., 2013; Bian, 1997; Bian & Huang, 2009) echo a call for research attention to be given to the role of social networks in metropolises such as Beijing. Social capital, which is interpreted as the social resources that people draw in pursuit of their livelihood objectives, has received much attention in Chinese society. However, most studies have been conducted in provincial level cities in the Chinese administrative system, which have populations of at least 10 million. Very few studies have specifically examined the relative importance of social capital with respect to household livelihoods in county-level cities in China.

1.4 Literature gaps

This thesis identifies the important role of kin and non-kin ties in terms of their relationship to social capital in a county level city.

As with many Confucian societies in East Asia, the importance of family or kin ties is an historical aspect of Chinese society. Trust is often encouraged in a relatively narrow circle of family and close friends in Confucian tradition (Dalton, et al., 2001). However, China's changing socio-economic conditions and increased interaction with foreign communities and culture may affect these orientations. With respect to population size, different development status and being exposed to foreign cultures, different city sizes may experience different effects on these traditions relating to social capital. In some large metropolitan cities in China, kin ties (i.e., with relatives) limit the growth of social capital (Lin, 2001; 2004), while in county-level cities in China, the effect of different types of social ties (kin and non-kin ties) on social capital has received less attention. Thus, this study aims to fill gaps in academic knowledge about whether different types of social ties (kin and non-kin ties) can lead to differential access to social capital in small communities in China. As a research target, this study is conducted in the context of small cities in China, which have been under-studied. In 2011, Mainland China had around 2,859 counties and county-level cities (Meng et al., 2012; Fung et al., 2013). The total geographic area of China's counties accounts for about 90% of the total national area (Mi & Ji 1999), while the population of these counties is more than 70% of the total population. Because of this, counties and county-level cities are crucial to the country's economic viability, social cohesion and effective leveraging of social capital. However, in contrast to the situation in large cities, economic development, related education, academic research institutions and the inadequacy of statistical departments in small cities in China have limited researchers' ability to undertake relevant investigations.

1.5 Research objectives and questions

The objectives of this study are to investigate the role of social capital during the transition to alternative livelihoods in a county-level city in China and present a dynamic simulation model of the livelihood outcome changes based on case studies. The intention of this thesis is to answer two questions:

- 1) Do different types of social ties (kin and non-kin ties) lead to differential access to social capital in a small community in China?
- 2) Does the strength of social capital alter local livelihood patterns by diversifying household livelihoods and providing new livelihoods in industries with potential for development?

Investigating the role of social capital in sustainable livelihood development is an important purpose for this thesis. Therefore, understanding local social capital is very important. This motivation generated the first research question. This thesis examines the social capital status of a mining community (the group most involved with the improvement of the local economy) in Zoucheng. The study provides social capital measurements, assesses the strength of relationships (kin and non-kin ties), and provides econometric evidence of factors associated with access to social capital. In doing so it draws out some of the links between social capital growth and relevant

factors such as demographic characteristics. As discussed previously (see Section 1.4), it pays special attention to the strength of relationships (i.e., kin and non-kin ties). The study uses the research results from the first question to adjust the strength of social capital, which is important in the next research step.

The second question is the key question in the whole research process. It includes the issues of current livelihoods, social capital, household strategies, and livelihood outcomes. Without these components, the simulated results would not reflect or achieve the research aim.

This study involves many issues common in China and worldwide, such as a resourcedependent economy and its associated negative effects on the environment and the local community, and investigates sustainable livelihood options. The risks associated with mining activity may become serious problems for future development. Thus, the shocks and risks of mining activity on the local environment and livelihoods provide background information for the simulation model.

The second question also considers the possibilities of alternative livelihoods in different industries. As discussed in Sections 1.2 and 1.3, a transition to an alternative industry is one of the ways to deal with natural resource dependence. Therefore, it is necessary to understand the current development status of alternative industries and their relationships with mining (including the adverse impacts from mining).

This study concentrates on improving the sustainability of household livelihoods, not those of individuals. If mining is the primary employment, diversifying household livelihoods may be achieved for other mining household members who currently receive low rates of pay. It might reduce the vulnerabilities of current household livelihoods, especially with the lack of livelihood choices for miners.

This study uses the sustainable livelihood framework (DFID, 1999), together with social capital investigations and assessment along with local households' and the city's economic background information to develop a theoretical and empirical sustainable

livelihoods simulation model. The data come from quantitative studies. Household survey data were used to analyze social capital status. Secondary data were also gathered to provide insight into local historical contexts that affected household livelihoods. Thus, this thesis provides a useful and innovative approach for studies on the applications of social capital in the process of moving to an alternative livelihood by combining appropriate econometric models and a simulation model. All of the collected data permit the construction of a simulation model to estimate possible outcomes on livelihoods with appropriate consideration given to relevant shocks.

The findings from this study may have implications for government policy regarding social capital implementation, revenue generation, and economic development as well as the personal career development of individuals. Considering the large number of county-level cities in China, this could have significant practical applications. The study also contributes new perspectives on social capital applications in the process of transition from a resource-dependent economy to a sustainable economy in a county-level city in China. The findings may be valuable to other small cities, and rural areas with similar socio-economic characteristics and cultural values (e.g. social networks, family values) in China, and even to other developing countries with similar characteristics in their social networks.

1.6 Outline of the thesis

Research Question (1) demonstrates the framework of empirical relationships between social capital and associated factors. Research Question (2) demonstrates a general research perspective for this thesis. Research Question (1) will be tested in Chapter Six. Research Question (2) will be tested in Chapters Seven and Eight. All of the relevant background information such as general issues and links between different industries are discussed in Chapter Two and Chapter Five.

The remainder of the thesis is organized as follows:

Chapter Two describes the case study area of this thesis (Zoucheng city). It notes some important resources (e.g. mineral resources and tourism) in the city, three industries (mining, tourism, and agriculture), and relevant livelihood alternatives. The aim is to help the reader to understand the background to the analyses and simulation model presented in the following chapters.

Chapter Three builds up the theoretical framework of this thesis. It firstly introduces the sustainable livelihood framework. Social capital is discussed in terms of the purpose of this thesis. The last section introduces the role of social capital in a sustainable livelihood framework, and the applications of social networks. All of these are important theoretical guidelines in establishing the regression model and simulation model.

Chapter Four describes the research methods used to investigate social capital and other questions listed in Section 1.5.

Chapter Five discusses possible issues, influences, shocks and risks in the local community with the growth of mining activities in Zoucheng city. The main livelihood issue is over-reliance on mining activities. Residents of Zoucheng city will have to use local resources to find an alternative livelihood in the future.

Chapter Six presents the strength of social ties in the local city. It also provides the econometric results exploring the factors associated with social capital scores.

Chapters Seven and Eight build up a model with similar characteristics to the current livelihood system, which mainly involves the influences of mining activities on two industries – tourism and agriculture – so as to simulate the current issues in the local community. Then, social capital as applied to livelihood development is included, to indicate the role of social capital in the development of alternative livelihoods (i.e., rural tourism).

Chapter Nine concludes the thesis, discusses relevant policy implications and study limitations, and makes some recommendations on the basis of this research.

In general, Chapters Two and Five are based on background information and issues related to mining activities. Chapters Three and Four provide the theoretical framework and methods for research. Chapter Six analyses the social capital survey results. Chapters Seven and Eight focus on model establishment and analysis.

CHAPTER TWO: Research Context

2.1 Description of the study area

The case study area refers to the whole of Zoucheng city, located in the south-west of Shandong province in China (Figure 2.1). It covers 1,613 square kilometers with a population of about one million, and includes about 350,000 households. About 64 percent (746,395 of 1.16 million) of the total population is registered as agricultural households (Zoucheng Statistical Yearbook, 2013).

In October 1992, Zoucheng became a county-level city. For the national top one hundred counties' ranking, Zoucheng ranked 25th in the 2012 order of economies and its highest place was 16th in 2006 (SURPDI, 2012). In 2013, the output of the county was 73.16 billion yuan (about 20.6 billion US\$⁴), of which about 58% was contributed by secondary industry. The average per capita annual income was 27,605 Yuan (about US\$ 7,776) in urban areas and 12,576 Yuan (about US\$ 3,543) in rural areas (Zoucheng Statistical Yearbook, 2013).

⁴ According to the World Bank, the PPP (Purchasing Power Parity) conversion factor for GDP is 3.55 by following the number of units of China's currency required to buy the same amounts of goods and services in the domestic market as a U.S. dollar would buy in the United States.



Figure 2.1: The location of Zoucheng County in Shandong Province, China

At present, coal mining, water, electricity and gas supply, equipment manufacturing, agricultural processing industries, textile industries, drink manufacturing, non-metal mineral product industries, and raw chemical materials and chemical products production represent about 88.6% of the local secondary industry (Zoucheng Statistical Yearbook, 2013). Coal mining in Zoucheng has experienced rapid growth since the early 1980s and has dominated the local economy in recent years. In the local employment market, the mining industry directly or indirectly provides more than 100,000 job opportunities (Zoucheng Statistical Yearbook, 2013). Local mining industries and connected local companies such as commercial institutions and hospitals, etc., employ more than 70% of the working population in Zoucheng (SURPDI, 2012). Thus, miners' networks (e.g. miners' friends, household members etc.) have significant influences and connections in local society. For this reason, people working in mining or related industries were chosen as the research group in Zoucheng city.

In contrast, tourism occupies a disadvantaged position in the local economy. In 2013, the total tourism revenue was about 3.3 billion yuan, which was lower than that of most other local industries (Zoucheng Statistical Yearbook, 2013). Although the rich tourism resources present potential growth capacity, local tourism development has been slow.

On average, the annual proportion of direct income from local tourism was less than 6% of tertiary industry added value from 2000-2013 (Zoucheng Statistical Yearbook, 2000-2013).

Mineral resources, agriculture and tourism are three important industries, which could support the local economy. However, the focus here is on tourism as a potential resource to support sustainable local economic development.

2.2 Distribution of livelihood alternatives: mining, tourism, and agriculture

Zoucheng city is separated administratively into 17 districts as shown in Figure 2.2 (in which three village towns are recognized as urban areas). The shaded area is urban area and the rest is rural. Figure 2.2 illustrates the distribution of the main mineral resources and main tourist attractions. The local mineral resources have a narrow distribution in the western area, while tourism resources are widely dispersed compared to mining in Zoucheng city.





The following information describes the basic background, the process of local economic development in Zoucheng, the living status of residents and tourism development.

2.2.1 Mining

In Zoucheng city, mineral resources include coal, granite, limestone, marble and kaoline, and industries related to these resources have been the pillars of the local economy. Geologically, Zoucheng is part of the Yanzhou Coalfield. In China, Yanzhou Coalfield is one of the eight major regions for coal resources (Hu, et al., 2012).

Zoucheng is a national energy industrial base and the coal deposit had total reserves of more than 3.014 billion tons at the end of 2007 (Xu, et al., 2014). In Zoucheng, coal is found over 357 square km, representing about 22.13% of the local land area (Yang et al., 2007). In recent years, the stock of minerals has been reduced with local economic development and the outputs are currently at a stable volume.

Zoucheng city has nine coal mines: BaoDian, DongTan, NanTun, BeiSu, LiYian, TaiPing, HengHe, TangCun, and LuoLing. Regarding the length of service remaining for these coal mines, two mines (TangCun & LuoLing) are already closed (The conciliation commission of regional economies, 2006, Dec). The productive lifespan for the rest of the local mines was estimated at more than 25 years on average at the end of 2006 (The conciliation commission of regional economies, 2006, Dec).

However, the excessive exploitation of mineral resources has decreased the local mines' life span in recent years. Taking four local mines as an example, the total coal deposits for NanTun, BeiSu, BaoDian, and DongTan are about 0.3 billion tons (Li, 2012). The life spans for these four mines have reduced from more than 25 years to only 15 years, if the actual annual extraction continues at 20 million tons instead of the designed annual capacity of about 12 million tons (Li, 2012).

A Chinese proverb says: "be prepared for danger in times of safety" and local economic development is no exception. First, for years coal exploitation has brought serious pollution to the natural environment (as noted in Chapter One), which is likely to increase environmental costs and challenge economic growth. Second, it is undeniable that coal is a non-renewable resource. The stock of local mine resources will be depleted in future, especially considering the life spans of the local mines discussed above. The question is how mining households can adapt to this before most of the mines close. It is important to develop alternative industries before mineral resource depletion is complete. Otherwise, the local economy will suffer serious problems such as increasing unemployment and decreasing local financial income, affecting the administrative resources for county development. For instance, Doncuan City in China was wealthy in copper ore. The product value from local copper ore made up more than two thirds of the local GDP. In 1958, Doncuan City became a prefecture-level city. However, when all four mines were closed, Dongcuan was downgraded to a county-level region (The conciliation commission of regional economies, 2006 Dec). When a prefecturelevel city is downgraded to a county-level region, allocations of central finance and the city's attractiveness to investors is affected. This, in turn, reduces local economic development and residents' livelihoods. In my fieldwork, I have seen people adapting to mine closures by changing their livelihoods; for example, planting vegetables and fruits. Therefore, the local government should have some awareness of shocks to the economy. For example, current miners may start to expand their livelihood choices by undertaking activities such as developing agricultural enterprises and ecological tourism.





Source: Author's compilation from Zoucheng Statistical Yearbook, 2000-2013

Figure 2.3 shows coal production in Zoucheng over the period 2000-2013. As can be seen from the graph, there is an obvious steady increase in coal production in the first four years (from about 29 million tons in 2000 to about 49 million tons in 2003). After that, coal production, with minor fluctuations (e.g. in 2006, 2008, 2010, and 2012) has gradually decreased to about 38 million tons in 2013.

With a wealth of coal and other minerals, Zoucheng has enjoyed remarkable levels of long-term economic growth. The average GDP growth rate was more than 14% per year from 2003 to 2013 (Zoucheng Statistical Yearbook, 2013). Hilson (2001), investigating the impacts of mines on community groups, showed that mines can create a positive socioeconomic impact in local communities.

However, recent changes in the macroeconomic environment may potentially change the benefits obtained from local coal enterprises. There has been a slowdown in China's economy; the GDP growth rate was 6.9% but coal consumption reduced by 3.7% in 2015, while the revenue from domestic tourism increased by 13.1% (NBSC, 2016). In my field work, people working in the local mining industry presented similar views on the coal market following the downturn – sales in the local coal market showed sluggish growth in recent years, as did the levels of coal production (see Figure 2.3). Therefore,

developing alternative enterprises (e.g. tourism) is an important step in removing dependence on a resource-based economy.

2.2.1.1 Resource dependence

The local economy depends upon mining industries. As a resource-based area, local industries are heavily dependent on coal mining, the coal chemical industry, electric power and other coal-related industries (Guo & Liu, 2012). In addition, employment shows similar trends: total employees numbered 230,000, of whom 46% (about 105,000) were working in the local mining industry (Guo & Liu, 2012). On average, employment in local mining was stable at around 100,000 from 2000 to 2013 (Zoucheng Statistical Yearbook, 2000-2013).

2.2.1.2 Economic growth

A case in point is Yanzhou Coal Mining Company Limited. It was the first big national enterprise that chose Zoucheng City as an operating site. In 1973, Yanzhou Coal Mining Company Limited (the largest mining company in Zoucheng city) was established in Zoucheng City. Thereafter, other foreign enterprises such as the Zouxian coal-fired power plant and Lineng Enterprises were also attracted by the availability of local mineral resources. Therefore, it is generally agreed that Zoucheng City's economic development started in that year. For instance, the average local GDP growth rate was only about 7.5% from 1949 to 1975; then the growth rate increased to about 11% from 1976 to 1989, and to about 21% in the 1990s (Liu & Liu, 2000). It must be emphasized that economic development is a very complicated process. It is not simply claimed that the boost to the local economy was completely caused by a mining company. However, it is apparent that the company had or potentially had significant positive impacts on the local economy.

2.2.1.3 Contribution of mining to industrial output value

In Zoucheng city, total industrial output values have been primarily calculated from the eight biggest industries: coal mining; electric power, gas and water supply; manufacture of general purpose machinery; agricultural and subsidiary product processing industries; textile industries; beverage manufacturing; manufacture of non-metallic mineral products; and manufacture of chemical raw materials and chemical products. In Zoucheng city, most of the local manufacturing is related to local mineral resources, and coal mining makes a large contribution to the total value of secondary industry (SURPDI, 2012). Local mineral resources and their related products have advantages in terms of scale and geography over other local products in terms of profitability because Zoucheng has abundant and potentially profitable mineral resources. For "quick financial gains", local industrial output values are mainly driven by the mining industry. For instance, the mining industry represented the majority of total industrial output from 2000-2013. Figure 2.4 shows that the local mining industry contributed more than half of the total industrial output value from 2000-2013, notably 77.6% in 2007 and an average of 66.7% over the total period.



Figure 2.4: *The contribution of local mining industries to total industrial output value,* 2000-2013 (Unit: %)

Source: Author's compilation from Zoucheng Statistical Yearbook, 2000-2013

2.2.1.4 Industry structure

In Zoucheng city, mining not only supplies financial income, but also changes the local industrial structure. With more financial support, local manufacturers, the chemical industry, textile industry and other industries have been boosted in recent decades.

Figure 2.5 shows the local industrial structure distribution from the 1950s to 2013. The bar chart illustrates the proportion contributed by each industry sector to the local economy. The local industrial structure distribution shows sharp contrasts between primary and secondary industries. For tertiary industry, there has been a substantial increase compared with the other two sectors. At the beginning of the establishment of the People's Republic of China, Zoucheng's economy was heavily dependent on agriculture and was largely self-sufficient. It can be seen that primary industry earned about 77% of total GDP in 1952. Primary industry employed more than 97% (about

186,500) of the local labour force in 1952, with only about 3% in the secondary and tertiary industries (Liu & Liu, 2000). However, primary industry has decreased over the years. Mining became a component of the local economy in the 1970s and has provided an ongoing boost to local economic development. Secondary industry has held the leading position in the state economy since the 1990s. Figures for tertiary industry have fluctuated slightly and increased over the same period. Overall, the local economy experienced a transition from an agricultural to an industrial base. Figure 2.5 indicates that the industrial structure is not static and can change with different industrial contributions to the local economy. Therefore, the local industrial structure could be moved to a higher reliance on tertiary industries as an alternative livelihood strategy for local community and economic development.



Figure 2.5: The industrial structure distribution of local GDP in Zoucheng (% of GDP)

Source: Author's compilation from Liu & Liu, May 2000; Zoucheng Statistical Yearbook, 2000-2013

The distribution of mineral and tourism resources (see Figure 2.2) influences the development of the relevant industries in Zoucheng. Industry in the west and agriculture in the east or "rich west and poor east" are general development features of Zoucheng in recent decades. In recent years, the outputs from western villages/towns provided more than 80% of the total output in Zoucheng (SURPDI, 2012). In consequence, the west developed areas of serious land subsidence due to the intensive operation of mineral resource-based industries in this area, while the ecological advantages of the

east played no role due to the underdeveloped transportation system and other relevant public infrastructures (SURPDI, 2012). Zoucheng seems to have had a passive division of the territory in terms of regional industry, but this division did not fully use all of the local resources (such as tourism). In Zoucheng, primary industry is distributed primarily in the eastern and southern areas; secondary industry is highly concentrated in the central and western areas; and tertiary industry is confined to the central city (SURPDI, 2012). However, tertiary industry could potentially develop in many other areas due to a wide distribution of local tourism resources (see Figure 2.2).

2.2.1.5 Residents

Mineral resources have brought benefits for local residents. For instance, householders use natural gas in place of firewood, coal, and electricity in cooking.

In March 1983, the Ministry of National Coal Resources ratified a gas-engineering project in Zou County (the original name for Zoucheng city). This project was a joint operation between Yanzhou Coal Mining Company Limited and the Zoucheng government. In 1989, the Zouxian Gas Company was established. Initially, only around 400 households benefited from this local facility. In 1998, the total length of gas pipes grew to 162 kilometers, which could provide services for about 32,000 households and about 117,500 people (about 59.1% of the total urban population) in the urban area (Liu & Liu 2000, May). In 2005, the gas pipes were increased to 368 kilometers and about 85% of urban citizens had access to the gas facility (The conciliation commission of regional economies, 2006, Dec). With more gas facilities provided, dwellers could conveniently use clean, cheap energy for cooking.

The most direct benefit for dwellers was net income growth. Figure 2.6 presents annual cash income per capita for rural dwellers from 1999-2013. It also divides the region into two categories: mining areas and non-mining areas. If at least one mine exists in a village town, it is defined as a mining area. Of 17 village towns, six are located in the mining area. As shown in Figure 2.6, annual cash income per capita increased in both

mining and non-mining village towns from 1999-2013. For instance, the cash income was about 13,652 yuan in mining areas in 2013, which was more than quadruple that in 1999. The income gap between mining and non-mining areas also expanded. In 1999, the annual cash income per capita for residents in mining areas was about 377 yuan more than non-mining residents on average. In 2013, the average gap increased to about 1,700 yuan. Residents in mining areas are thus likely to earn higher incomes than those in the non-mining areas of Zoucheng city.



Figure 2.6: Net incomes per capita for mining and non-mining areas, 1999-2013

Source: Author's compilation from Zoucheng Statistical Yearbook, 2000-2013

2.2.2 Tourism

This section will introduce the characteristics of tourism development in Zoucheng.

2.2.2.1 Tourism resources

The following paragraphs include the structure of local tourism resources (i.e., category diversification) and discuss cultural areas and scenic spots, the area's excellent geographical position, the status of traditional culture in local society, and market trends.

a) Diverse categories

Zoucheng is a vast county, rich in tourism resources. In Zoucheng, there are hundreds of unmovable cultural relics, more than 15,000 pieces in local museums, and more than 60 local relic protection units (DPZ, 2015).

Numerous historical sites and scenic spots provide great potential for developing tourism. The local government and the Tourism Planning and Design Institute of Beijing conducted joint research on local tourism development (March, 2008). They collected relevant information about local tourism resources and divided these resources into eight categories. Table 2.1 shows the structure of local tourism resources, including natural and historical resources. Landscape architecture takes first place in the total number of tourism resources. Following that, geological landscape (about 19%) and water scenery (about 19%) take second place. Historic ruins and sites (10%) follow. Taking local tourism resources and the prospect of tourism marketing into account, local government and the Tourism Planning and Design Institute of Beijing (March, 2008) divided 131 local tourist attractions into three classifications: A-grade resources that are rare, typical, specially protected by the state, and well-known both at home and abroad; B-grade resources under provincial protection; and C-grade resources under protection at city or county level. These resources play a supplementary role on the local tourist track. In the Zoucheng area, there are 15 A-grade, 46 B-grade and 70 Cgrade resources (Zoucheng municipal government, March, 2008).
Resource Type	Ν	Proportion	
Landscape architecture	48	22%	
Geological landscape	41	19%	
Water scenery	41	19%	
Human activities	27	13%	
Historic ruins and sites	22	10%	
Biodiversity	20	9%	
Tourist commodities	11	5%	
Others	5	2%	
Total	215	100%	

Table 2.1: The structure of local tourism resources in Zoucheng

Source: Author's compilation from Zoucheng government, March, 2008

In recent years, the local government formulated regional plans to strengthen tourism resource management. The national key cultural relics protection units were increased from four to seven and 28 provincial levels were established in Zoucheng by the end of 2015 (DPZ, 2015). According to the master plans in Zoucheng (Bureau of City Planning of Zoucheng city, 2014), local government is undertaking further programming on natural scenic places and cultural monuments, with the intention of creating about 13 potential tourist resorts in the next few years. These tourist resorts will combine ecological and modern tourism concepts. These scenic areas will include forests, mountains, ecological environments in subsidence areas, and some cultural heritage assets.

b) Potential tourist attractions

Zoucheng is rich in natural landscape and cultural tourism resources. The east of the city is surrounded by mountains (there are about 190 mountains in Zoucheng), and there are more hill areas in the west (SURODI, 2012). The elevation ranges between about 35m and 645 m (SURODI, 2012). There are three nature reserves (Shibapan, Fushan, and Yishan) in the local area (SURODI, 2012). The area of Shibapan is about 3,000 ha; it is a reserve that protects primarily plants such as spruce, and other rare tree species. Fushan (about 3,000 ha) preserves forest ecology. Yishan is about 5,000 ha, and focuses

on historical sites and scenic areas in Zoucheng (SURODI, 2012). Zoucheng also has two wetland parks (Beisu & Xiangcheng). Beisu Wetland Park is about 1,500 ha and is located in a subsidence area.

In addition, Zoucheng city has a long history and a flourishing culture. In the Qin Dynasty (221 B.C-207 B.C), Zou was named as one of the counties by the government. Confucius and Mencius, both of whom were great philosophers and educationalists in Chinese history, were born in Zoucheng city. Therefore, Zoucheng is considered the home of the Confucian system. In 1995, the State Department listed Zoucheng city in the third group of National Famous History and Culture Cities (Zoucheng government, March 2008). Local places of historic interest include the "Three Mengs," which include Meng Mansion, Meng Temple (the place where people feted Mencius) and Meng Forest. In addition, Zoucheng has beautiful scenery such as Yi Mountain, Gang Mountain and the Fenghuangshan Buddha. These cultural and natural resources represent potential tourist attractions for Zoucheng city:

The Three Mengs:

The "Three Mengs" are the most famous and typical historical sites in the city. They include the Meng Temple, the Meng Family Mansion, and the Meng Forest. The Mencius Temple is also called the Second Saint Temple in China. It is located in the southern part of the urban area and is now a key cultural relic under state protection.

Yi Mountain:

Yi Mountain, a beautiful scenic spot, sits in the southeastern part of the city, about 10 kilometers away from the city centre. Mencius said: "Confucius ascended the eastern hill, and Lu⁵ appeared to him small. He ascended the Tai mountain⁶, and all beneath the

⁵ The State of Lu was one of the kingdoms in China during the Spring and Autumn Period (770 B. C-476B. C).

⁶ It is the most famous mountain in China.

heavens appeared to him small." It symbolizes high aspirations in people's lives. Many scholars believe that the "eastern hill" is Yi Mountain.

Fenghuangshan Buddha:

This is located in the east part of Zoucheng city and was carved in the Tang Dynasty (618 A.D.-907 A.D.). The Buddha is set into a cliff on Fenghuangshan Mountain. It is about 4.2 metres tall.

Chongxing Tower:

Located in the central part of the city, Chongxing Tower is an octagonal building, 9 stories (33 metres) high, with 10 eaves on the roof. It was built in the Song Dynasty (960 A.C.-1279 A.C.), about 1,000 years ago.

Double Saints Stones:

These are two important stones that indicate the city is the birth place of both Confucius and Mencius. Confucius (551-479 B.C.) was one of the world's greatest philosophers and ancient thinkers. Confucius' theories emphasized the importance of societal ethics and bonds, which profoundly influenced Chinese society and culture. Mencius (372-289 B.C.) inherited and passed on Confucius' theories. He stressed the idea of benevolent government and developed Confucianism as the core of Chinese culture. The Double Saints Stones are in the Railway Station Square of Zoucheng city.





Source: Authors' compilation from Liu, 2000

c) Geographical position

Transportation, public transport and infrastructure can directly impact the growth of tourism (Wen, 1997). Due to its superior geographic location, Zoucheng has made great

achievements in the modernization of its transportation networks in the past 20 years. The completed road system in the city reached 3545.14 km by the end of 2013 (The Bureau of Statistics of Zoucheng City, 2014). Many railways, highways, and other traffic routes run across the urban centre, including the Beijing-Shanghai railway, Beijing-Fuzhou highway, Beijing-Shanghai highway, 104 National Highway, Beijing-Hangzhou Canal, and the Beijing-Shanghai high-speed railway. According to the Zoucheng municipal government (March, 2008), the distance from the city centre to Jinan international airport is about 180 km. It is about 60 km to Jining⁷ domestic airport; about 210 km to Rizhao ocean port, and about 400 km to Tsingtao ocean port. In Shandong Province, Jining, Jinan, Rizhao, and Tsingtao are common entry points for tourists. These convenient traffic routes provide the means and capacity to attract more tourists and develop local tourism, if distinctive products are developed and marketed.

d) Market demand

In the local tourism market, annual tourism receipts are increasing steadily with increased annual arrivals. It indicates the positive relationship between number of tourist arrivals and tourism benefits, which provides some hints for building up the tourism simulation model in Chapter Seven.

Figure 2.8 shows tourism development in Zoucheng from 2000-2013 (Zoucheng Statistical Yearbook, 2000-2013). The bar chart and line graph represent annual arrivals and annual tourism receipts in the period 2000-2013 respectively. As can be seen from the graph, there is a steady increase in both of these tourism indicators. In 2000, there were about 543,000 tourists and 0.51 billion yuan in tourism receipts. In 2013, arrivals increased to about 1,925,000 and tourism receipts were approximately 6 times more (3.3 billion yuan) than in 2000. The growth rate in annual receipts has been greater than that of annual arrivals, particularly from 2003-2004, due to the ticket price rise for local tourist attractions in this period.

⁷ Zoucheng's higher level administrative city



Figure 2.8: Tourism development in Zoucheng, 2000-2013

Source: Author's compilation from Zoucheng Statistical Yearbook, 2000-2013

2.2.2.2 Characteristics of tourism operations

The following paragraphs introduce the characteristics of local tourism operations that may influence tourism development.

a) Government role and private business

In the case of Zoucheng city, government structure and market mechanisms both influence local tourism resources and enterprises. Private business operation and the ownership of tourist attractions are clearly distinguished.

In the city, almost all of the tourist attractions are owned by the state or local government. Table 2.2 below gives a brief review of the nature of ownership of the main tourist attractions in the city. Of the 17 local tourist attractions, 16 belong to the state. Most of these tourist attractions are affiliated public institutions of the Zoucheng Cultural Relics and Travel Bureau (World Bank, 2010). Because of the government role in ownership of the main tourist attractions, the financial support and preservation of local cultural relics, and salary payments for relevant employees are still dependent on state and local government (World Bank, 2010; Zoucheng government, March 2008).

Tourist attractions	Higher Authority		
Mencius Temple & Mencius	Zoucheng Cultural Relics and Travel		
Family Mansion	Bureau (ZCRTB)		
Meng Forest	ZCRTB		
Zoucheng Museum	ZCRTB		
Tieshan, Gangshan Carved	ZCRTB		
Stone Calligraphy			
Yi Mountain	ZCRTB		
Fushan tourist district*	None		
LuHuangWang Mausoleum	ZCRTB		
Mencius Forest	ZCRTB		
Former residence of Mencius in	Fu Village of Qufu city		
Fu village			
Scenic spot of Shibapan	Zoucheng Forestry Bureau		
Zouxi Subsidence area	Beisu, Pingyangsi Village Town		
Mencius's Mother Forest	Qufu Cultural Relics Bureau		
Tiexi public park	YanZhou Coal Mine Bureau		
Sports park	Zoucheng Administration of Sports		
Wubaoan Mountain	Zoucheng Forestry Bureau		
	Bureau of Water Resources of Zoucheng		
Mencius Lake	city		
Tieshan Park	Zoucheng Landscape Department		

Table 2.2: Nature of ownership (Publicly owned) of the main tourist attractions

*:None ownership

Source: Author's compilation from Zoucheng government, March 2008; World Bank, 2010

On the other hand, business operations are moving toward a market economy. Based on local government studies (2013), the proportion of privatization in 11 local hotels or restaurants and in 14 travel agencies⁸ was higher than that of local tourist attractions. The proportion of private and public distribution of local tourist attractions, hotels or restaurants and travel agencies is illustrated in Figure 2.9. For local tourist attractions, ownership is dominated by the state or large national companies. This compares to less than 65% state ownership of hotels and restaurants. For local travel agencies, the rate

⁸ These hotels, restaurants, and travel agencies are relatively large in terms of their business operations and are well known in the local area.

of private ownership is above 85%. Therefore, local government plays an essential role in local tourism activities. Local government can implement relevant policies on tourism development. With government support, residents might gain benefits such as tourism employment and income from tourist arrivals.



Figure 2.9: The distribution of private and public ownership in local tourism

As previously discussed, the Zoucheng government plays an important role in the nature of ownership of the local tourist attractions. The operations of these local tourist attractions rely primarily on state or local government financial support (Zoucheng government, 2008). The entrance fees provide only about 20% of the required budget to maintain and repair the cultural heritage assets in Zoucheng (World Bank, 2010). Lack of required financial resources might result in failure to manage and protect local cultural heritage assets and reduce the satisfaction of tourists in Zoucheng (World Bank, 2010). Furthermore, local mining activity generates pollution issues, which may impact on the quality of the local environment and consequently reduce tourist arrivals. On the other hand, mining contributes greatly to the local economy. Restricting mining activities may have an impact on financial assistance at county level, which in turn would affect financial investment in the operation of local tourist attractions. In other words, local mining activities can indirectly support the operation of local tourist

Source: Author's compilation from Zoucheng Tourist Administration, 2013

attractions. Tourism development relies on mining income, which is not necessarily self-sustaining. Thus, the development of local tourism must not be conditional upon the willful neglect of mining activities.

Given the substantial involvement of government, understanding what type of role the government plays in tourism development is very important. Zhang, Chong, and Ap (1999) stated that "without identifying the nature of the government's role, neither past tourism development in China can be systematically analyzed, nor can future trends be forecasted." (p. 472). They also summarized six roles that the Chinese government should play in the development of tourism:

- Operator: concerning relevant ownership in tourism resources, relevant tourism business activities and providing infrastructure for tourism development;
- Coordinator: Coordinating relevant activities of different government branches to develop tourism;
- Promoter: promotion and marketing activities for tourism;
- Regulator: making, regulating, formulating and implementing relevant policies to manage tourism activities;
- Investment stimulator: attracting more tourism investment or public investment programmes for tourism development through the provision of financial incentives;
- Educator: stimulating tourism development by establishing tourism education institutions and training programmes.

In the Zoucheng area, the local government is expected to play these roles to stimulate tourism and economy development. For instance, local government might invest in public infrastructure improvements, transport planning and road building between tourist attractions and rural households. Furthermore, local government can coordinate, organize, and train farmers to help them pursue an alternative livelihood, including rural-tourism business operation based on families or villages, and the design and operation of broad activities and interests so as to attract more tourists. Local government can also carry out market promotion for tourism such as the design and distribution of posters, the placing of newspaper advertisements, maintenance of web site content, development of smart phone apps, and the posting of billboards.

b) Lack of destination identity

According to tourism studies in Zoucheng (Zoucheng government, 2008), more than 70% of total tourist arrivals selected the "Three Mengs" and Yi Mountain as their main travel destinations in Zoucheng. About 38.5% of visitors were day-trippers and most of the tourists stayed at relatives' or friends' homes rather than hotels. In Zoucheng, tourists' average stay was only 0.3 days (World Bank, 2010). The factors most in need of improvement for local tourism purposes were found to be traffic conditions and entertainment facilities (Zoucheng government, 2008). Furthermore, a lack of available information, incentive or opportunity also prevents visitors from spending more time and money in Zoucheng (World Bank, 2010). Therefore, despite the large numbers of tourists (see Figure 2.8) in Zoucheng, residents derive limited benefits (e.g. improved local standards of living) from tourism (World Bank, 2010). A large number of tourists with short stay times and low consumption may be a serious issue for tourism development in the Zoucheng area (Guo & Liu, 2012).

In addition, Zoucheng's tourism resources have similar characteristics to those in other surrounding cities and the area lacks a distinctive destination identity. More unfavourably, local tourism resources are overshadowed in architectural scale, area, and history by other cities' resources. For instance, the architecture of the Mencius temple and mansion are very similar to those of Confucius, so most tourists do not extend their visit to Zoucheng (World Bank, 2010).

Table 2.3 selects four main tourist attractions (the Mencius Temple, Mencius Family Mansion, Mencius Forest, and Yi Mountain) to compare with similar resources in other surrounding cities (i.e., Qufu and Taian). The main tourist attractions in Qufu city comprise the Confucian Temple, Confucian Family Mansion, and Confucian Forest, each of which has a longer history than the Three Mengs. The Confucian Temple was established in 478 BC, while by contrast, the Mencius Temple was established about 1,400 years later (i.e., 1037 A.D.).

A further potential problem is that Zoucheng's tourist attractions are smaller in terms of size and reach. For instance, the Mencius Family Mansion is about 21,000 square metres, which is approximately one quarter the size of the Confucian Family Mansion, at about 75,000 square metres.

In quality ratings, most of the Zoucheng tourist attractions are national cultural relics. However, local tourism development is limited by its small regional or county-level environment, even though some tourism resources are evaluated as state-level.

Of course, area, history, and the quality ratings of tourism sites are not enough to measure the competitiveness of local tourism resources. However, lack of distinctive features could reduce the city's ability to cope with regional competition and risks. Geographic closeness and very similar characteristics enhance the potential to develop regional competitiveness for tourism development in Zoucheng. For instance, the farthest tourist attraction is Tai Mountain, which is less than 100 kilometres from Zoucheng. Outside visitors might prefer to travel to Tai Mountain instead of Yi Mountain because of the history and reputation. Thus, tourism in Zoucheng needs to develop a different emphasis (e.g. rural tourism) in order to be more distinctive, more competitive and attract visitors. The emphasis in further developing the local tourism market should be on providing a more diverse range of goods and services to satisfy the needs of a broader range of visitors (World Bank, 2010). With a more diverse range of goods and services, local tourism can present its own destination identity, which encourages visitors to venture across the boundaries of local cultural heritage assets. Furthermore, local residents may gain more work opportunities related to the tourism industry and become engaged in the care and maintenance of cultural heritage assets as well.

			- ·	
		Tourism Area or	Renown and	
	History	Distance	honours	Distance from Zoucheng
			World cultural	
Confucian			heritage &	
Tomplo	478 BC	96,000 square meters	State cultural	23 km
remple			relics	
			protection unit	
			State cultural	
Mencius	1037 A.D.	23,250 square meters	relics	urban city centre
Temple			protection unit	·
			World cultural	
Confucian			heritage &	
Family	1038 A D	75 000 square meters	State cultural	23 km
Mansion	2000/021		relics	
			protection unit	
	Song			
Mencius	Dvnastv:		State cultural	_
Family	about 800	20,812.5 square meters	relics	urban city centre
Mansion	vears		protection unit	
	,		World cultural	
			heritage &	
Confucian	about 2,400	2 km ²	State cultural	23 km
Forest	years		relics	
			protection unit	
			State cultural	
Mencius	1037 A.D.	0.61km ²	relics &	urban city centre
Forest			protection unit	,
			World natural	
Tai		426 km ²	& cultural	Less than 100 km
Mountain			heritage	
			Provincial-	
Yi Mountain		8.000 meters	level scenic	Southeast of urban city
		e,eee meters	snot	
			spor	

Table 2.3: Comparison between Zoucheng tourism resources and tourism resources in nearby cities

Source: Author's compilation from Zoucheng government, March 2008

2.2.3 Agriculture

a) Rural population & labourers' demand for work

Traditionally, China was and still is the biggest agrarian country in the world. Even with the past several decades of reform, the rural population still accounts for a considerable proportion of the total population. In Zoucheng city, rural residents account for nearly 65 percent of the total population. Figure 2.10 shows the proportion of the Zoucheng population classified as rural from 2003-2013.

It is clear that in recent years, the rural population has declined, but it still accounts for a majority of the population. Due to administrative redesign, some rural areas were redesignated as urban in 2002. About 75% of the population was classified as rural prior to the administrative re-design in 2000-2002. Thus, Figure 2.10 only involves the period from 2003-2013. As shown in Figure 2.10, the rural population fluctuated slightly and then fell to about 64% of the total in 2013. The biggest decline was in 2005, possibly because of reforms of the household registration system by the Chinese government in 2005. By following new policies, some of the rural migrant workers who met certain eligibility requirements became urban residents.

It is worth noting that the rural population stabilized at around 64% of the total population after 2007 in Zoucheng. Surplus labourers may exist in the local rural area, which is a general scenario in the Chinese labour market (Fan et al., 2003; Wang & Cheng, 2001). Fan, et al., (2003) and Wang & Cheng (2001) suggested that the marginal labour productivity in the agricultural sector remained at a very low level in China. The main reason is the continuing large surplus of rural labourers. In other words, the issues of rural development need more attention as Zoucheng undergoes the process of urbanization and industrialization.



Figure 2.10: Rural population percentage in Zoucheng city, 2003-2013.

Source: Author's compilation from Zoucheng Statistical Yearbook, 2003-2013

Mining cannot satisfy the demand for employment from the rural labour market. According to local Yearbooks (2003-2013), each village town (17 village towns in total) on average has about 14,287 labourers in agriculture. The annual net income per capita for rural dwellers increased from 1999 to 2013. For instance, the net income per capita was less than 3,000 yuan in 1999, and increased to around 12,500 yuan in 2013 (Zoucheng Statistical Yearbook, 1999 & 2013). Although the net income per capita in rural areas increased more than fourfold from 1999 to 2013, the growth rate and income are still low compared to other industries. For instance, the monthly wages for miners increased from 1,213 yuan in 2000 to nearly 3,593 yuan in 2013 (Zoucheng Statistical Yearbooks, 2000; YanZhou Mining Group 2014). Although net income per capita and monthly income are two different standards, the income gap is still obvious. Thus, rural labourers may prefer working in local mining to get higher levels of pay. However, mining and relevant industries have stably provided about 100,000 jobs in the local labour market in recent years, including rural and urban labourers (Zoucheng Statistical Yearbook, 2000-2013). Local mining cannot satisfy the job demands of the total number of rural labourers ($14,287 \times 17 = 242,879$). Rural households need alternative livelihood choices such as rural tourism.

b) Owning & using farmland

Rural households have difficulty transferring farmland to provide financial capital and engage in any alternative livelihoods in China. The ownership of farmland is one of the remarkable differences between rural and urban households, which could allow residents to gain financial capital by selling or transferring farmland in the business market. However, farmland property and living places are difficult to convert to financial capital in rural China. According to The Law of Land Management of the People's Republic of China, "the land of the countryside and suburb of towns, except those legally stated as the property of the country, belongs to the collective" (as cited in Luo & Fu, p 445-p446, 2009). However, the Village Committee normally fulfills the legal position of farmland owner. Peasants' private land property rights are weakened through the use of policies to constrain peasants' ability to dispose of land, such as prohibiting leases, loans involving land mortgages, or transfer, sale, or lease for non-agricultural construction purposes (Cai, 2003; Luo & Fu, 2009). Due to the difficulties of land transfer, making good use of farmland and increasing the value of farm products would be a key link in alternative livelihood development for rural households.

Even for rural households with distributed farmlands, the profit motive or opportunity cost might change rural labourers' incentives to cultivate farmland. For instance, rural labourers might choose mining or other well-paid jobs instead of farming as their main activities. Each rural household might have a similar or limited farmland area, but they might have a different arable land size each year. Different rural households with different alternative livelihood choices might change their intentions to cultivate their land. In other words, rural households cannot fully take advantage of the natural capital (i.e., farmland) in their hands. Some alternative livelihood choices in rural areas might be necessary to make full use of available farmland resources. Figure 2.11 illustrates the arable land per household in Zoucheng city from 2000 to 2013. From 2000 to 2009, while the average arable land size per household increased slightly in 2001, 2004, and 2006, in general the arable land per household decreased gradually from about 3.8 mu⁹ per household in 2000 to about 3.65 mu/household in 2009. After 2009, the arable land size per household gradually increased again to about 3.79 mu per household in 2013. Coincidentally, many local government policies to regulate agricultural structure, develop agricultural industrial production, ameliorate resource distribution, develop agriculture and boost market competence so as to increase rural households incomes, were begun around 2010 (e.g. Zhang, 2010; Ma & Duan, 2010; Du, 2010; Sun et al., 2010; Zoucheng government, 2012). Therefore, carrying out relevant land operation policies may be a useful way to raise enthusiasm for farming and allow households to develop more livelihood choices for themselves.

 $^{^9}$ A traditional unit of measurement for a rable land in China. 1 Ha=15 Mu and 1 Mu $\approx\!666.7$ square meters



Figure 2.11 Arable land per household in Zoucheng city 2000-2013 (mu/household)

Notes: 1. Mining area: at least one mine in a village town 2. Mu: A traditional unit of measurement for arable land in China. 1 Ha=15 Mu and 1 Mu \approx 6666.7 square meters. *Source:* Author's compilation from Zoucheng Statistical Yearbook, 2000-2013

c) Market demand

In Zoucheng city, farmers have two harvest seasons (summer and autumn). According to Zoucheng Statistical Yearbooks (2000-2013), the main crops that local farmers grow are corn and wheat and typical livestock includes pigs, goats, and cows. Furthermore, local farmers also plant some cash crops such as grapes, apples, kiwifruit, and cotton. This represents a potential way to enable diversification of local agricultural products to satisfy different demands in the market. However, farm products have lacked sales, because they rely on local government to find a market.

In the early 1980s, the socialist reformation of agriculture (Household Responsibility System) was implemented in China and Chinese rural households were free to choose the crops they grew. However, farmers' planting choices in terms of diversification may still depend on local villages' strategies (Demurger, Fournier, and Yang, 2010). Local farmers will increase agricultural outputs and higher-value farm products when they find a lucrative market.

In Zoucheng village towns, the government encourages outside investors and corporates, along with local peasant associations, to plan, manage, and readjust the

types of food crops grown or livestock raised, even finding relevant outlets for local farmers' products. For instance, Taiping village town, together with the Langu Gardening Company, established in total a 3,000 mu area of pear plantings distributed among about 1,000 local farmers, using contracted management of rural land in Zoucheng (Sun & Shao, 2010). The government in Zhongxin village town in Zoucheng developed 5,000 mu for cherry production and 1,500 mu for peach cultivation, which benefited about 3,800 rural households (Wang & Tang, 2013). Both of these village towns implemented relevant policies about cash subsidies, longer land-use rights, favourable terms, and training to encourage rural activity restructuring. The Zoucheng government also helps local farmers to find marketing channels for farm products. For instance, the government in Kanzhuang village town helped rural households to sell potatoes by contacting potential customers, promoting crops on the internet, and signing sales contracts (Wang, 2013). The government in Kanzhuan village town established specialist cherry farming cooperatives to attract wholesalers to purchase about 20,000 kg of cherries (Sun, 2014).

Activity diversification depends heavily on policies implemented at local level to promote economic restructuring and ignores disparities at the "individual level" (households). For instance, disparities in livelihood capitals might affect rural households' ability to create income opportunities out of traditional cropping. Local rural households may need an alternative livelihood with scope for initiative rather than being passive order takers.

d) Eating habits

Because of Zoucheng's location in North China, the average local family uses little rice; the staples are primarily wheat or corn. Local farmers also prefer to use corn to feed animals. These two uses are the primary reason for planting wheat and corn in Zoucheng. Wheat and corn provide the main raw materials for food, which affects the characteristics of local food. A characteristic food in Zoucheng is "Sichuan Noodles", which are made from flour and have a spicy, hot flavour. More than 30 years ago, migrant workers in local mining companies used local food materials together with the characteristics of "Sichuan cuisine¹⁰" to invent "Sichuan Noodles". "Sichuan Noodles" are viewed as a major component of the cultural mix in Zoucheng, and could become one of the characteristics for local tourism development.

2.2.4 Comparing livelihood alternatives

For the two important resources (mining and tourism) in Zoucheng city, the number of jobs created and the monthly wages from these two industries are used as the basis for comparison.

In 2010, the monthly per capita average income in mining was about 3,100 yuan, compared to around 1,070 yuan in tourism. In 2013, the monthly per capita average income in mining had increased to about 3,600, while the tourism monthly income had increased to about 1,350 yuan (Zoucheng Yearbooks 2010-2013, Yanzhou Ming Group 2014 & Fang, et al., 2012). Mining has shown its advantages in local employment recruitment.

On average, the number of local labourers working in mining reliably fluctuated between 90,000 and 100,000 between 2010 and 2013, while formal tourism jobs were between 29,000 and 34,000 (Author's compilation from Zoucheng Yearbooks 2010-2013, Yanzhou Ming Group 2014 & Fang, et al., 2012).

In the local labour market, mining industries recruit their employees with the attraction of high pay, while many tourism jobs (e.g. restaurants) cannot hire workers (see Figure 2.12). Therefore, mining plays a dominant role in the local labour market due to its high rates of pay.

¹⁰ China's four major styles of cooking are: Sichuan, Guangdong, Shandong, and Huaiyang.

Figure 2.12: Recruitment advertisements in a local restaurant



The recruitment advertisement says: (1) five administrative office receptionists wanted; negotiable salary (2) ten waitresses wanted; 1,000-1,500 yuan/month (3) ten kitchen stewards wanted; 800 yuan/month (3) two cleaners wanted; 600 yuan/month. Food provided. Tele: 5*57716. Photograph taken by author in July 2010

a) Opportunities for rural tourism in Zoucheng city

In the last 20 years, the Chinese government has adopted tourism as one of the most important rural development strategies (Hu, 2008; Zheng&Zhong, 2004) and has given great support to the 'Speed up the Development of Tourism Industry' programme (Xu et al, 2009). Furthermore, tourism-based farm diversification and relevant rural tourism development have increasingly been considered a significant driving force for rural development, poverty elimination, sustainable development, and environmental protection (Sharpley & Vass, 2006; Ryan, Gu, & Zhang, 2009a & 2009b). Section 2.2.3 discussed rural population size, local farm products and relevant lifestyle (i.e., food), and some barriers to local agricultural development in Zoucheng, which may be important background information to provide a development basis for rural tourism.

b) Rural tourism operators

In local agriculture, the surplus rural labour force, the difficulty of transferring farmland to financial capital, and the influence of local government strategies on growing food crops or raising livestock can impact on farmers' livelihood choices. In tourism, financial dependence and lack of tourism products can limit local tourism development. In other words, local agricultural products need a market and tourism needs product diversification. Therefore, there is a possibility for cooperation between agriculture and tourism. Agriculture provides a premise for developing rural tourism, which may encompass diverse products such as "green" food and rural entertainment areas. Rural traditional customs, settlement culture, folk customs, and characteristic buildings provide rural tourism products of cultural value and can improve the tourist experience (He & Qi, 2012). These various rural assets could be "packaged" to encourage outside visitors and create tourism business opportunities (Yang, Cai, & Sliuzas, 2010). With respect to the tourist market, increasing numbers of tourists may consume the local cuisine, experiencing meals with different local flavors and local traditional customs, which creates more employment opportunities for rural tourism operators. In addition, tourists may prefer local and natural "green food", which will affect the prices and outputs of agricultural products in the whole supply chain and consequently improve farmers' incomes.

In Zoucheng, the government began to stress rural tourism development several years ago. At present, local rural tourism development is in its infancy and has not yet occurred on a large scale. However, it may give local farmers more livelihood choices and a chance for a better life.

Zhongxin is located in a coal mining area, and is also one of the local village towns that has promoted rural tourism development. Although mining still has a clear advantage in the local job market, local authorities are in support of the development of rural tourism products to diversify livelihood choices for residents. For instance, there are more than 2,000 greenhouses growing vegetables and fruits, which are used for tourism (including "pick your own" experiences) and ecological tourism (Pan, 2014). The local government has also set up special funds to strengthen support for characteristic plantings (e.g. strawberries), processing and eco-tourism. In Longshan village in Zhongxin, eight rural households have built greenhouses employing a total of 300 people. Each greenhouse received 5,000 yuan in funds (Zhang, 2013). In addition, they can hire more than ten temporary workers during the busiest seasons. The maximum visitors to Zhongxin are 3,000 daily (Zhang, 2013). Tianhuang village town is another case of developing local rural tourism. In the 2015 Spring Festival, Tianhuang promoted its "Happy farmer in Spring Festival" to provide local rural cultural experiences (e.g. experiencing the local life style), catering and accommodation for urban tourists (2015, STAI). With a contract operating system for rural tourism and relevant planting industries (e.g. fruit, vegetables and sweet corn), the monthly net profit is about 1,080 yuan on average (Sun, 2013; Pan, 2014;). This monthly income from rural tourism is very close to the formal tourism wage in the local city.

Figure 2.13 below illustrates most of the rural tourism activities run by self-employed tourism operators in Zoucheng.



Figure 2.13: Rural tourism activities in Zoucheng

All photos were taken by Zoucheng Government Propaganda and Zoucheng Tourism Administration and permission was given for their use in this thesis.

2.2.5 Concluding summary

Chapter Two provided the background information for this research. In Zoucheng, farmland, mineral and tourist resources support three industries (agriculture, mining,

and tourism) and constitute the main "engine" of the region's economic development. However, based on vast mineral resources and profitability, the mining industry has been the backbone of local economic development. The local mining industry provides higher pay, more job opportunities, and makes a higher contribution to local industrial outputs, which consequently influences the characteristics of local economic growth such as industry structure and the development of a mineral resource-dependent economy.

Because of the significant benefits for the local community, the idea arose that local economic development's first consideration should be mining, which disregards the long-term sustainability of local economic and livelihood development. Under the constraint conditions of limited and non-renewable mineral resources such as limited length of service remaining for local coal mines, resource depletion and industrial decline after large-scale exploitation, there will be problems facing local economies and livelihood development. Therefore, industrial transformation should be studied in advance, particularly focusing on diversity and sustainability. The industry structure must be adjusted; the key is to build potential alternative industries such as rural tourism to replace the mineral resources industry before the resources become exhausted. Zoucheng has rich tourism resources (e.g. tourist attractions) and agricultural resources (e.g. rural labourers and diverse agricultural products). Therefore, the obvious choice is to use these two resources (tourism and agricultural resources) to develop the local economy and seek new economic growth before mineral resource depletion becomes severe. Examining these three resources (agricultural, tourist and mining resources) and related industries, the following chapters will focus on shifting the local economy and livelihoods towards sustainability.

CHAPTER THREE: CONCEPTUAL AND THEORETICAL FRAMEWORK

This chapter reviews the literature on sustainable livelihood and social capital. Sustainable livelihoods development is the basis of the conceptual framework to establish the simulation model in this thesis. By following the sustainable livelihoods framework, some issues related to natural resource dependency (vulnerability context), social capital (household livelihood assets), government policies on developing alternative livelihoods (policies, institutions & process), livelihood strategies and livelihood outcomes can be integrated in a model. It is important in answering the second research question in Chapter One (i.e., Does the strength of social capital alter local livelihood patterns by diversifying household livelihoods and providing new livelihoods in industries with potential for development?). In addition, the discussion focuses on social capital concepts, because both research questions in Chapter One are related to social capital. In theoretical discussions of social capital, the reasons of using possessed social capital, the definitions, strength of social ties, economics of social capital, and the role of social capital in livleihoods are proposed as important research components in this study. All of these provide the practical guidelines for analyzing social capital status and the development of the simulation model. In the following chapters, the risks of natural resource dependency, social capital measures, and establishing and analyzing the simulation model are the concrete implementations of the research.

3.1 Introduction

As discussed in Chapter Two, the base of problem in Zoucheng is government overemphasis on mining to achieve economic growth targets and less emphasis on helping people to realise alternative livelihoods. So, this thesis focuses on social capital as way to open doors to new livelihoods. By following the main objective of this study (i.e., to investigate the role of social capital during the transition to alternative livelihoods, also see Chapter One), a hypothesis is that strengthed social capital can significantly alter household livelihoods and their sustainability. Thus, livelihood theory and social capital are two important components to guide the studies on social capital and livelihood development in this thesis.

By following hypotheses, the essential components of the conceptual framework for this thesis include two important theories: sustainable livelihood and social capital (see Figure 3.1). Sustainable livelihoods provide a practical and operational framework to establish the simulation model for this study (see Chapter Seven). This chapter will particularly discuss the framework of sustainable livleihoods, its main compnents (e.g., vulnerability context, policies, institutions & process, livelihood strategies and outcomes, etc.) integrating with local features (e.g., the issues of natural resource dependency, alternative livelihood strategies for local households) in Zoucheng city. This chapter then moves to social capital, including background information on social capital, definitions, the strength of social capital (kin & non-kin ties), and the economics of social capital. This section provides the practical guidelines for measuring and analyzing the social capital so as to answer the first research question (i.e., Do different types of social ties lead to differential access to social capital in a small community in China?) in this study. Finally, this chapter will review the literature on the role of social capital in the sustainable livleihood and its applications, which may provide some logical and theoretical supports to establish the causal relationship between social capital and livelihood choices in simulation model. This part also provides some basic theries to answer the second research question.



3.2 Sustainable development theory as related to livelihood

A sustainable livelihood theory can recognize shocks (e.g. mining pollution) and offer a way to investigate transformations of people's livelihoods. More importantly, this approach can investigate livelihood activity choices and the factors determining these choices (Lambini & Nguyen, 2014; Nguyen et al., 2010; Wunder et al., 2014). Many previous studies (Allison & Ellis, 2001; Bebbington, 1999; Rakodi, 1999; Ellis, 2003; Barrett, et al., 2001) employed a sustainable livelihood approach to evaluate household livelihoods. With respect to the issues of natural resource dependence (shocks), current livelihoods (mining) and alternative livelihoods (rural-tourism) in Zoucheng, the sustainable development approach is an appropriate one with which capture the complexity of the situation in Zoucheng and to establish the simulation model.

The livelihood framework presents the main factors that can impact on people's livelihoods and the relationships between these components. It attempts to explain and assess important outcomes of the project in terms of impacts on stakeholders. It can be used in reducing poverty, improving the quality of life and strengthening the sustainability of poor people's livelihoods. The IDS (Institute of Development Study) defines sustainable livelihoods as follows:

A livelihood is sustainable when it can cope with and recover from stresses and shocks and maintain or enhance its capabilities and assets, while not undermining the natural resource base (Scoones, 1998, p.5).

As people's livelihoods are highly dynamic, the livelihoods approach is a way to monitor and capture these changes and increase the scope of livelihood analysis (DFID, 1999). It also attempts to involve the relationship between micro- and macro-level contexts in the process of households seeking their livelihoods (Stokes, 2002).

The most prominent framework of sustainable livelihoods as developed by the UK Department for International Development (DFID). The essential components of the

framework include: five sets of livelihood assets upon which individuals draw to build their livelihoods – financial capital (savings, credit), natural capital (mineral resources, farmland owned, crops cultivated), human capital (people's 'capabilities' in terms of their education, health, labour), physical capital (agricultural machinery, bicycle, other vehicles.) and social capital¹¹ (social networks, social resourcese); vulnerability context - the shocks, trends or other influences which affect livelihoods; policies, institutions, and processes - organizations, in both private and government sectors; processes (laws, rules, policies, incentives), which define people's livelihood strategies; and livelihood outcomes that indicate the outputs of the livelihood strategies.

A livelihood asset portfolio consists of natural, financial, human, physical, and social assets, which can complement and substitute for each other (Scoones, 1998). These assets impact policy implementations, access to organisations and regimes and how these organisations and regimes work (Allison & Horemans, 2006). On the other hand, policies, institutions and processes can either enable or hinder people's access to assets and activities. Access to livelihood assets can also be affected by some factors, referred to as the vulnerability context. In Zoucheng, these trends or shocks in vulnerability context may include issues of natural resource dependency, (air pollution, subsidence area of farmland), and reduced livelihood choices. At a household level, these factors could be shocks that impact well-being in terms of sustainable development. Understanding the "vulnerability context" is also important in the design of policies and interventions to help people's existing coping and adaptive strategies. For instance, they may include developing and supporting rural tourism activities to reduce the issues of over-reliance on mining activity. The government might implement relevant policies on cash subsidies and other favourable terms to encourage rural activity restructuring, which might further benefit the development of sustainable livelihoods. A household's livelihood choice to pursue alternative activities (e.g. tourism) is influenced by its

¹¹ By following a sustainable livelihood framework (DFID, 1999), social capital is seen as the social resources that people draw on in pursuit of their livelihood objectives. In my thesis, the definition of social capital will be broader than DFID's definition and will be discussed in social capital sections.

livelihood assets access, difficulties with specific livelihoods and relevant policies. Finally, the livelihood framework points to livelihood outcomes. As Allison & Horemans (2006) pointed out people's ability to maintain or improve their standard of living related to income and well-being and reduce their vulnerability to shocks and issues could be the result of sustainable livelihoods. Many studies (e.g. Coomer et al., 2004; Jansen et al., 2006; Ellis & Freeman, 2004; Babulo et al., 2008) have pursued similar causality paths in both empirical and conceptual studies on income diversification and livelihood choices.

3.2.1 Vulnerability context

Trends, shocks and seasonality may directly impact upon people's livelihoods and their asset status. Trends are classified under the headings of population (e.g. population density, migration), resource (e.g. loss of resources, resource stocks), economic (e.g. price trends of produced goods), technological (e.g. technologies in use in the local area), and governance (e.g. politics, voting figures) (DFID, 1999; Carney, 1998). Shocks include human health shocks, natural shocks, economic shocks, and crop/livestock health shocks (DFID, 1999). In addition, Carney (1998) includes climate (e.g. annual rainfall, climatic disasters) and conflict (e.g. legal cases pending, accounts of conflict) as shocks. Seasonality, prices, production, health, and employment opportunities are all potential influences (DFID, 1999). Carney (1998) stressed the importance of culture (e.g. unexplained differences between two areas) instead of seasonality. All of these factors contribute to the Vulnerability Context. Furthermore, Scoones (1998) arranged these components into groups (e.g. history, terms of trade, social differentiation), which applied a different viewpoint to the Vulnerability Context.

3.2.1.1 Natural resource dependency-issues for mining activities

Regarding the sustainable livelihoods approach as the conceptual framework for the thesis, thinking through issues related to activities or industries (e.g. mining) is the first step in applying a 'livelihoods approach' (Allison & Horemans, 2006). All of these

issues are related to the current situation of the natural resource-dependent economy (i.e., mining activity) in Zoucheng.

Although mining can bring many rewards for mining communities, such as industrial profits and individual income, intensive mining activity can cause environmental problems and render land unsuitable for other industrial applications such as farming and forestry. First, coal mining inevitably causes negative environmental effects. Coal is a dirty energy source because of the air pollution that occurs during mining (e.g. drilling, blasting, transportation). Many serious environmental problems such as air pollution, noise pollution, ground fissure, and water pollution arise in the coal mining areas (Shi & He, 2012).

Air pollution and land subsidence are two major environmental challenges from coal mining (Bial, et al., 2010). Mining operations include surface mining operations (drilling, blasting, movement of heavy earth-moving machinery on haul roads, transportation, collection and handling of coal and the operation of screening, sizing and segregation units) and underground mining (uncovered waste dumps and coal piles). Both of these operations can emit dust (Ghose & Majee, 2000; Bian, et al., 2010). For instance, there is normally 4-6 inches or more accumulated dust on the surface of haul roads (Ghose & Majee, 2000; Nair & Singh, 1990), which can easily become airborne during adverse conditions. In agricultural areas, the excess dust on crops may affect respiration, photosynthesis, transpiration, and allow the penetration of phytotoxic gaseous pollutants. As a result, there is decreased productivity (Farmer, 1993).

Land subsidence is another important adverse impact of mining on the environment. As Li et al., (2009) pointed out coal exploitation causes an increase in direct surface subsidence and a decline in arable land in mining areas. The issue of land subsidence can also cause environmental problems, such as plant death, surface fractures and soil loss, utility failures, damage to buildings, and drainage system failures (Bian et al., 2010). For instance, mine subsidence can directly impact on the water environment due to the connection of underground water bodies to the mined space through fractured overburden (Bian et al., 2010). Mine drainage and the disposal of mining wastes can affect water quality, including the surrounding surface or ground water, which further impact on aquatic habitats and the growth of agricultural products (Bian et al., 2010).

In recent years, enjoying a higher standard of living and requiring a good environment must be taken into account when studying tourists' behaviour (Vargas, et.al 2009). Huang, Zhou and Ali (2011) used correlation analyses to measure the relationship between tourism visits and mining activities. They found that tourism income is negatively affected by mining when they exist in the same area. In Zoucheng city, mining activity has negatively affected on the local environment (discussed in section one of Chapter Five), which demonstrates another conflict between tourism development and mining activity.

3.2.2 Institutional influences, transforming structures and processes

The analysis of institutional and organizational influences on livelihood resources and strategies is also important. Kong (2007) pointed out that government failure often leads to economic backwardness. Scoones (1998) gave three reasons to explain the importance of institutions in the policy and practice of development of sustainable livelihoods:

- Understanding institutional processes can facilitate the design of interventions to improve livelihoods outcomes.
- 2. For social processes, an institutional approach can support sustainable livelihoods.
- 3. With regard to the combination of different livelihood strategies resulting in sustainable livelihood outcomes, both formal and informal institutions and underlying rules and norms can operate at different levels (e.g. household or national level) to mediate the processes of livelihood change.

Within the livelihoods framework, the transformation of structures and processes shape people's livelihood options. In the framework, structures can be thought of as hardware,

and processes can be thought of as software. Structures include the organizations (private and public) that set and implement policy, legislation, and deliver other services linked to livelihoods in the framework. Processes such as policies, laws, and 'rules of the game' determine operations and interactions among structures and individuals (DFID, 1999).

These structures and processes affect livelihoods in two major ways. First, they play a critical role in determining who can gain access to which types of capital and the effective value of that capital. Second, they help define the livelihood strategies, the decision-making bodies and the terms of exchange between different types of assets with regard to people's asset status (Carney, 1998). So, policies, institutions and processes might affect the options available to people. With these options, people can resist shocks and improve their livelihoods. Thus policy and institutional analysis have an important place in the livelihoods framework.

3.2.3 Livelihood strategies and outcomes

Livelihood strategies are the range and combination of activities and choices that people pursue in order to achieve their livelihood objectives (DFID, 1999). Livelihood strategies show enormous diversity at every level (e.g. within households and over time), across sectors and within geographic areas. The links between urban and rural centres possibly impact on people's decision-making and asset usage as well. Thus, there is a dynamic process in which people undertake a combination of activities to meet their goals at different times. A common example is migration (e.g. seasonal migration).

There are many ways to categorise livelihoods. For instance, livelihood strategies can be divided into three clusters in the sustainable livelihoods framework: agricultural intensification (productivity is increased by capital investment or labour inputs)/extensification (more land fit to be cultivated); livelihood diversification such as developing a wide income-earning portfolio; and migration (Scoones, 1998). In many areas in developing countries, rural households normally choose nonfarm activities to diversify their livelihoods and income-earning portfolio. In developing countries, nonfarm activities make up about 35-50% of rural residents' total income (Haggblade, Hazell, & Reardon, Feb 2009; Knight & Song, 2003). For instance, total nonfarm earnings accounted for 51% of rural income in Asia (Haggblade, et al., Feb, 2009). In China, non-agricultural income also plays an important role for rural households. According to Zhao and Barry's rural household survey in China, about two-thirds of farm households obtained their income from at least two sources such as agricultural crops and nonagricultural activities (Zhao and Barry, 2014). On average, nonfarm income has comprised more than 50% of rural incomes in rural households (Zhao and Barry, 2014). Davis et al. (2007) analyzed rural income-generating activities by using a constructed cross-country database. They selected the rural samples and analyzed the income-generating activities of rural households. In all of the selected countries, the ranges of non-agricultural income were from 20 to 75% of total rural income, and 47% on average. Yang (2009) showed that the labour shift from cropping to non-agricultural activities has significantly contributed to income increase and income stability in rural China. Off-farm income is by far the most remunerative for rural households, allowing them to increase both the level and the stability of household income (Demurger et al., 2010; Yang, 2009; Ellis, 2000; Hoogeveen, 2001). Thus, there is an increasing tendency for rural households to generate more livelihood choices (i.e., non-farm activities).

The combination of activities and strategies also depends on health conditions, changes in dependency ratios and other factors over different time-scales. The resource endowments available and the level of risk relating to alternative options may influence the degree of diversification. Therefore, in the process of livelihood strategy analysis, a key step is to identify what livelihood resources are required for different strategies. The different types of 'capital' such as basic materials and social assets can influence livelihood strategies. Livelihood outcomes are required to be a component of the framework as they are considered to be the achievements or outputs of livelihood strategies (DFID, 1999). Considering the livelihoods and the sustainability dimensions, Scoones (1998) defines five key elements to assess sustainable livelihood outcomes:

- Creation of working days a particular combination of livelihood strategies and the proportion of the population available for gainful employment.
- 2) Poverty reduction an absolute poverty line.
- Capabilities and well-being sustenance and intrinsic elements such as happiness, stress and security.
- 4) Vulnerability, livelihood adaptation and resilience this plays a key role in the concept of sustainable livelihoods. The required analysis factor is the history of responses to different types of shock, such as tolerance or avoidance.
- 5) Natural resource base sustainability the ability of the natural resource-based system to maintain productivity to yield benefits for livelihoods. However, it is very hard to measure the sustainability of natural resources.

When assessing sustainable livelihood outcomes, the critical issue is whether the livelihood portfolios (the combination of activities that are pursued) produce positive or negative effects on the range of sustainable livelihood outcome indicators. This study is especially interested in natural resource base sustainability (e.g. agricultural and tourism resources), vulnerability (e.g. over-reliance on non-renewable resources such as mineral resources), and creation of jobs (e.g. more employment opportunities by increasing the competitive advantage¹² of alternative livelihoods) to assess sustainable livelihood outcomes in Zoucheng.

¹² The term competitive advantage is the ability to gain resources and attributes or make intensive use of factors to perform at a superior level to others in the same industry, area, or market (Deardorff, 2004). In general, superiority in production resources and superior performance outcomes reflect competitive advantage (as cited in Lau 2002).

3.2.3.1 Over-reliance one particular economic activity versus diverse livelihood strategies

Over-reliance on mining revenue for household income may limit the development of other potential alternatives (such as tourism). The lack of diversity in livelihoods may increase households' vulnerability to shocks and risks (Soriano, 2007). For instance, most mine labourers are categorized as farmers in China. However, they work most of their time and derive most of their income from mining (Gunson & Jian, 2001). A problem is that when mining operations are shut down, these miners are seen as being able to 'just return to their farms' (Gunson & Jian, 2001).

Furthermore, there is an ongoing large surplus of labour in rural areas in China (Knight, et al., 2011; Golley & Meng, 2011). Some scholars (Kwan 2009; Guo 2007) have estimated the total amount of surplus labour at between 110 million and 180 million, equivalent to between one-third and two-thirds of the total rural labour force in 2000. It is a big challenge for the Chinese government to shift this labour pool to the nonagricultural sector or urban areas, particularly in view of the large number of unemployed workers already in urban areas (China Daily 2006; Yang & Wang, 2006). Surplus labour together with an unsustainable industry may potentially increase the vulnerability of rural household livelihoods in mining areas.

Some non-agricultural activities such as service sectors and manufacturing could absorb rural workers and reduce the rural labour surplus in China (Wang, et al., 2011; Mai & Peng, 2012), which might diversify livelihood choices for rural labourers, particularly with regard to the role of rural tourism in China. Rural tourism has the potential to alleviate poverty and facilitate employment in rural regions in China (Zou, et al., 2014). According to the Centre for Agriculture Research in the Ministry of Agriculture of China (as cited in Zou, et al., 2014 and He & Qi, 2012), each direct employment position in rural tourism could lead to five indirect job opportunities. Developing rural tourism will broaden rural industrial development and promote rural economic diversification such as extending tourism into traditional agriculture, the restaurant industry, commerce, service industries, rural transportation and other relevant industries (He & Qi, 2012).

In many developing countries, some rural households diversify their livelihood strategies, while others pursue one or few activities. For rural livelihoods, the reduction of poverty hinges on the capability of the poor to diversify and combine different livelihood strategies and sources (Soriano, 2007). Rural households have incentives to find alternative income sources to secure their livelihoods (Demurger, et al., 2010). In addition, a household's reliance on a particular economic activity in general, and on the level of environmental resource use in particular, may result in different livelihood outcomes (wellbeing, income, etc.) by considering the household's demographic and economic characteristics (Babulo et.al, 2008). Diverse livelihood systems are related to resilience, flexibility and stability, which allow rural livelihoods to become less vulnerable to shocks and risks (Ellis, 2000; 1999; Soriano, 2007). In this regard, it is important to understand the influences on livelihood outcomes of a household's activity choice so as to either develop or change the targeted policies.

3.2.3.2 Alternative livelihood strategy

Rural tourism normally occurs in non-urban areas where rural residents mainly engage in agricultural activities attached to rural resources such as land (Demoi, 1991). With a big rural population and rich tourism resources, rural tourism could be a new development for Zoucheng city. The rural tourism concept involves many interpretations. It is a multi-faceted activity rather than farm-based tourism only, and includes activities in nature, adventure, farms, health, arts, sports, education, and heritage, all taking place in the countryside (as cited in MacDonald & Jolliffe, 2003, Keyim, et al., 2005; Bramwell & Lane, 1994; Reichel et al., 2000; He & Li, 2003).

Chinese people traditionally enjoy sitting calmly and chatting or playing games with friends or relatives (He & Chen, 2012), which potentially increases the rural tourism market such as walking around tourism sites, enjoying the beauty of nature, and eating
rural home-made meals with friends. By the end of 2010, there were about 18,000 largescale leisure agriculture parks and more than 1.5 million agritainment households in China (Wang, et al., 2013). According to Su (2011), rural tourism attractions entertain more than 300 million tourists and create more than 40 billion RMB¹³ in revenue per year in China. In addition, about 70% of urban residents prefer rural tourism when planning trips, which could create about 60 million tourists for each long holiday such as National Day and Chinese New Year (Shao, 2007).

In many rural regions, researchers have been concerned at the trade-off between tourism and agriculture and debates over the allocation of resources between the two sectors. Agriculture provides the natural resources for tourism development; meanwhile tourism encourages the agricultural production of high value crops. There are lots of beneficial effects from tourism on agriculture, such as providing higher wages and resource values for farmers, infrastructure improvements in rural areas, introducing high-value agricultural products into the market system, stimulating other services in agriculture, and increased commercialization of agriculture (Cox, Fox and Bowen, 1995; Zhang, 2012).

As previously stated, rural dwellers constitute nearly 65% of Zoucheng city's population and there are rich tourism resources in the area. Thus, a series of rural tourism activities could be created by the local government so as to generate income to address the problems of poverty in rural areas and create sustainable development for local communities. According to MacDonald & Jolliffe (2003) and Ying & Zhou (2007), cultural and historic heritage elements and rural tourism could be integrated as a new concept 'cultural rural tourism.' This new form could also be owned and operated by individual farmers or rural households, and could constitute a kind of privately-owned small enterprise for millions of Chinese farm families (Zou, 2005; MacDonald & Jolliffe, 2003).

¹³ RMB (renminbi) is an abbreviation of the official currency of the People's Republic of China.

In rural China, about 1.3 million farming families operated rural tourism enterprises by 2009 (as cited in Su, 2011). Normally, this features having rustic meals, experiencing rural lifestyles (e.g. doing hard farming work, participating in home-made meals with accommodation services such as farmhouse provided), and other amusements such as special adventure opportunities and cultural attractions provided by farming families (Ying & Zhou, 2007).

One typical type of rural tourism in China is known as "Nong Jia Le", which means, "enjoying farm home stay." Rural people normally use their own houses to cater for and accommodate tourists and provide rural cultural and natural environments for them (Fu, 2009; Su, 2011; Zeng, 2008). All of these services and amusements particularly suit tourists coming from urban areas for weekends and holidays, even if only for a few hours (MacDonald & Jolliffe, 2003).

For the operation of rural tourism in China, Zheng & Zhong (2004), He (2005) and Wang & Fang (2008) introduced six inter-related models for rural tourism development. All the models are based on levels of off-farm employment opportunities, operational scale and scope, tourism corporations specializing in the development and management of rural tourism, rural tourism associations representing the local community authority, and the role of government in rural tourism development.

- 1. Household-run small businesses are represented by the individual farming family. For instance, the family's work in meal cooking and bed making are basic and important for the business to survive. It is normally the first step in establishing rural tourism.
- 2. Individual farmsteads involve individuals specializing in rural tourism. The operation of this model is responsible for running and transforming tourism products and facilities into a tourist-designated spot. It can be developed as an autonomous business entity and encourage unemployed labour forces nearby to join in the business of rural tourism.
- 3. Farmer family plus farmer family may help to adjust and improve the

agricultural economy of rural communities via rural tourism development in China.

- 4. Corporation plus farmers involves setting up a tourism corporation specializing in the development and management of a village's rural tourism, in which the investors are outside the villages. The corporation can lease land and other resources from the farmer and provide technical guidance and the necessary training.
- 5. Corporation plus community plus farmers further develops the model "corporation plus farmers." In this model, all the owners of family-run businesses take part in a rural tourism association (e.g. village committee in most cases) and make the association function as the board of directors of the corporation. The corporation is trusted by the association to run and operate the village's tourism business. The farmers are employed and trained by the corporation.
- 6. Government plus corporation plus farmers involves large-scale tourism projects. The government, local community authority, and outside investors can form a collectively-owned rural enterprise to employ and pay salaries and bonuses to farmers.

Regarding local rural tourism development, models 1, 2 and 3 are all possible as main business operation models for rural tourism in Zoucheng.

Localization is essential to the sustainable development of rural tourism in China (Sharpley, 2003). It is important to coordinate the local people involved in rural tourism planning and development processes (Li, 2000). The local community can share the benefit from rural tourism development. Zou, et al., (2014) illustrates using the example of three villages in Beijing, where a community-driven development method is the proposed model. This approach can ensure the community's high level of participation and manage the costs under an effective supervision and evaluation system. On the other hand, government involvement also plays an important role in rural tourism

development in China by providing funds and developing local tourism products (Zou, et al., 2014). This study focuses on models involving a high level of community participation and government support. With more local participants involved, social networks might be an efficient option for further developing rural tourism.

With local mineral and tourism resources, planning in regional development policies is designed to encourage local endogenous economic growth, to reduce socioeconomic inequalities, and to ensure natural resource stability in local areas. Sharpley (2003) also claims that developing rural tourism should address five concerns: encouraging production of local crafts; establishing a supply chain of local products or services; satisfying the needs of the local community; ensuring maximum retention of profits locally; and ensuring that development is within the capacity of the local society and environment. Based on McIntosh's comments (1977) and Sharpley (2003) concerning a community framework with goals for tourism development, local tourism planning should be based on the following goals:

- To create more job opportunities for rural households. For instance, policies may create more jobs in related transport, processing and marketing of agricultural produce and relevant services by cooperation between local tourism and agriculture industries.
- To reduce negative effects on the local environment with economic development.
- To reduce the development gap between advantageous industries and disadvantageous industries.
- 4. To encourage coordination between public and private efforts.
- 5. To maintain and protect cultural heritage, and the traditional pattern of human activities.
- 6. To promote tourism in a small resource-based community with the greatest local benefit.

The simulation model presented in Chapters Seven and Eight will concentrate on the first three points as important planning policies to diversify livelihoods and provide sustainable economic development for the local community.

3.3 Social capital

This section focus on providing some background information, definitions of social capital, the strength of social ties (kin & non-kin ties), and some economics of social capital because of the significance of social capital in this study.

3.3.1 Introducing social capital

Individuals can access relevant resources through the use of personal networks and connections (possessed social capital) or memberships in groups (structural social capital) to secure benefits (Sobel, March 2002). Possessed social capital specifically aims at embedding network resources in personal networks (Zhao, 2002). Structural social capital involves the behavioural manifestations of social capital, which is generally subsumed as memberships in formal associations (Harpham, Grant, & Thomas, 2002; Yip et al., 2007).

Social networks consist of family, friendships, formal (e.g. work-based) and informal community relationships (Gilchrist & Kyprianou, 2011). Many studies (e.g. Adler & Kwon, 2002; Wu & Choi, 2004; Xiao & Tsui, 2007; and Hitt et al., 2002) indicate the special relationships between relevant resources (e.g. access to financial resources, information etc) and social networks. Appropriate information combined with assets can help to motivate rural households to use and improve their livelihoods and expand their asset base (Soriano, 2007). The types of social relations, the quality of relationships, and the configuration of ties in a network can influence a firm's ability to obtain loans as well as the cost of borrowing (Uzzi, Aug 1999; 1997; Portes & Sensenbrener, 1993; Romo & Schwartz 1995; Uzzi 1997).

Membership of a group or organization is viewed as a form of *guanxi*, which might benefit individuals' careers or livelihoods by increasing their own structural social capital (Bian, Shu, & Logan, 2001; Knight & Yueh, 2008). For example, some formal groups not only have complete control of organs of State, but also provide contacts, information and influence for their members (Bian et al., 2001; Knight & Yueh, 2008). Putman (2000) emphasized that participation in multiple groups creates the rich network of ties, which may increase monitoring, information flows, and mutual respect.

Much of the literature relevant to China (e.g. Yueh & Knight, 2008; Liu, 2003; Lam, 2003; Yip et al., 2007) has used membership of the Communist Party or other formal associations as a measure of individual social capital in China. However, formal associations or networks may only be a small factor in the production of social capital (Lyon, 2000). First, party membership is not simply a matter of personal choice; members are selected and vetted by the Party secretary or Party organization (Bian, Shu, & Logan, 2001; Yueh & Knight, 2008). Therefore, the recruitment requirements may limit studies to use memberships of the Communist Party to represent individual or household's social capital in China. Second, some formal organisations have different degrees of power in the administrative system in China. For instance, the grassroots organizations of the Labor Union remain incapable of representing workers in China (Chen, 2009). The reason is the union bureaucracy's operation and the reliance of its power upon its formal government status (Chen, 2009). The All-China Federation of Trade Unions or its regional branches can play an active role in labour dispute settlements compared to grassroots organizations in China (Chen, 2009).

Because of the limitations of group memberships, the social capital in this study has focused on broader concepts in personal connections and networks. In this thesis, all of the definitions and measurements of social capital are based on personal connections and networks. Personal connections and networks are more common and easier to access in Chinese communities, compared to the recruitment requirements in formal organisations.

3.3.2 The definitions of social capital

As one of the important livelihood assets, social capital is easily overlooked as an influencing factor, since it cannot be directly observed compared to other assets (Bebbington, 1999). Social capital is probably the least tangible of the five capital assets, and is notoriously difficult to understand, identify and assess (Bebbington, 1999). In Section 1.3, each research article may define social capital differently. For instance, Nguyen et al., (2015) used the number of mobile phones in each household to represent social networks; Ferrary (2003) & Mayoux (2001) used financial credits; and Habibah et al., (2010) defined social capital based on local involvement in conservation. Lin (2001) paid attention to social networks. Given the difficulties of defining social capital and the different ways it is used in the literature, caution must be used before attributing too much weight to empirical results linking social capital and livelihoods (Bebbington, 1999). In other words, it is necessary to indicate the definition and measurement method on which indicators of social capital are based.

Over the last thirty years, many scholars have contributed to the definition of social capital (Coleman, 1990, Burt, 1992; Portes, 1998; Bourdieu, 1983/1986; Putnam, 1993b, 1995 among others). Although many of these definitions were constructed from different data and focused on either individual or collective levels of social capital, the general definition is consistent: "Investment in social relations with expected returns in the marketplace." (Lin, 2001, p. 19). Other similar definitions from Coleman (1990) and Putnam (1993b) portray social capital as a web of cooperative relationships or connections to facilitate resolution of some confronting collective action problems among citizens. Lin (2001) also pointed out that social capital is the resources embedded in one's social networks, which can be accessed or mobilized through ties in the networks. In other words, social capital is about networks plus resources. Paldam (2000), Dasgupta (2005), Stiglitz (1999), and Sobel (2002) express similar views in which social capital can be recognized as a collection of social networks.

Research on social networks has thus produced two conceptualisations to understand social capital. Coleman (1998) and Putnam (1993) proposed social capital as the emergent properties of social networks (e.g. trust and reciprocity). These "social networks" provide the social control and solidarity to faciliate "coordinated actions," pursue "shared objectives," and enhance collective well-being. Lin (1982, 2001a) conceptualises another view of social capital. Social capital is the resource instrument for social actions, which are accessed through interpersonal ties and distributed in a hierarchical structure. As Franke (2005) pointed out social networks are recognized as an important element of social capital and provide the resources accessed for social capital (Portes, 1998; Burt, 1997; and Lin 1999b). Dufhues, Buchenrieder, and Fischer (2006) defined useful social capital in terms of its relevant network sources (not in terms of outcomes such as trust).

In terms of social ties (Lin 2001a), people in high-ranking positions are likely to access better resources for instrumental social actions than those in the lower ranks. Furthermore, social network extensity (e.g. social connections in diverse places) increases the variety of available resources. For instance, miners' social networks in Zoucheng might exist among various objective strata within local society. Among these social resources, miners might use different kinds of relations with different ranking positions to pursue their benefits. With regard to this research on the hierarchical structure of social ties, I adopt Lin's conceptualisation. The social capital definition in this thesis is derived from Lin (2001) to investigate social resources in a small community in China.

3.3.3 Strength of social ties

As important sources of social capital, role relations have been widely used to measure the strength of social ties (non-kin ties and kin ties). Lin et al. (1981), Lin and Dumin (1986), Marsden and Hurlbert (1988), and Bian (1997), have similar views regarding the nature of Chinese tie-strengths. That is, social ties between relatives are strong, while social ties between acquaintances, neighbours and peer workers are weaker. Furthermore, the strength of the relationships leads to different values of social resources. Strong (intimate, usually kin) ties imply obligation and trust to provide reliable information in one's own interest (Granovetter, 2005; Bian, 1997). On the other hand, weak ties (usually non-kin ties) may serve as bridges that link individuals to diverse social milieu and provide other valued information and resources not present in their own immediate social circle (Granovetter, 1995; Lin 2001a; Burt, 2001).

Globally different types of social ties can lead to differential access to social capital. Strong (intimate, usually kin) ties can be important for realizing these opportunities, by providing relevant information to preserve one's interests and ensuring collective action (Granovetter, 2005). For example, kinship serves to facilitate access to information where formal institutions, channels and processes limit legitimate access to resources (Boxman, Graaf and Flap, 1991). However, strong ties may be critical under institutional uncertainties or constraints to the availability of other resources (Rus, 1995; Bian 1997). For instance, strong ties (kin ties) have negative effects on general social capital in some large metropolitan cities in China (Lin, 2001; 2004). It may be that in these contexts the more homogeneous networks (e.g. kin ties) reduce the effect of networking as a way of accessing better social capital (Lin, 2004). On the other hand, weak ties (usually non-kin ties) may extend and strengthen general social capital by bridging social boundaries, providing diverse information and gaining access to opportunities and resources (Granovetter, 1973, 1974), which also applies in Chinese society (Lin, 2001; Ruan, 1998).

The literature discussed above indicate some causalities between social ties and social capital, which might provide theoretical support to answer the first research question (i.e., do different types of social ties lead to differential access to social capital in a small community in China?). However, they didn't concern themselves with the small cities in China. So, it is necessary to use a regression model to analyse the factors associated with access to social capital in a county-level city (see Chapter Six).

3.3.4 Some economics of social capital

There is a debate on the classification of social capital as either public goods (non-rival and non-excludable) or private goods (rival and excludable) (Leana & Van Buren, 1999; Adhikari & Goldey, 2009; Putnan, 1998; Cheng & Luo, 2011). From a macro-view, social capital is more likely to be a public good. Social network resources are embedded among society and wouldn't be reduced as one actor accesses more. Thus, public-good social capital brings resources to the whole society or group (Keana & Van Buren, 1999). From a micro-view, social capital becomes a private good. Individual social resources have a strong excludable characteristic. Private-good social capital benefits the interests of the individual (Keana & Van Buren, 1999). My study focuses on private goods only.

Establishing a relationship and joining a social network are typical and common forms of social capital investment (Glaeser et al., 2002). If social capital is a private good, social networks and different social ties (kin & non-kin) may provide excludable and rival relations to help individual to expand personal social resources. These relationships might be regarded as a kind of investment in a person's social capital.

Social capital has both a positive and negative side. For the positive side, social capital has the functions of ensuring collaboration among communities to improve the efficiency of social operations and lower their cos (Dowla, 2006; Portes & Landolt, 1996; DFID, 1999; Kah et al., 2005). People can receive assistance from social resources to support their livleihoods (DFID, 1999). In the simulation model of this thesis (see Chapter Seven), people might use social networks to access information to operate rural tourism business and increase their expected tourism income. This is the productive side of social capital. The negative sides of social capital (e.g., drawbacks of personal social ties) may reinforce inequality, reject outsiders (i.e., protectionism), and break community trust (Field, 2003; Gargiulo & Benassi, 1999; Putzel, 1997). For instance, people with social capital can collude and increase the price of products in a

market¹⁴. This is counterproductive. Thus, supporting institutions, processes (e.g., relevant policies and laws), supervision from below and transparency are all possible methods to reduce these drawbacks. This is also one of the important reasons for the simulation model in this thesis that involved institutions and processes to support people accessing social networks and relevant resources.

3.4 Role of social capital

The next two sections introduce relevant social capital information related to the second research question presented in Chapter One, which may provide some basic theories to establish the causal links between social capital and livelihood choices for the simulation model.

3.4.1 Role of social capital in the sustainable livelihood framework

Social capital in the form of social networks has gained much attention in terms of livelihood frameworks. Figure 3.2 shows the inter-linkage among livelihood capital assets for household, social capital and networks. The different types of household asset portfolios (natural, financial, human, physical, and social) can complement and substitute each other (Robison et al., 2002; Scoones, 1998). Bebbington (1999) and Moser (1998) state that social capital plays an important role in asset portfolios. Social capital capital (Fafchamps & Minten, 2002; Annen, 2001). On the other hand, social capital, made up of structural and embedded network resources (accessible and mobilised), plays a pivotal role in the capacity of households to access material resources and thus, livelihood strategies (Foley & Edwards, 1999; Moser 1998). However, since social capital is rooted in social networks, comprised of actors, resource flow, exchange relations, exchange opportunities and structured association (Cook et al., 1983), other

¹⁴ It wouldn't be a problem in the simulation model of this thesis. Due to lots of business operators in simulation model, tourism business can compete each other. So, social capital is still postivie because negative sides are small.

factors such as income, poverty, or education level may influence the formation and stability of networks (De Weerdt, 2002; Narayan & Pritchett, 1999).



Figure 3.2. Assets, social capital, and networks

Embedded network resources

Note: * Based on Scoones (1998); ** based on Foley & Edwards (1999) and Lin (1999a); *** based on Cook et al. (1983)

Source: Dufhues, Buchenrieder, & Fischer, 2006, p.11.

In the access of households to material resources, social capital plays a special role in determining different livelihood strategies. For instance, access to productive resources or scarce resources depends not only on material wealth but also on engaging in relationships with other actors, or on special social networks in social institutions, or with cadres in some developing countries (Bebbington, 1999; Dalton & Ong, 2005; Winkels & Adger, 2002; and Berry, 1989). Some forms of social capital and social structures may facilitate group-based approaches to lending (DFID, 1999), which may become the financial supports in alternative livelihood development. In addition, access to livelihood assets such as social capital is an important factor in determining the choice of a household's livelihood strategy. As a feature of social structure in Chinese society, social capital generally can increase the achievement of certain economic and social ends (Coleman, 1988; Lin & Si, 2010). In the context of the social network theory, people with better interpersonal connections tend to have better careers, get more frequent promotions, and earn more income (Burt, 1997; Granovetter, 1985). The

livelihood approach states that households' available assets can affect the type of activity undertaken and the amount of income earned (Barrett et al., 2005; Brown et al., 2006). For instance, research on social networks has demonstrated the important role of social capital in the labour market (Lu, et al., 2013; Amuedo-Dorantes & Mundra, 2007; Aguilera & Massey, 2003; Sanders et al, 2002; Fong & Ooka, 2002). Therefore, knowing the "social game" is crucial to understanding the functioning of politics, business and other important areas in developing countries (Berry, 1989).

3.4.2 Applications of social capital: social networks and social resources

Understanding social networks and relationships with the broader concept of social capital is very important in understanding the ability to access key resources such as information and financial capital (Lyon, 2000). Social networks provide patterns of social interaction, social cues, and social identities, which are mechanisms to connect the individual to society (Inglehart & Baker 2000; Inkeles & Smith 1974; Inkeles 1983). Blair and Carroll (2008) noted that social capital contributes to economic development strategies. In China, it is important to maintain and cultivate the social networks, which function as the 'operational code' to get things done (Kipnis, 1997; Yan, 1996; Yang, 1994; Oi 1999). Social capital can be a source of skills, information, technologies, loan collateral, and contacts among neighbourhood dwellers (Reinke 1998 & Blair and Carroll 2008). With social contacts improving, inside information and connections can increase promotions and earnings (Knight & Yueh 2008; Lee 1998), although more effort is required to support the growing social network. With regard to the benefits generated from social relationships, many studies (Adler & Kwon, 2002; Wu & Choi, 2004; Xiao & Tsui, 2007 and Hitt et al., 2002) have used access to information and relevant resources to indicate the web of social relations.

A social network is considered as an information channel, which provides powerful, efficient means of access to information and resources (Coleman, 1988; Lin & Si 2010; Burt, 1992; Christakis & Flowler, 2010). There is a broad area in which social networks can increase information access. For instance, the information transfer between

recruiting firms and work-seekers could be improved through the use of contacts in an imperfect labour market (Knight & Song 2005; Zhao 2002). In the business area, an effective way of helping to find clients, suppliers, and investors is through personal networks of relationships to get relevant resources and information (Batijargal 2006 & 2007).

Social capital is expected to pay off in many ways by enhancing the efficiency of the labour and capital markets, improving information access, simplifying application procedures to meet people's credit needs, improving borrowers' access to formal credit, slashing administrative costs (e.g. lowering the costs of monitoring and enforcing existing rules and norms), and motivating repayment (e.g. introducing the relevant financial discipline by using peer pressure) etc. (Dowla, 2006; Kah et al., 2005; Rankin, 2002; Ostrom, 1992). Some social networks, communities, or groups are considered as support mechanism which help in developing social relations, improving information access in the market, making more rigorous business efforts, getting advice from successful group members, and trainings (Dowla, 2006).

Social capital can play an important role in tourism as well. In rural tourism communities, the participation of community residents plays a determining role in the process of community-driven rural tourism development (Zou et al., 2014). Many prior studies (Sharpley, 2003; Taylor, 1995; Tang, 1998; Liu, 2000; Zou, et al, 2014; Zhou & Liu, 2008; Su & Wall, 2014) claim that localization or community participation is essential to the healthy and sustainable development of rural tourism. The cooperation, support and participation of residents cannot ignore social capital in the process of establishing a sustainable tourism industry. Some studies (Nunkoo & Ramkisson, 2012; Mbaiwa & Stronza, 2010; Choi & Murray, 2010; Gursoy et al., 2010) suggest that social capital is likely to influence residents' attitudes and collective actions with regard to tourism development. Social capital is a necessary condition for sustainable tourism development (Park et al., 2012; Macbeth et al., 2004). Investigations in different contexts and diverse countries clearly show that high levels of social capital in

communities act collectively to produce superior development results over multiple sectors and diverse activities (Krishna, 2001; Grootaert, 1998; & Putnam, 1993a). Community tourism approaches cannot avoid the influence of guanxi in China (Li et al., 2007), especially in the early stages of tourism development (Jamal & Getz, 1995).

For theories and frameworks about sustainable livelihood development and the applications of social capital in livleihoods (e.g., information access, community participantion in rural tourism, etc.) discussed earlier, an analytical framework and causal linkages between social capital and livelihood strategies in terms of households' sustainable livelihoods in Zoucheng city can be generated as a theoritical basis for simulation model in Chapter Seven. The general process of developing a sustainable alternative livelihood for local residents is supported by social capital. Given the natural landscape and tourist resources in Zoucheng city, rural tourism is expected to be an alternative livelihood choice for city dwellers. Rural households could take advantage of more tourists in the local city by selling more agricultural products, which may be one of their income sources. By using their social capital networks, residents may have more opportunities to access relevant information to operate their own rural tourism business. The activities chosen by mining household members generate livelihood outcomes such as income diversification, and environmental and economic sustainability. This framework provides a conceptual depiction of the dynamics and endogenous interdependence for a simulation model (see Chapters Seven and Eight).

CHAPTER FOUR: RESEARCH METHODS

This chapter will firstly introduce the research design and data collection process for the whole study. Then, it discusses the analytical research framework design. After that, data entry, calculation and summary of collected data are introduced. Finally, two research methods (the regression model and the system dynamics model) are discussed with respect to answering the two research questions in Chapter One. This chapter pays particular attention to social capital, because social capital cannot be directly measured. The research methods mainly include the survey data collection, descriptive data, and the specific mathematical models that are used for regression analysis in Chapter Six, along with the system dynamics model that is used in Chapters Seven and Eight.

4.1 Research Design

The background data and information were directly collected from the Zoucheng government, Zoucheng Tourist Administration, the Bureau of Statistics of Zoucheng City, the Bureau of Agriculture of Zoucheng City, the relevant mining companies, and local official sites.

The household survey in this study focuses on the micro level (i.e., individuals and households). It collected data about social relationships and social resources in the communities most affected by local mining activities. In addition, relevant household characteristics such as location (rural or urban), participants' gender, and livelihood assets such as educational level and household expenditure were also included in this survey. This survey design was used to establish a composite variable for social capital, and a deeper analysis of the impacts and the associated factors with social capital. The questionnaire was completed in 2010 and obtained ethics approval from the Ethics Committees of the Management School of The University of Waikato in same year.

The following sections will provide more detail about this household survey.

4.2 Data collection for household survey

All of the data were collected in Zoucheng city, Shandong Province, China. The survey was a household survey, which was conducted among employees in local mining and relevant industries from April 2010 to July 2010. Participants were identified and recruited using a modified snowball method. The collection method for the household survey is described in the following paragraphs.

4.2.1 Sampling frame

There are over 100,000 people employed in local mining or related companies in Zoucheng city. This survey involved 228 participants in local mines and related departments (e.g. electricity, and the corporate headquarters of the largest local mining company Yanzhou Coal Mining Group).

Although the literature does not provide a consistent answer to the minimum sample size in a multiple regression analysis, a combination of methods is likely to be most valuable in many situations. According to the Rules of Thumb¹⁵ for minimum smaple size and methods for calculating effect sizes in multiple regression (Knofcznski & Mundfrom, 2007; Algina & Olejnik, 2003), about 200 samples and the nine predictor variables used in regression model in Chapter Six of this thesis are technically acceptable and provide a good prediction level in multiple regression analysis.

The local mines involved included Baodian, DongTan, Nantuan, Beisu, Liyan, Taiping, and Henghe. The survey included other relevant employees in the total sample since mining-related employees work in multiple fields in the mining industry such as coal

¹⁵ As Harris (1985) and Wampold and Freund (1987) have pointed out, historically the most common rule seems to have been that the ratio of subjects to predictors be at least 10:1. Green (1991)

recommended that a reasonable rule of thumb for minimum sample size is 104+number of predictors. Tabachnick and Fidell (1989) suggested that the shample size should be at least 5 \times

number of predictors. Comb (2010) noted that a "one-size fits all" formula of having a sample of at least 100 subjects is a widespread assumption in the applied sciences, irrespective of the value of the predictor variables in a regression model.

digging, transportation, marketing, and expansions of the industrial chain (e.g. electricity, chemical industries).

Table 4.1 summarizes the relevant mines or departments that were involved in this survey. For instance, Electricity and Headquarters are considered as employment expansions of the local mining industrial chain. For each targeted mine or department the numbers of participants who completed the survey are listed.

Targeted	Number of participants
BaoDian	31
DongTan	22
NanTun	39
BeiSu	24
LiYan	21
TaiPing	26
HengHe	27
Electricity	21
Headquarters	17

Table 4.1: Participant distribution among local relevant mining companies

4.2.2 Sample selection and collection methods for household surveys

The household survey conducted in this study firstly used a form of snowball recruitment to help the author to access leaders in local mining groups. The snowball sampling approach is also called link-tracing or chain-referral (Illenberger & Flotterod, 2012). It is initialized by sampling a set of respondents (referred to as seeds) who report relevant alters (members of a person's networks or social contacts) or other subjects without replacement (Illenberger & Flotterod, 2012; Goodman, 1961).

These local leaders can help the author to access their networks or social contacts that can provide a list of miners for research survey of this thesis. The main issue from the snowball sampling approach is a homogeneous cluster may exist among the participants in the sample (Illenberger & Flotterod, 2012; Atkinson & Flint, 2001; Goodman, 1961; Thompson & Frank, 2000; Thompson, 2006; Chow & Thompson, 2003). However, this study didn't use the collected data from leaders (considered as pretesting data) due to

they may have close individual connection. The participants used in this thesis were selected randomly from the list of miners provided by leaders in local mining groups.

Data were collected either through face-to-face interviews or a self-administered survey. The household survey was started by contacting group leaders or middle-level leadership who were working for the local mining company. Then, these people were interviewed by the author to pretest the survey questions as research assistants. Thus, there was at least one research assistant in each company. After that, the author randomly selected the participants from the list of miners provided by local leaderships. The most common approach was to spread the survey in a meeting room or in a place at a time arranged by the author, group leader and middle-level leadership. In the survey process, if participants could not understand the questions, they could ask the author or the research assistant. However, there were some participants who took the questionnaires home and returned them the next day as they could not attend the organised survey at that time. Some other individuals were interviewed by the author, as they finished work.

4.2.3 Questionnaire design

The household survey¹⁶ was designed to include questions relevant to social capital and general background information.

4.2.3.1 Social capital

a) Measurement of individual social capital

To investigate the social capital concept, many researchers (Van der Gaag & Snijders, 2005; Lin, 2001; McCallister & Fischer, 1978) have used access-type social capital data to measure potential social resources in social networks. However, this type of measurement is in its formative stages and there is a lack of consensus on the meaning

¹⁶ See Appendix D.

of the term (i.e., the connection between measurements of social capital and the theoretical definition of social capital) (Dufhues, Buchenrieder, & Fischer, 2006). Adam and Roncevic (2003) noted that it is important to be cautious about its use in parametric measurement. Nevertheless, according to the definition used in this thesis, social capital is the resources embedded in one's social networks and social relationships. In other words, social capital is rooted in social relations and social networks (Lin, 2001b).

Most studies tend to examine formal or semiformal social networks (e.g. membership of clubs, organizations or associations). Bourdieu and Wacquant (1992) believed that the sum of the resources attainable through institutionalized relations would be best operationalized for social capital. But this kind of social resource does not represent the entire strength of social capital if this information is not accompanied by what people do as members and does not consider informal networks (friends, workmates, school class networks) (Schuller et al., 2000). In other words, social capital cannot be simply measured by counting the number of registered groups in a community, if the group nature and quality are not considered during the process (Lyon, 2000; DFID, 1999). Sobel (2002) stated that the extent of an individual's social resources depends on three factors: (1) a person's connections through individual or common group membership; (2) the strength of these connections; (3) available resources within their connections. Social networks accessed, and the resources embedded in them, both constitute social capital.

Measurement can focus on two important dimensions at the individual level of social capital: social resources; and ways in which these can be made available to individuals (Lin, 2001; Flap, 2002). The range of social resources, their variety, the best possible resources, and the composition of resources among social ties or networks can be used to measure social networks (Dufhues, Buchenrieder, & Fischer, 2006).

Three important measurement instruments are available to collect data about social resources possessed by network members. There are the Name Generator (McCallister

& Fischer, 1978), Resources Generator (Snijders, 1999; Van der Gaag & Snijders, 2003), and Position Generator (Lin, Fu, and Hsung, 2001, Lin, 1982; 2001). Each of these social capital measurement instruments emphasizes a different dimension of social capital.

McCallister & Fischer (1978) designed the Name Generator. It emphasises on both relationships and resources. It starts with individual-centered social networks or social contexts and collects a social resource inventory. Typically, questions are posed, such as "Do you know anyone who helped you get your current job/helped you get this house" etc., and the respondent is asked to provide a list of names of those who did help. The results of this type of research can provide very detailed social capital descriptions (Burt, 1984; Snijders, 1999). Furthermore, the relevant ties between individuals, locations, and network resources can be computed (Lin 2001b). However, the data from Name Generator consist only of the direct ties of an individual (thus there is a bias towards stronger ties), and is impractical to list all of the members in a network (e.g. if the individual's network is very big, it can be relatively time consuming and costly) (McCarty et al., 1997; Fu, 2005). The Resource Generator and the Position Generators can partly overcome those shortcomings.

The Resource Generator puts more emphasis on resource-oriented social capital. With a fixed list of specific social resources, this instrument asks about access status. For instance, "Do you know anyone who can repair a car or bike/can loan you a large sum of money/is active in a political party." The Resource Generator has been developed in the context of certain developed countries but without major preparatory work, it should not be used in different cultural contexts (Van der Gaag & Snijders, 2005).

The Position Generator focuses more on social resources than relationships in social networks by mapping a set of occupations in a hierarchically modelled society in terms of job prestige for individuals. Lin (2005) commented that "Social capital is seen as contingent on network features... but not identical to these features." (p. 19). The Position Generator methodology allows measurement of both capacity (accessed

resources) and actual uses for particular actions (mobilized resources) (Lin, 2005). Typical questions are "Do you know anyone who is a farmer/lawyer... etc." The Position Generator enhances the Name Generator, but the prerequisite for the Position Generator is that most of the national occupational job-prestige ratings are known (Dufhues, Buchenrieder, & Fischer, 2006). However, the Position Generator is more suitable in this case than the Resource Generator, because Lin (2001) has already applied it in several cities in China, which is the focus of this study.

b) Selected measures of social capital

The first and most important purpose in building up a questionnaire is to create a composite social capital value. Regarding the case study area in this thesis, Zoucheng is a county-level city; membership of some grassroots organizations might not be enough to represent individual social capitals. Thus, the social capital section was based on Lin's study (2001) in order to create social capital scores. Lin has studied social capital in China for many years and has listed the relevant prestige scores for "job positions" in terms of the Chinese situation. The Prestige Rating scores were ranked and calculated from the contact status, social reputation, and census listing of occupations (Lin, 2001). A sample of structural positions that are salient in a society that Lin (2001) used for his research, is based on occupations, authorities, work units, class, or sector. With these sampled positions, respondents need to indicate contacts in each of the positions (e.g. those known on a first-name basis). Considering the sampling content or role areas, this collection method (i.e., the Position Generator) samples hierarchical position, and is role/location neutral (Lin 2004).

In the process, each respondent was asked if he knew anyone in each of the sampled positions. If the respondent indicated more than one contact, the respondent was asked to focus on the most familiar one. Then, Lin constructed three social resource access measures: the position accessed with the highest prestige score, the difference between the highest and lowest accessed statuses, and number of job positions accessed. Finally,

Lin (2001; 2004) used a factor analysis method to generate a single value for social capital scores.

Regarding details of the questionnaires of the social capital, participants were asked the question "Of your relatives, friends, and acquaintances... (If respondents knew more than one occupant of any position, they were asked to think of the first occupant who came to mind.) Do you know people in the position?" The respondent was then asked whether any of his or her kin ties (i.e., relatives), non-kin ties (i.e., friends, acquaintances) or others held such a position.

The sampled job positions were distributed in general social capital levels. Uphoff (1999) pointed out that investment of time, prestige, money, and information can yield a benefit flow in employment, other payoffs, income, and knowledge for networks. Thus, each sampled job position has its own prestige score (see Table 4.3). The survey followed the prestige ratings constructed by Lin (2001), Bian (1994) and Lin et al. (1997) for Chinese occupations to overcome potential differentials between countries for prestige scores on job positions.

General social capital	
Accessed positions	Prestige score
University Professor	91
Mayor	83
Head of Bureau	76
Lawyer	72
Journalist	68
Head of an Enterprise	67
Chief of a Section	60
Elementary School Teacher	58
Worker	45
Administrative Personnel	45
Electrician	44
Farmer	30
Housemaid	11

Table 4.2: Prestige rating for job positions in China

Source: Author's compilation from Lin (2001), Bian (1994) and Lin & Ye (1997)

4.2.3.2 Other information

Other information including demographic questions, and other relevant questions such as monthly income and expenditure were based on the Living Standards Survey in China (World Bank, 1995; 1997), the Vietnamese Household Living Standards Survey (General Statistics Office, 2006), and a manual for planning and implementing a the living standards measurement study survey (World Bank, 2003).

Household surveys also collected information about farm products that households had planted last year, which were based on local Yearbook listings. The survey also used car ownership as a measure of wealth. Along with the increase in private wealth, many Chinese have come to aspire to car ownership, thus this is used as an indicator of family income and status. In the auto market in China, although the prices of domestic brands have fallen rapidly in recent years, many households still choose foreign-brand cars because they are recognized as one of the important indicators of household wealth and status.

4.2.4 Sample size and basic information

A total of 228 respondents completed the survey. Respondent characteristics are summarized in Table 4.4. More than 60% of survey participants were household heads. There were more men (88%) than women which reflects the fact that mining is more restricted to males. The sample consisted of nearly equal numbers of respondents living in either the rural or urban neighborhoods of a county city, Zoucheng. The mean age of interviewees was about 35 years. More than three-quarters (78%) of the respondents were married. On average there were nearly four household members and one child in a family. In terms of local living and working experiences, participants had on average 26 resident years and 14 working years respectively. The collected data were expected to measure local social capital (e.g. social ties, social relationships, social resources etc.) in a local city since most of the participants lived or worked in the local area.

4.3 Analytical framework design

The general framework for the research process for this thesis is based on the sustainable livelihood framework. In Figure 4.1, different boxes focus on different components of the sustainable livelihood (discussed in Chapter Three), such as shocks and risks of natural resources-dependence (vulnerability context), social capital measurement (household livelihood assets), and analysis in a simulation mode (relevant policy implementation, livelihood strategies and outcomes). The livelihood framework is based on the theory discussed in previous chapters. Relevant issues, risks, and shocks to the local community are very important factors in building up the background relationships among three local industries (mining, farming, and tourism). Ordinary least squares (OLS) regression was used to establish associations with the composite measure of access to social capital. Using the relevant coefficients and variables from regression analysis and background information on Zoucheng, the relevant livelihood strategies and outcomes were generated using a simulation model. This simulation model was used to simulate the socio-economic status for the local community, household behaviours in response to different livelihood type, and relevant policy implementations. The relevant issues, influences and risks relating to mining activities are explored in Chapter Five. Social capital measurements and analyses are discussed in Chapter Six. The simulation model is introduced in Chapters Seven and Chapter Eight.

Figure 4.1 illustrates the four sections (represented in the four boxes) of this research. The bold and highlighted arrows and frames indicate the main flow of the research process. Figure 4.1: *Research framework*



4.3.1 Issues, risks and shocks relevant to the local community

Box (1) in Figure 4.1 explores the relevant shocks, risks, and issues relating to natural resource-dependent activities (mining). It involves collecting secondary data from local documents and literature. Although Box (1) cannot directly answer any of the research questions mentioned in Chapter One, it is indispensable. It provides the general relationships between the three important industries (mining, farming, and tourism) in Zoucheng, which are important links in building up the background scenario for the simulation model. Without these risks analyses, the whole research purpose could be out of focus.

The relevant risks, issues, and potential shocks of mining are measured in two areas: environmental issues and livelihoods. Environmental pollution focuses on local farmland quality such as subsidence areas and atmospheric pollution, which may affect both agricultural output and number of tourist arrivals. Issues related to livelihoods were examined in terms of the amount of income resources produced by different household members. With the dominance of mining activities in household incomes, developing alternative livelihoods for other mining household members might be more feasible than focusing on miners themselves, which provides an important guideline to in diversifying livelihood choices for mining households in the simulation model.

4.3.2 Measures of social capital

Measuring social capital is one of the main purposes of this study. There are three research areas in social capital measurement: collecting and calculating social capital scores, discussion of the strength of social ties (e.g. kin and non-kin ties) in accessing work, and regression analysis of factors associated with social capital (see Box (2)).

As discussed earlier (see Section 4.2.3.1), three variables (i.e., number of sampled job positions accessed, upper reachability, and range of positions accessed) were used to generate a composite value for social capital scores. Role relations (e.g. kin & non-kin

ties) are also important sources of social capital, which have been widely used to measure the strength of social ties (see Section 3.3.1). This section summarizes and discusses the frequency of accessing different sampled job positions in terms of the strength of ties (kin & non-kin ties). With this summary, the characteristics of local role relationships on accessing jobs or other useful information can be understood, and can then be applied in the livelihood development policies and simulation model analysis in following Chapters. It is particularly useful to analyse the role of social capital in Chinese society, including the role of family relationships in accessing social resources.

The third section in Box (2) uses regression analysis. The dependent variable is social capital, which is calculated from variables that measure the range, variety and composition of social resource. The aim is to examine the empirical relationships between the strength of social ties (e.g. kin and non-kin ties) and social capital to address question (1) in Chapter One. Research Question (1) from Chapter One is: do different types of social ties (kin and non-kin ties) lead to differential access to social capital in a small community in China? Some other independent variables are also expected to positively associate with social capital, such as higher education and length of local residence. On the other hand, some negative linkage is expected as well (e.g. rural residents may have fewer opportunities to access social resources).

4.3.3 Simulation model and analysis

Boxes (3) and (4) relate to the process of establishing and analysing the simulation models. With all of the above analysis (e.g. risks and social capital), the simulation model (Box (3)) deals with the complex and multiple interactions among the variables that were described above, in order to simulate the background livelihood status for a local community. This simulation analysis draws out the role of social capital in a sustainable livelihood development model and the possibility of an alternative livelihood choice (tourism) for rural households. With different policies applied in the simulation model, the simulation can investigate different livelihood strategies and outcomes to determine possible alternative livelihoods for rural households. The

simulation model is discussed in Chapters Seven and Eight. The results of the regression and simulation models will evaluate the validity of Research Question (2) in Chapter One.

4.4 Translation, data entry and consistency checks

The data collected from household surveys were recorded in the Chinese language. Since the targeted city is the hometown of the author, the questionnaires were translated into the local language to help participants to thoroughly understand the meaning of each question. The format of the questionnaire was very similar to the English version for ease of data entry.

The raw data were firstly entered into Excel and double-checked. Some handwriting was hard to understand, and some extreme or inconsistent data such as improbably high income or expenditure were removed.

4.5 Calculation of composite social capital value

Following Lin (2001; 2004), three indicators of social capital were then constructed from the data, using the Position Generator items: (1) the total number of positions accessed; (2) the prestige score of the highest position accessed (upper reachability); and (3) the difference between the highest and lowest positions accessed (range of positions accessed). Numbers of positions accessed were the total number of job positions that participants could access; the highest position accessed was the highest prestige score that participants could access; and the range of prestige was the difference between the highest and lowest prestige scores that participants could access. To assess whether the three variables for each type of social capital could be considered as indicators of a single dimension, factor analysis was performed on these three indicators of these three variables were used to generate a composite value for general social capital.

Next, factor structures for access to social capital were constructed. Following Lin (2001b), factor analysis was used to determine whether one value could be generated to represent a composite of each individual's access to social capital, called a *Social Capital Score*. If so, then the score could be used in subsequent analysis as a quantitative measure representing access to social capital. The three position generator variables (number, prestige, and range) were included in the analysis (using principal component and varimax rotation). The general mathematical model is shown below:

$$X_{i} = a_{i1}F_{1} + a_{i2}F_{2} + a_{i3}F_{3} \dots + a_{ij}F_{j} + a_{im}F_{m} + \varepsilon_{i}$$

i=1,2,...,*p*; *m*<*p*

where:

F values are measures of social position *a* values are factor loadings *Xi* is the score for the *i*th subject ϵ_i are independently distributed error terms, E (ϵ)=0

Table 4.3 presents the results of a three-factor solution for access to social capital. The first factor has an eigenvalue of 2.63 and explains 88% of the covariance structure. The second and third factors have very small eigenvalues (the second factor had an eigenvalue of 0.27, and explain 9% of the covariance structure.). All factor loadings have the expected (positive) sign. Thus, these three measures of position data are highly correlated and produce a single factor. A constraint condition (i.e., factors having eigenvalues greater than 1 to generate factor loadings of the three variables imposed on the single factor) produce very high factor loadings; see Table 4.3 (0.922 on numbers of positions accessed, 0.966 on range, and 0.92 on highest position accessed).

A Kaiser-Meyer-Olkin test (KMO, overall 0.707) was used to assess the appropriateness of using factor analysis, in which the measure of sampling adequacy provides an index (between 0 and 1) of the proportion of variance among the variables

that might be common variance. If the KMO index¹⁷ is near 1.0, the factor analysis is applicable; if KMO is near 0, the factor analysis is not relevant. Table 4.3 shows the KMO index is 0.707, which is an acceptable level. The squared multiple correlations (SMCs) of variables with all other variables (in this case, between 0.735 and 0.843) were also used to test the strength of the relationships among the measured variables. According to Bagozzi & Yi (1988) and Hair et al. (2006), the recommended value for SMC is greater than 0.5. Thus, the SMC test results in Table 4.3 are acceptable.

Variable		Sample
General social cap	ital	
Factor eigenvalues		
	Factor 1	2.631
	Factor 2	0.265
	Factor 3	0.104
Cumulative		
	Factor 1	0.877
	Factor 2	0.965
	Factor 3	1
Factor loading on l	Factor 1	
Number of positions accessed		0.922
Range of prestige of positions accessed		0.966
Prestige of highest accessed position		0.92
LR test: Prob>chi	2	< 0.000
KMO test:		
number of positions accessed		0.750
Prestige of highest accessed position		0.757
Range of prestige of positions accessed		0.638
Overall		0.707
SMC test:		
number of positions accessed		0.735
Prestige of highest accessed position		0.727
Range of prestige of positions accessed		0.843

Table 4.3: Factor Structures of Access to Social Capital

¹⁷ Kaiser (1974, citied in Dziuban and Shirkey, 1974) refined the index further and suggested that anything in the .90s was 'marvelous', in the .80s 'meritorious', in the .70s 'middling', in the .60s 'mediocre', in the .50s 'miserable' and below .5 'unacceptable'.

Therefore, we can be confident that the factor loadings from the factor analysis can be used to construct a plausible index representing social capital with a single score. The distribution of predictive scores on social capital ranged from -2.397 to 2.136.

4.6 Definitions and summary of variables

All of the variables from the household survey that were used in the regression analysis in this thesis were summaried in Table 4.4.

Table 4.4: Definitions and si	ummary of variables fo	or social capital	models $(n=228)$
-------------------------------	------------------------	-------------------	------------------

		Percent	Std.		
Variables	Description	or Mean	Dev	Min	Max
Individual characteristics					
Desident also a must en orben	Rural area=1	55%			
Resident place-rural or urban	Urban area=0	45%			
Marital status:					
Married	1 if manial 0 simple	77.63%			
Divorced or widowed	1 in married, 0 single	1.79%			
Never married		19%			
Gender	1 if male, 0 if female	87.50%			
	Including yourself, how				
Size of family	many people live in your	3.6	0.974	2	8
-	household				
Number of children	How many children in your	1	0.670	0	4
	household	1	0.070	0	4
Duration in local city area	Years resident in the area	26	12.68	1	68
Working experiences	How many years worked	14	9.00	1	37
working experiences	since graduation	11	2.00	1	57
Age		35.2	8.84	20	68
Education level					
Education level (participants):	The highest education level at	tained for pa	rticipants		
Less than high school		31.53%			
High school or similar		32.89%			
College or more		35.59%			
Expenditure					
Phone	Monthly expenditure on	127.44	88.14	0	600
Wealth and wealth indicator	pnone/Internet				
weath and weath indicator	Ormanskin of a familian and	1.00/			
Car	Ownersnip of a foreign car	10%			
Source: Household Survey da	ata				

4.7 Applications to research questions

There are two main questions in this study (see Chapter One). The first question is addressed by the regression model. The second question is explored in the simulation model.

4.7.1 Regression model

1) Do different types of social ties (kin and non-kin ties) lead to differential access to social capital in a small community in China?

In accordance with the livelihood frameworks, the first step is to analyze the livelihood assets and their interrelationships. Thus, the first question is how to evaluate the social capital.

A). What is social capital and how is it generated?

The relevant concepts and calculation for social capital have been discussed in Chapter Three and Section 4.5 of this chapter respectively.

B). How are other factors are associated with access to social capital?

Ordinary least squares (OLS) regression was used to establish associations with the composite measure of access to social capital. A number of variables were posited to be related to social capital; these included human capital, length and location of residency, household size, access to social networks, and wealth. Greater human capital (education level of the respondent) and wealth (ownership of a foreign car) open up more opportunities for interactions, particularly with people in higher prestige positions. Human capital is measured in terms of numbers of respondents who graduated either from middle school, high school or higher educational levels. People naturally accumulate more social capital the longer they live in a given location; thus the length of residency was included. Similarly, people who live in urban areas and those with larger families might have more access to different positions, so rural location and household size was also included. The respondents' monthly expenditure on the

Internet and telephone was taken as a proxy for access to communication, which can be used to develop social capital. Finally, the percentages of kin ties and non-kin ties involved in accessing job positions were included to investigate their relative importance to accessing social capital.

The parameters for the equation were estimated using ordinary least squares (OLS).

Regression Model:

 $SK_{i} = \partial + \beta_{0}H_{i} + \beta_{1}RS_{i} + \beta_{2}R_{i} + \beta_{3}NM_{i} + \beta_{4}TEL_{i} + \beta_{5}CA_{i} + \beta_{6}K_{i} + \beta_{7}NK_{i} + \epsilon_{i}$

where:

- SK: The estimated scores for general social capital performances
- H: Human capital resources for respondents. The human capital part includes two dummy variables for respondents (1) Junior middle school (compulsory studying in China), (2) higher educational level (diploma or more)
- RS: Number of years that respondent has lived in the local city
- R: The living place (rural area=1) for each respondent
- NM: Number of household members
- TEL: Respondents' monthly spend on internet and phones
- CA: Ownership of a foreign car
- K: The percentages of kin ties accessing job positions
- NK: The percentages of non-kin ties accessing job positions

4.7.2 Simulation model: system dynamics methodology and Vensim

After the regression analysis of the factors associated with social capital, the relevant linkages and parameters on social capital can be applied to livelihood development to explore Research Question (2):

Does the strength of social capital alter local livelihood patterns, such as diversifying household livelihoods and finding new livelihoods in an industry with potential for development?

This is the main question in this study. The social capital information is collected from this chapter and Chapter Six. Using results from multiple regression is a coomon method for inferring values for unknown parameters in System Dynamics models (Houghton, et al., 2014; Higuichi et al., 1996; Mayerthaler, et al., 2009). The risks of mining activities and most of the background information are from Chapter Two and Chapter Five. With all of the relevant information, this study will establish a simulation model to analyse different livelihood choices and outcomes. The following paragraphs introduce the theory and software that the study will use in establishing and analyzing the simulation model in Chapters Seven and Eight.

There are complex economic and environmental considerations in the real world. Thus, the question of how natural capitals or environmental issues should be managed in the context of local cities' rapid development led to the idea of building a sustainable development model. The system dynamics model is a simple and powerful method for analysing the dynamic relationships between interrelated systems (Guan, Gao, & Fukahori, 2010), which simplifies complex real-world phenomena and can represent a good understanding of the problems and interdependence within the real-world system (Wiranatha & Smith, 2000). In the development planning process, system dynamics models can simulate different policy scenarios and contribute to improvements in the decision-making process (Saeed, 1994; Bergh, 1991).

System dynamics methodology was initially developed by Jay Forrester and others at the Massachusetts Institute of Technology (Maani & Cavana, 2007). The system dynamics approach creates operational maps and simulation models to capture the interrelationships of physical and behavioural processes, information feedback, policies, and time delays and test the holistic outcomes of alternative plans and ideas (Wolstenholme, 1997). It also provides an experimental tool to integrate these issues (i.e., economy and environment) with suitable government structures and sector policies to evaluate alternative long-term capacity planning policies (Vlachos, Georgiadis, & Iakovou, 2007). Policy makers can use models to test different policy options via learning-by-doing, particularly since some experiments are costly or even impossible to conduct in real-world conditions (Holling et al., 1998; Bontkes & Keulen, 2003). It is a well-established method to develop understanding of the complex interactions between the environment and human actions (Costanza & Ruth, 1998; Hannon & Ruth, 1994). In addition, the system dynamics model can efficiently obtain a general indication of changing behaviour over time in different scenarios (Sterman, 2000 & Ford, 1999). These feedbacks can be considered as a platform to facilitate the most relevant interactions among the several agents involved in the region (Stave, 2003; Vennix, 2001).

Vensim is one of the available softwares (others include Stella, Powersim, and Simile) used to develop System Dynamics models diagrammatically by drawing stock-and-flow diagrams. The simulation model in this study uses Vensim Ple Plus software (Ventana Systems, 2010) to analyze relevant components of local issues involved during the development.

Vensim is one of the "visual modeling systems" approaches to constructing a simulation model. The programs on Vensim can link and assemble various components to construct visual modeling systems, conceptualize dynamic relationships, and optimise models of dynamic systems (Venstems Systems, 2010; Muetzelfeld & Massheder, 2003; Smith, Holzworth, & Robertson, 2005; Costanza & Voinov, 2001). It provides a highly visual representation of the structure of the system and the existing relationships between variables so as to facilitate understanding of the behaviour of the system (Chaves, et al., 2013).

Vensim is used for a wide range of economic and environmental applications. Fraser, et al., (2010) used Vensim to evaluate the vulnerability in food systems. Chen, et al., (2005) developed a simulation model to assist government in obtaining alternatives for sustainable land use management. Other areas of application include the labour market
(Kara & Zaim, 2012); hydrology (Fleury, et al., 2007); environmental pollution (Chaves, et al., 2013); poverty and livelihoods (Tshimpanga, 2012); municipal waste management (Karavezyris, et al., 2002); public finance (Mantel, et al., 1994); and an urban transportation system and its application (Wang, et al., 2008).

Parsons, et al. (2011) concentrated on the agricultural area. They used Vensim to simulate farming systems (crop-livestock) to assess the effects on crop-livestock production for smallholders. For environmental management, Crooks, et al. (2013) concluded that it is feasible to use systems dynamic modeling (viaVensim) for economic evaluation of ecological restoration activities. Pedercini (2004) simulated the socio-economic-environmental system of Papua and identified a developmental path for local residents. Development methods, relationships among variables, and results in the literature provided a general framework and some insights for the development of the simulation model in this thesis. However, the simulation model in this thesis has its own characteristics. This is the first known application that couples social capital with system dynamics and livelihood choices in terms of the sustainable livelihood development framework.

There are several advantages in using Vensim to construct a simulation model: (1). It is easy and quick to learn and use; (2). It has a broad range of predefined functions and tools to build, operate and test the models; (3). It is easy to generate graphs and tables, which can be viewed as dynamic outputs (Smith et al., 2005). On the other hand, the major weakness of this approach is its limited flexibility (Smith et al., 2005). For instance, the user generally has less capacity to modify the system with respect to advanced modeling tasks or particular objectives (Costanza & Voinov, 2001).

CHAPTER FIVE:

ISSUES, RISKS AND SHOCKS TO THE LOCAL COMMUNITY

Chapter Two discussed the background of Zoucheng city. The economic activities indicate a mineral resource-dependent lifestyle in the local community. However, coal mining activity is normally accompanied by serious threats to the environment such as land subsidence, seasonal or perennial waterlogging and soil erosion, which can cause losses in agricultural production (Hu et al., 2013). In addition, poor atmospheric quality caused by mining may decrease the number of potential tourists. Finally, over-reliance on mining activity may also limit households' livelihood choices. This chapter explores the risks, issues and potential shocks of mining activities in the local community (i.e., in Zoucheng).

5.1 The impact on agriculture: Farmland quality

Zoucheng belongs to an area where there is heavy production of both coal and grain. In Zoucheng city, the increase in mining activities has resulted in the loss of farmland and negative impacts on water quality and areas, which further impact farmland quality and crop yields.

Large areas of farmland and construction land were eroded by land subsidence in the Jining mining area (Guo & Liu, 2012). The total subsidence area was about 66,000 mu (more than 4,200 ha) (Yang et al., 2007) and these areas were predicted to increase by 2,500-3,000 mu per year in Zoucheng city (The conciliation commission of regional economies, 2006, Dec).

Although these subsidence areas are a relatively small proportion of the total land area in Zoucheng, they have a direct effect on the life of residents, especially with regard to agricultural livelihoods. Coal mining activities lead to serious damage to nearby cultivated fields (Hu et al., 2012). In Yanzhou Coalfield, a large number of local farmers have lost their lands, which in turn intensifies the conflict of interests between the extraction companies and the need for cultivated land (Hu & Zhao 2007; Xu et al., 2014). In Zoucheng, the subsidence areas increased to around 80,300 mu (about 5,353 ha), of which about 43,000 mu (about 2,867 ha) were agricultural land in 2010 (Li, 2012). Table 5.1 summarises the subsidence areas in Zoucheng city in 1998, 2000, 2002, 2004, and 2010. The subsidence areas showed a gradual increase during this period, directly affecting five village towns (Zhongxin, Beisu, Taiping, Pingyangsi, and Tangcun) and more than 20 villages (Yang et al 2007). By 2006, seven schools, 17 factories and 23 villages had to be removed and relocated as a result of the increase in subsidence areas (The conciliation commission of regional economies, 2006, Dec). In the most affected areas, the farmland per capita was reduced to less than 0.2 mu in more than ten villages in 2009. With the current growth rate, the total subsidence areas in Zoucheng will have increased to 150,000 mu, about 67 villages will have to be relocated and about 200,000 people will lose their land by 2050 (The conciliation commission of regional economies, 2006, Dec).

Table 5.1: Subsidence areas in Zoucheng city

Year	1998	2000	2002	2004	2010
Subsidence areas					
(ha)	2,600	3,182.34	3,515.6	4,266.67	5,353.3

Source: Author's compilation from Yang et al. 2007; Li, 2012.

Table 5.2 shows the changed rate of land use in six village towns in the local mining area from 1990 to 2005. Farmland size clearly declined, while urban and industrial areas significantly increased (except Pingyangsi). The changes in Pingyangsi village town may be due to labour migration from farming to mining in other village towns as a result of the reduced service life for the local mines. In addition, Pingyangsi is one of the village towns most affected by land subsidence in Zoucheng (The conciliation commission of regional economies, 2006, Dec).

Village town	Farmland	Water area	Urban and industrial area
Pingyangsi	-16.29	122.56	-7.01
Beisu	-10.4	43.07	13.84
Taiping	-73.47	-2.3	293.94
Tangcun	-1.18	27.72	1.82
Guoli	-1.41	11.8	3.48
Shiqiang	-0.91	-0.45	17.96

Table 5.2: Changes in rate of land use in village towns from 1990 to 2005 (%)

Source: Author's compilation from Yang et al 2007

Furthermore, mining activities led to profound modifications of water quality and area. In the Yanzhou Coalfield, underground coal mining affects the main water-rock interactions that lead to altered chemical characteristics, such as alkalinity of the groundwater (Han et al., 2013).

In addition, some water reservoirs were created with the increase in subsidence, due to the collapse of the areas above underground coal excavations (as cited in Sierka et.al, 2009; Jankowski & Molenda, 2007). Water reservoirs in subsidence area are treated as mining damages (as cited in Sierka et.al, 2009). In the Yanzhou Coalfield, the depth of subsidence is up to 3-10 m below the surface, while the water table is high, at about 3-4 m (Hu et al., 2012). As a result, water collects in the middle of the sinking basins, destroying nearby arable land (Hu et al., 2012). In 2007, there were 15 water bodies and a total water area of 18 km² in the mining area that were the result of subsidence in the Yanzhou Coalfield (Hu et al., 2012). In the Zoucheng mining area, Table 5.2 shows that water area increased as farmland shrank, except in Taiping and Shiqiang village towns, where both slightly decreased.

In the Jining area, there is a potential causal relationship between subsidence areas and water areas or wetland systems in the mining area (Guo & Liu, 2012). Because of the degree of subsidence and the rising of the water table, the effective moisture of the soil changes. The increased soil moisture content may lead to fewer crop roots and problems with root establishment (Loheide, 2011). High moisture in the soil also affects

salinization (accumulation of salt in the soil profile) and waterlogging in the rainy season (Loheide, 2011). All of these can cause degradation of land quality, soil erosion, and crop yield reduction in mining areas (Xu et al., 2014).

As a result of the influences of subsidence and water area or wetland systems in mining areas, damaged cultivated land is categorized into three grades: mild yield reduction areas; moderate yield reduction areas; and out of yield areas (Bi & Ding 2003; Hu et al., 2007; Kong, 2007). Mild yield reduction areas suffer from a reduction in soil moisture and fertility which influences crop yields to a certain extent. Moderate yield reduction areas have more serious damages because of the subsidence, seasonal waterlogging, and land salinization, leading to crop yield reduction and potentially an eventual total loss of the crop yields. In loss of the crop yield areas the ground surface falls below the groundwater level as a result of mining subsidence. Consequently, crop yield is totally lost in these areas because they are under water (Xu et al., 2014). According to Xu et al. (2014), land subsidence and groundwater level influence the farmland quality and crop yields in the Zoucheng mining area.

For water areas and subsidence areas, land reclamation consists of either gravitational drainage or producing different, value-added products such as developing aquaculture in Zoucheng. For instance, Xigu village in Beisu village town established an aquaculture base using this kind of water area (The conciliation commission of regional economies, 2006 Dec). However, Bury (2004) and Molenda (2006) pointed out that the deposition of post-coal mine wastes significantly changed the water quality in reclamation reservoirs.

In terms of soil quality in the Zoucheng area, mining subsidence can lead to soil desertification, pollution, and soil erosion (Wang et al., 2009; Meng et al., 2009). Mining subsidence together with declining regional underground water levels changes the slope of the ground surface, which aggravates soil erosion (Meng et al., 2009). Subsidence changes the slope of the cultivated land and in some cases, increased surface runoff leads to more serious soil erosion (Xu et al., 2014). In the Yanzhou coal

mining area, the pH, sulfate levels, total hardness and solid content in subsidence areas are extremely high, which poses potential risks to the quality of groundwater, public health, and soil quality (Meng et al., 2009). Yang, et al. (2007) used statistical data in Zoucheng city to analyze coal mining effects on local ecosystems and land use change. The results indicated that the land use structure changes rapidly and soil quality gradually decreases with the growth of coal mining, which negatively impacts on agricultural production in the area.

5.2 The impact on tourism: Air pollution

One of the negative effects of coal production is air pollution (see Chapter One and Chapter Three), which may affect the attractiveness of Zoucheng and reduce the number of tourist arrivals.

As discussed in Chapter Two, Zoucheng and other surrounding cities lie within the Yanzhou Coalfield. Given spillover effects, pollution within Zoucheng, particularly air pollution, is not only caused by mining activities in Zoucheng, but also by activities outside the county. More broadly, there may be various causes of local air pollution. However, the actual development characteristics of Zoucheng, such as the dominant role of mining in the local economy, cause the authorities to engage in introspection rather than take responsibility. I have lived in Zoucheng for more than 20 years. The widely recognized and accepted belief in the local community is that local mining is the direct and most important cause of worsening air quality, with the altered distribution and dispersion of airborne dust in the local urban area being due to increased mining activities in the west of the city.

In Zoucheng city, tourism is relatively small and undeveloped, so it contributes less than mining in the economic area. As with coal mining's effects on agriculture, there is little research about the relationship between mining and tourism in Zoucheng. However, poor air quality may impair environmental tourism products and reduce the numbers of potential tourists. It is important to analyse the effect of coal mining on local air quality.

In Zoucheng city, coal mining operations in local mines and coal consumption by the local coal-fired power plant (Zouxian power plant) may degrade the local atmospheric environment. In addition, mining subsidence leads to soil desertification and erosion, which increase dust in the atmosphere, and consequently increases air pollution (Wang et al., 2009). Figure 5.1 illustrates the growth trend for coal production and industrial dust discharge in Zoucheng city from 2000 to 2013. The left axis represents coal production and the right axis is the volume of industrial dust discharged. One thing that must be noted is that coal production is not the only source of industrial dust discharged in the city. However, with the high level of local GDP growth and the dominant role of mining in local industries, coal production is largely to blame for this pollution issue. As shown in Figure 5.1, coal production and industrial dust discharge showed very similar movements in the period 2000-2005 (for example coal production increased from about 29 million tons in 2000 to more than 39 million tons in 2005, which was associated with industrial dust discharged increasing from about 1,140 tons in 2000 to about 5,067 tons in 2005), and there were similar results for the period 2006-2012. Only two periods (i.e., 2005-2006 and 2012-2013) indicated an opposite movement.

Figure 5.1: Coal production and industrial dust discharged in Zoucheng from 2000 to 2013



Source: Author's compilation from Zoucheng Statistical Yearbook, 2000-2013

Although the local economy has grown rapidly, rising incomes tend to be correlated with a greater demand for environmental quality (Anderson & Locker, 2002). In the coal mining areas, the typical political issues are the problems that arise between local people and coal mine-induced environmental problems (Shi, 2015). In Zoucheng city, there is growing awareness among residents that environmental issues have caused health risks, even when total coal production was generally decreasing after 2003. Figure 5.2 shows the number of pollution issues petitioned by residents to the local Administrative Petition System from 2000 to 2013. This can be considered the residents' feedback on local environment quality. The bar chart represents the total number of pollution issues including water, air, noise, air and others each year. The environmental issue most directly related to the operation of the mines is air pollution (Ghose & Majee, 2000). Therefore, the line graph presents the number of exhaust emission petitions each year. On both graphs, the general trends are upward. In the first three years, pollution issues were kept at a low level. Then, complaints about general pollution climbed rapidly to 54 and air pollution was down at 16. In 2005, the total pollution complaints increased to a peak of 87 and air pollution rose to 28. In the next two years, the number of general pollution cases dropped slowly to 81 in 2006 and sharply to 61 in 2007, while air pollution issues continued to rise, increasing to 35 in 2007. The following year,

general pollution petitions rose slightly to 70 then dropped to 65 in 2009, before gradually rising again to 84 in 2012. In 2013, the number of general pollution cases decreased to 72. Meanwhile from 2008 to 2013, the number of air pollution issues fluctuated around 33, peaking at 37 in 2012. These graphs therefore indicate that: 1. Local residents have gradually paid more attention to local environmental issues; and 2. More sustainable policies are needed to coordinate economic development and environmental protection.



Figure 5.2: Number of pollution issues in the Zoucheng local administrative petition system, 2000-2013.

Source: Author's compilation from Zoucheng Statistical Yearbook, 2000-2013.

In addition to being an industry, rural tourism is a multi-faceted activity rooted in environment and social resources (Yang et al., 2010). The local community can take advantage of various environmental and social resources to develop a variety of environmental and cultural tourism products (Cawley & Gillmor, 2008; Garrod et al., 2006). However, lack of cleanliness can make cities unattractive (World Bank, 2010).

In Zoucheng, most of the tourism resources are scenic spots (see Section 2.2.2.1), which are often sensitive to environmental damage. Poor air quality may also damage historical sites in the long term. Furthermore, a high quality environment is a decisive element of tourism development, in terms of attracting more visitors (Yang et al., 2010).

Thus, coal production, the industrial dust discharged, and related pollution issues may potentially threaten local tourism resources and the attractiveness of Zoucheng, which in turn decreases local tourism development.

5.3 Over-reliance on mining activity

Working in mining is a livelihood strategy for many households in Zoucheng. The sustainable livelihoods approach seeks to understand the factors that lie behind people's choice of livelihood strategy. Then, it develops and reinforces the positive sides and mitigates the constraints or negative effects of current livelihood activities (DFID, 1999). Examining the amount of income resources provided by different household members is important when thinking about livelihood strategies (DFID, 1999). Thus, this section discusses the distribution of income resources between miners and their partners (married or lived together).

Because of privacy issues about people's incomes and the difficulty of gathering information about "underground economic activities", this study collected information on base payments and bonuses as part of monthly income. Total family income is based on that of the survey respondent and their partner. In this household survey the average mining wage was about 2,943 yuan per month, with the partner's wage being about 1,689 yuan per month. The miners' proportion of the household income (64%) was almost double their partners' (36%).

When the responses were divided into urban and rural households, the effect of occupation on total income structure as divided between participants and partners produced different results. In urban households, the average mining wage was around 3,192 yuan per month, which made up about 57% of total household income. In rural households, the situation is more significant. On average, the participants' incomes decreased to about 2,736 yuan, which made up about 73% of the total household income. In general, miners received significantly better levels of pay than their partners in households in both urban and rural areas.

Of the total 194 participants of the household survey who completed the mining income section, about 84% of households had a mining income as their family's main source of income, and 85% considered that the income of the current job could cope with daily household expenses. For income sources in rural households (105 participants), only about 1% of households had farm activity (i.e., total income from crop production) as the main source of their households income, and about 32% of rural households considered their agricultural production income could deal with their routine household expenses.

Comparing the significance of mining incomes in rural and urban households, mine work plays a more important role in rural households than in urban ones. This income source dependence potentially increases the vulnerability of rural household, which might further reduce the range of options and the ability to switch between multiple strategies to secure their livelihoods. Because mining is dependent on a non-renewable resource, it is not a sustainable industry for the local community in the long term. Thus, coal miners and other related-industry workers may lose their jobs when local mineral resources have been consumed or mines are unexpectedly closed by government command. Rural mining households are likely to suffer more pressures since they derive more of their income from mining. As discussed in Chapter Three, rural mine labourers are normally categorized as farmers in China. They have to return to their farms once mining operations are shut down. This is one of the important reasons for this study focusing on rural households in the simulation model (see Chapters Seven and Eight) to help them to deal with these potential shocks to their livelihoods. The preceding discussion shows that rural dwellers in Zoucheng city have to be encouraged to access more livelihood assets and local resources to find an alternative sustainable livelihoods with the help of relevant policy implementations.

However, because mining is highly paid in the local labour market, this may slow down the process of transition to alternative livelihoods. For instance, a highly paid miner is very unlikely to leave his current job and operate an alternative industry (e.g., rural tourism) with relatively low rates of pay. As discussed in Chapter Two, local mining cannot satisfy the level of job demand from rural households. It is not possible to provide enough mining jobs to accommodate all members of rural households. However, because of the low rates of pay for partners in mining households, these household members are a more likely target group for alternative livelihoods. Although miners' own self-interests are not changed, the interests of their households are enhanced. In this case, the purpose of encouraging other household members to work in alternative livelihoods is to diversify household livelihood choices and reduce potential shocks and risks of over-reliance on mining activity. With the development of alternative industries (e.g., rural tourism), mining households always have the ability to deal with mining job losses.

CHAPTER SIX: SOCIAL CAPITAL

6.1 Introduction

This chapter presents the empirical study results on social capital in Zoucheng. It firstly measures the strength of social ties (non-kin and kin ties) in current social resources. After that, regression analysis is used to examine Research Question (1). Research Question (1) involves associations with the composite measure of access to social capital, which was outlined in Sections 4.3.2 and 4.7.1. More specifically, it investigates whether there is a significant relationship between social capital and other livelihood assets such as human capital (e.g., graduated from middle school or higher educational levels) and resident characteristics (e.g. years resident in local city). The main purpose of this regression model is to examine the empirical relationships between the strength of social ties (e.g. kin and non-kin ties) and social capital (also see Research Question (1) in Chapter One).

These research results have implications for government policy regarding revenue generation, increasing competitive advantages in alternatives such as tourism, and economic development. The applications are used in a simulation model (Chapter Seven), which tests the Research Question (2) in Chapter One.

By following the research methods in Chapter Four, the following sections provides the general distribution of the role relations (non-kin and kin ties) among research participants and the regression results of factors associated with access to social capital.

6.2 The strength of social ties

Table 6.1 summarizes the frequency with which each position was accessed in the sample, by the strength of ties. Due to missing data, these data are only available for 218 of the 228 respondents.

Three indicators of social capital (number of positions accessed, upper reachability, and range of positions accessed) indicate the overall measures of social capital. The mean number of positions accessed was 4.32; the mean prestige of the highest accessed position was 69.3 (about the position of Journalist) and the mean range between the lowest and highest prestige scores was 41.8. All three indicators were lower than Lin's (2001; 2004) study in several large metropolitan cities in China (number of positions, 6.80; highest accessed position, 73.90; range, 44.4). Urban respondents were more likely to have advantages in prestige of highest accessed position and range of prestige of positions accessed than rural respondents. However, the number of positions accessed is not statistically significantly different between rural and urban residents.

In general, the respondents tended to use more non-kin ties (e.g. friends, acquaintances) than kin ties to access most of the social positions (10 of the 13 positions are statistically significant). The exception to this was the 'Farmer' position, which was accessed more using kin (56%), than non-kin ties (26.9%). These kin ties for the 'Farmer' position may be spouses or other relatives living in the adjacent Zoucheng city. This special relationship between miner and farmer may have some implications for government policies on developing alternative economic activities such as rural tourism for rural households. With regard to access to high-ranking positions enabling the availability of better resources for social actions (discussed in Chapter Three), miners could help their relatives (farmers) to access more information to diversify their livelihood choices.

Table 6.1 also compares the use of kin and non-kin ties in accessing each position, separately for respondents living in rural or urban areas. Urban respondents were more likely to use kin ties for 3 of the 13 positions (Mayor, Head of Bureau, and Farmer), which might indicate a special and close relationship between rural and urban residents in a small city. Urban respondents were also more likely to access job positions using non-kin ties (9 of the 13 positions are statistically significant). Furthermore, both urban and rural respondents used more non-kin ties than kin ties to access the positions (8 of the 13 positions are statistically significant).

In summary, both kin ties and weak ties are important factors associated with social capital for the mining sector residents in Zoucheng County and adjacent rural areas. Urban respondents use weak ties more often than kin ties to access job positions.

Variable		Mean		P-value for sig		Percent Using (Kin & Non-Kin)									
General social capital (Position score)	Whole sample	Rural	Urban	difference between urban and rural	Percentage Using (Kin)										
Number of positions accessed	4.32	4.11	4.52	0.20				F	ercentage U	Jsing (Non-l	P-value	P-value	P-value		
Prestige of highest accessed position	69.27	65.97	71.79	0.01								difference between	difference between	for sig difference between	
Range of prestige of positions accessed	41.84	35.33	45.79	0.00	1							Non-Kin	Kin & Non-Kin (Urban)	Kin & Non-Kin	
		Percent		P-value for				P-value for sig				P-value for	(itului)	(eroun)	(Whole sample)
Accessed positions (prestige score)	Whole sample	Rural	Urban	difference between urban and rural	Whole sample	Rural	Urban	difference between urban and rural	Whole sample	Rural	Urban	difference between urban and rural			
University professor (91)	30.8%	29%	32.9%	0.07	2.9%	3.5%	2.4%	0.65	12.8%	9.2%	16.5%	0.06	0.10	0.02	0.00
Mayor (83)	18.7%	16%	21.4%	0.47	2.9%	1.2%	4.8%	0.04	4.7%	3.5%	6%	0.30	0.05	0.52	0.1
Head of Bureau (76)	35.7%	26%	45.7%	0.00	9.9%	6.8%	13.3%	0.00	14%	6.8%	21.7%	0.00	0.69	0.14	0.31
Lawyer (72)	32.4%	23%	42.1%	0.02	2.9%	2.2%	3.6%	0.49	17.2%	10%	26.5%	0.00	0.02	0.00	0.00
Journalist (68)	24.7%	23%	26.7%	0.15	2.9%	3.4%	2.3%	0.34	10.9%	8%	14%	0.09	0.01	0.00	0.00
Head of Enterprise (67)	40.7%	39%	42.9%	0.14	6.4%	9.1%	3.6%	0.28	23.8%	19.3%	28.6%	0.05	0.29	0.00	0.00
Chief of a Section (60)	62.9%	58%	67.8%	0.00	11.2%	14.29%	8.1%	0.79	45.5%	38.5%	52.9%	0.01	0.02	0.00	0.00
Elementary School Teacher (58)	91.6%	92%	90.8%	0.64	23.2%	26.2%	19.5%	0.80	61.1%	57.3%	65.5%	0.16	0.49	0.29	0.22
Worker (45)	88.7%	87%	90%	0.11	35.1%	37.9%	32.2%	0.59	48.7%	44.2%	53.3%	0.03	0.9	0.15	0.4
Administrative Personnel (45)	52.5%	50%	55.3%	0.01	12.2%	11.5%	12.9%	0.83	30.9%	29.2%	32.9%	0.35	0.05	0.06	0.01
Electrician (44)	78.7%	80%	77.3%	0.80	16%	21%	10.2%	0.35	56.9%	53%	61.4%	0.09	0.05	0.00	0.00
Farmer (30)	85.7%	83%	88.5%	0.09	56%	52.6%	59.8%	0.04	26.9%	29.5%	24.1%	0.78	0.00	0.00	0.00
Housemaid (11)	26%	15%	37.8%	0.00	2.4%	1.2%	3.7%	0.22	13%	2.3%	24.4%	0.00	0.13	0.35	0.06

Table 6.1 Comparison of Access to Social Capital by kin and non-kin ties between rural and urban areas (N=218)

6.2 Factors associated with access to social capital

Table 6.3 describes the results of the regression analysis on factors associated with access to social capital. The survey dataset contains data on more than 190 households from 17 village towns in Zoucheng. It is very possible that the residents in each town may not be independent, and this could lead to residuals that are not independent within each village town. In other words, people in the same social network are likely to have similar characteristics and similar relationships to social capital. Thus, the regression also corrected for clustering at the village level.

The equation used the social capital score as the dependent variable. The results demonstrate that access to social capital increases with most of the explanatory variables, except for expenditure on phone and the Internet, low education level, and family size and rural area, which decreases social capital. The nature of the variables being used on this regression model such as household wealth (own a car) and expenditure may have a multi-collinearity effect on the findings. Thus, the study firstly looked for multi-collinearity in the regressors. This study does not encounter the problem of multi-collinearity (see Table 6.2).

Table 6.2: Collinearity Diagnostics for	Variables used in t	the Social G	Capital Model
(OLS)			

Variable	VIF	1/VIF
Years resident in Zoucheng city area	1.27	0.79
Own a car (Foreign)	1.1	0.91
Expenditure on phone and internet	1.02	0.98
Higher education (e.g. diploma, bachelor or higher)	1.33	0.75
Middle School	1.43	0.7
Job accessed through kin ties (%)	1.06	0.94
Job accessed through non-kin ties (%)	1.05	0.96
Family size	1.05	0.95
Resident area (rural=1)	1.08	0.93
Mean VIF	1.	15

Note: High multi-collinearity exists if VIF values are larger than 10 (Gujarati & Porter, 2009, p. 362). All the variance inflation factors (VIF) values are less than 10, which confirms that this study does not encounter the problem of multi-collinearity.

Living in rural areas had a negative association with access to social capital, which contrasts with the higher numbers and range of positions that could be accessed in big cities (Lin, 2001; 2004).

The number of years that a person had lived in the Zoucheng city area had a pronounced, positive association with access to social capital. For instance, people who grew up together tend to have interacted frequently and built up long-term relationships in the community (Gertler, et al., 2006; Onyx & Bullen, 2000). The longer respondents have lived in the local city area, the more opportunities they have to access a greater number of people, acquire information through friend and expand their social circles.

	OLS with Cluster Option					
Explanatory variable	Coefficient estimate	Standard error				
Years resident in Zoucheng City area	0.018**	0.008				
Own a car (foreign)	0.604***	0.121				
Expenditure on phone and internet	0.001	0.001				
Higher education (e.g. diploma, bachelor or higher)	0.394**	0.185				
Middle School	-0.09	0.095				
Job accessed through kin ties (proportion)	1.395***	0.357				
Job accessed through non-kin ties (proportion)	0.975***	0.327				
Family size	-0.076	0.078				
Resident area (rural=1)	-0.409**	0.15				
Constant	-0.769** 0.3					
R-squared	0.337					
Adjusted R-squared	-					
F Test	58.85					
Probability>F	0					

Table 6.3: Factors associated with Access to Social Capital (Partial Regression Coefficients, N=191)

Statistically significant at 10 %(*), at 5% (**) and at 1% (***)

The results show that people who graduated from school with at least a college education (e.g. diploma or above) have greater access to social capital. This may be because these people have more opportunities to access higher positions in social hierarchies. In addition, people with higher education normally have better employment, and can use these positions to access a greater number of occupational and political positions, or higher positions within the industry. By contrast, respondents with a lower-level nine-year compulsory education experience some disadvantages when trying to access social capital.

Furthermore, the percentages of kin ties and weak ties possessed by individuals have a positive relationship with access to social capital. Both kin and non-kin ties are equally likely to be able to access social capital since the two coefficients are not statistically different from each other (Prob>F=0.359).

The results confirm firstly the importance of kin ties in China for access to social capital. In China, many villages are named after the family that makes up the largest proportion of the population, emphasising the characteristic of kinship. Kinship networks play a very important role in rural Chinese life. For instance, relatives help each other in events such as death, birth, and marriage. (Chu, June 2001). These findings are consistent with Lin's (2001; 2004; 2001a) point that vulnerable groups tend to gain benefit through kin ties. In addition, the coefficient of weak ties is also significant. Bian & Ang (1997) suggests that weak ties are also important means to access resources in the Confucianist Chinese society in rural areas and small county cities.

Regression analysis showed that ownership of a foreign car had a positive association with social capital, possibly meaning that higher income or financial support is associated with greater access to social capital.

Some modern tools (cyber networks and mobile phones) have become popular means of communication within kin and friendship networks (Lin, 2001; 2004; 1999a; Gilchrist & Kyprianou, 2011). However, in this study, the results showed that monthly expenditure on phone and the Internet was not significantly related to access to social capital. Nor does family size appear to have a significant association.

6.3 Conclusions

The results of studies on social capital summarized the strength of social ties and social network resources in social capital in a small city in China. Drawing on the work of Lin (2001; 2004), this chapter discussed kin and non-kin social ties and how these different types of ties lead to differential access to social capital. The results of

this chapter indicate that kin and non-kin ties are equally effective ways to access social capital in a small community.

CHAPTER SEVEN: SIMULATION MODELING

7.1 Introduction

From the previous discussions on local economic background and social capital analysis, several results can be applied in this chapter. First, Zoucheng is rich in mineral and tourism resources. Coal mining brings higher incomes and more job opportunities for local residents, while tourism shows a slow development. Second, the danger is that coal mining pollutes the local environment, particularly affecting air quality, and degrades farmland quality. However, mining is still a preferred job in the local community because of the high pay rates in mining and the lack of alternative livelihood choices for local coal miners. However, mining household members (i.e., partners) with lower income levels may be a target group for the development in this region. These results have implications for government policy regarding livelihood choices. The objective of the work in this chapter and the next is to identify and examine potential livelihood paths that might be more sustainable for related households that suffer the issues of mining activities in Zoucheng.

This chapter develops an economic model based on the system dynamics framework to simulate the impacts of alternative policy implementation so as to diversify livelihoods. It develops and simulates a conceptual and empirical basis for the role of social capital in facilitating people's access to an alternative livelihood. Compared to mining development, the competitive advantage of tourism is expected to increase through social capital application and the development of tourist resources in the local city, which may provide more livelihood choices for rural labourers.

The system dynamics approach synthesizes relevant information that is based on previous studies and data collected in this study to analyze their dynamics over time. The simulation results are important to help local decision makers to better address the issues, tradeoffs and conflicts associated with sustainable management of residents' livelihoods and resources.

There are three stages in complete the process of system design, model development, and behavioural analysis. These stages include development of the model concept, development of the simulation model, and simulation of system behaviour in response to policy analysis design (Tshimpanga, 2012). Chapter Seven and Chapter Eight undertake these stages. Chapter Seven first outlines the analytical framework. Then, it introduces the process of simulation model development. Chapter Eight presents the results of the model based on different policy implementations.

7.2 The Analytical Framework

The framework broadly treats natural resources, livelihood choices, labour distributions, livelihood assets (e.g. social capital and natural capital), industries (e.g. mining, agriculture and tourism) and products, and individual incomes as endogenous and interrelated to simulate the general economic and livelihood scenarios.

A wide variety of models are based on systems theory to develop approaches to understanding the dynamic behaviour of complex systems (e.g. Garcia-Barrios, Speelman, & Pimm, 2008; Jogo & Hassan, 2010; Walker et al., 2005; Hao, et al., 2007). All of these complex systems include multiple elements that are tightly interwoven into one system. The adapted framework used in this study consists of five subsystems: mining, agricultural production, social capital, tourism, and livelihood outcome. All of these subsystems are defined as different components of local economic and household livelihood development. Moreover, a set of causal relationships are designed to investigate interconnections among subsystems to determine how changes in one subsystem affect others, and the influences of the resulting feedback. Finally, following DFID (1999), Lin (2004) and the household survey analysis conducted in this study, this chapter develops indicators to measure relevant types of assets such as social and human capital and their interrelationships for local mining households. Social capital is as defined by Lin (2001), natural capital consists of the quality of farmland and the coal deposits, and human capital is represented by numbers of workers. All of these components and their interrelationships are illustrated in Figure 7.1.

Figure 7.1: Analytical framework showing the interactions between components of the system



Figure 7.1 illustrates the main sectors of the model and how they connect with each other. The modelling design and process are based on data and information from Zoucheng. The model built here will first simulate the environmental issues resulting from coal production. Coal production degrades farmland quality and pollutes the local environment, which in turn affects the growth of agricultural outputs and the number of tourist arrivals.

The traditional livelihood of Chinese farmers is agriculture (He & Chen, 2012). Thus, agriculture is an important subsystem in the simulation model. The agricultural production subsystem includes crop production and farmland quality at a household level. The growth of agricultural output depends on the growth of agricultural inputs, for example, natural growth (such as natural generation of fruit crops in orchards or re-seeding of land) and labour input. Farmland quality is negatively linked to coal mining activity, as discussed in Chapter Three and Chapter Five. The influence of total labour input (i.e., number of farmers) on the growth of agricultural output is assumed to be very low in the simulation

model. This is because of the issue of surplus agricultural labourers and the very low marginal productivity in the agricultural sector, which is almost zero for some crops in China (Fan et al., 2003; Wan & Cheng, 2001). Rural workers are simulated as having more livelihood choices (e.g. tourism) and moving between farming and tourism activities. Finally, total net profits (seasonally) from farming activities are generated from the difference between crop cost and price. In the simulation model, crop price and the total cost are influenced by agricultural outputs. Tourism also brings benefits for farm products. More tourists may drive the price of agricultural products up, which may change the crop profits in the simulation model.

The negative correlation between tourist numbers and coal production is simulated in the tourism subsystem. Tourists prefer a good environment, particularly in terms of air quality (Vargas, et al., 2009). In the tourism subsystem, air pollution resulting from the mining industry will reduce the number of tourists. On the other hand, government investment in public infrastructure may attract more tourists as more facilities become available for on tourism services. In the simulation model, government investment in public infrastructure alters with the growth of total sectoral output values (mining, agriculture and tourism). The quality of tourist services is affected by the number of rural tourism operators; more of them will attract more tourists. An increase in the number of tourist arrivals in Zoucheng is likely to mean an increase in expected tourism incomes and in the number of workers required in the tourism industry.

The social capital component is influenced by the length of residence in the area because of the increased opportunities and time available to maintain and increase social networks and relationships. According to some studies (Aguilera & Massey, 2003; Lin, 2001a; Bian, 1997; Granovetter, 1995), through social connections, individuals obtain valuable information and overcome structural barriers in the labour market, which may improve the effectiveness of the job search process and subsequent earnings. In the simulation model, the expected tourism income not only follows the net profit from each tourist and the number of tourist arrivals in Zoucheng, but is also limited by the absence of social capital. Low social capital levels mean fewer channels to access information and consult about operating rural tourism.

The livelihood outcome of the system is influenced by benefits derived from two areas: farming (agriculture) and non-farming activities (rural tourism, mining). Due to the dominant role of local mining, income from exploiting mineral resources is obviously higher than from agriculture and tourism. Therefore, income and employment in mining are assumed to be constant so as to simulate the possible alternative livelihoods (in agriculture and tourism) for other mining household members (excluding miners themselves). In the simulation model, there are two livelihood choices for mining households' members, including farming activities and rural tourism operation. The labour distribution is determined by the different expected incomes in agriculture and rural tourism. The model allows the identification of social capital effects on the development of an alternative livelihood (rural tourism).

7.3 The Empirical Model Components and Assumptions

Farming, coal mining, and tourism business are the most important activities supporting the well-being of the local community. The empirical model concentrates on these three livelihoods, their interrelationships, social capital applications, and simulated results (livelihood outcomes). The model integrates five subsystems, which are discussed in detail below. The key equations are given in each section. Furthermore, all key variables are defined as follows:

Coal Mining Module:

С	Total coal output	Ċ	Growth rate of coal output
Ē	Potential or maximum coal output	RG _C	Intrinsic growth rate of coal output
Agricul	tural Production Module:		
A	Agricultural output per household	À	Growth rate of agricultural output per household
Ā	Potential or maximum agricultural output	P _A	Price of agricultural output per kilo
QF	Farmland quality per household	PL _A	Mining effects on farmland quality
Spla	Scaling constant of coal mining effect on farmland quality	RG _A	Intrinsic growth rate of agricultural output
NA	Number of labourers in farming	Sna	Scaling constant of labourers in farming on growth rate of agricultural output
Tourisr	n Module:		
Т	Number of tourists	T	Carrying capacity of tourist arrivals
Ť	Growth rate of tourist arrivals	RG _T	Intrinsic growth rate of tourist arrivals
INV	Government investment in public infrastructure or relevant tourism facilities	Y	Sectoral output value from Y_C (coal mining), Y_A (agriculture), and Y_T (tourism)
SLT	Scaling constant of tourism quality on growth rate of tourist arrivals	NT	Number of rural tourism operators
Splt	Scaling constant of pollution effect on tourist arrivals	PL _T	Mining pollution on local atmosphere
S _{invt}	Scaling constant of government investment effect on the intrinsic growth rate of tourist arrivals		
Social (Capital Module:		
RS	Residence period	SK	Social capital
Livelih	ood Outcome Module:		
π_{A}	Crop profit	PC _A	Cost of agricultural output per kilo
$\pi_{ m T}$	Net profit per person in tourism	DI	Relative expected income difference between farming and tourism
I	Total expected incomes of I_A (agriculture) and I_T (tourism)	L	Total workers

S _{SKI}	Scaling constant of social capital	STIDT	Scaling constant of tourist effect on
	effect on tourism incomes	STIKI	expected income in agriculture.
P _A	Price of agricultural output per kilo	TIR	Tourism intensity rate

7.3.1 Coal Mining Module

The coal mining subsystem mainly comprises two linked variables: overall growth rate per unit time of coal output (\dot{C}) and total coal output (C). The growth of coal output is defined here inclusively to present the dynamic process of coal production, which further impacts on farmland quality and air pollution in other subsystems.

The length of service remaining for the nine coal mines in Zoucheng city is more than 15 years (See Chapter Two). The time span is 10 years, which is within the length of service remaining for local mines because mineral resources are finite. There is also a five-year term for central and local officials, who are normally selected for two successive terms in China's administrative system. It is plausible to make the ten-year institutional arrangements, policies, and interventions based on local resources for local government, which can help the rural households achieve more secure and diversified livelihoods.

Growth factors such as labourers, technical investment, machinery and equipment influence coal output. In the simulation model, coal output is changed by a general growth rate (intrinsic growth rate) that combines these effects on the growth of coal production. The intrinsic growth rate (RG_C) is based on the collected historical annual time series data of Zoucheng city. On the other hand, coal output should be subject to an upper limit (\overline{C}). For instance, available technology and fixed amount of coal in a mine may limit the mining capacity.

Coal mining yields a great variety of products of varying quality, including coking coal, steam coal, and washed coal; all of these coal products have different prices. As with the farm product sector, the model uses one representative product for mining.

The growth function of coal output in this subsystem is based on logistic growth. American geophysicist M. King Hubbert first introduced a logistic equation to estimate the peak and lifetime production for oil (Hubbert, 1956). Based on Hubbert's study, many subsequent studies (Lin & Liu, 2010; Tao & Li, 2007; Hook, et al., 2010) have used logistic growth curves to simulate the historical data of coal production and predict coal production in the future. However, this production model neglects the impact of economic cycles and other factors (Lin & Liu, 2010). Focusing on the main purposes of this thesis (see Chapter One), this simulation model concentrates only on social capital application in livelihood development and assumes other macroeconomic environmental factors such as the effects of economic cycles and inflation remain constant.

The coal mining subsystem in this simulation will use logistic equations to simulate the coal output. The production of coal fields is assumed to generally resemble logistic growth. With the input of equipment, labourers, and technical investment in coal mining, the initial stage of growth is approximately exponential; then, as some limitations arise such as technological issues and the fixed amount of coal, growth slows, and at maturity, growth stops. The coal mining module is represented in the simulation by the two-equation system given in Eq. (1) and Eq. (2) determining coal in the local mining industry.

The key simulation equations for Coal Mining Subsystem are given below.

(1) Overall growth rate of coal output per unit time (tons/unit time):

$$\dot{C} = RG_C(1 - \frac{C}{\overline{C}})C$$

(2) Total coal output over time (tons):

$$C = \int \dot{C} dt$$

7.3.2 Agricultural Production Module

Many previous studies (e.g. Gradiz et al., 2007; Van Ittersum et al., 2008; Herrero et al., 2007; Castelan-Ortega et al., 2003; Keating et al., 2003; Moore et al., 2007; Van Wijk et al., 2009; Tittonell et al., 2007, 2008; Rufino et al., 2007) have developed and evaluated farming simulation models to assess the effects and characteristics of smallholders' diverse agricultural production. These models simulated a range of systems for farming such as the dynamic linkages among crops, livestock, and common land for cropping and grazing and relevant socioeconomic components. All of these simulation models were based on specific scenarios (e.g. Gradiz et al, 2007) or geographic region (e.g. van Lttersum et al., 2008). These previous simulation models show that it is important to integrate scientific and practical knowledge as well as background data and information for the target region and livelihood strategies for local inhabitatns.

In this study, the agricultural production subsystem tightly couples agricultural output per household (A) with intrinsic growth rate (RG_A) and the effect of mining on farmland quality (PL_A). It captures the dynamics of changes in total agricultural outputs per household. The development of coal resources and the intensity of coal resource exploitation may increase the number of collapsed areas on farmland, the pH of soil, and the effect of dust wastes on farmland quality. Thus, two issues are simulated in this subsystem: (1) The growth of agricultural output per unit time; and (2) The relationship between the quality of agricultural land and the increase in local mining production.

In many developing countries, crop cultivation and animal husbandry are integrated components of a crop-livestock system (Parsons et al., 2011). For instance, more than 66% of the rural farmers worldwide rely on crop-livestock systems for their livelihoods (ILRI, 2000). The interactions between crops, livestock, and natural resources are complex in this kind of farming system (Parsons et al., 2011). However, crop and livestock activities are driven by the same resources, including land, labour,

skill and capital (Lead, 2007; Parsons et al., 2011). Thus, several assumptions are possible and necessary in order to simplify the agricultural subsystem in the simulation model.

The time step is controlled by the "harvest seasons" of crop-livestock integration. Farmers in North China generally have two harvest seasons together with another one resulting from feeding poultry and livestock. The three-stage income is consistent with the time steps of the model, which are set to be four-monthly. Therefore, the time step (t) of the model is based on farming activities. It is postulated that the agriculture sector is following a four-monthly cycle.

As discussed in Chapter Two, there are many farm products in Zoucheng. However, a diverse crop-livestock system or diverse farm products potentially limit the ability to dynamically simulate scenarios. The variety of farm products in Zoucheng in terms of crops planted or types of livestock raised increases the complexity of the model in comparison to simulating one specific agricultural product. For instance, different crops have different growth rates and yields. Looking at the popularity rate and preference for corn and wheat in Zoucheng (Zoucheng Statistical Yearbook, 2000-2013), planting corn and wheat can be assumed to be a basic income source for local farmers. Finally, most of the respondents prefer to also plant some other crops such as fruits and vegetables, and breed livestock such as chickens, pigs, and goats in order to diversify their agricultural outputs, which may consequently increase their incomes from farm activities.

The different prices, demands, supplies in the market, growing periods, and investments in diverse farm products, all enable the model to make assumptions and create a representative farm product. The yield of the representative farm product in the model is based on the average yield of wheat and corn in this area. The reason is that these two crops are the preferred choices in North China and Zoucheng. Farmers in North China generally use a large area of farmland to plant corn and wheat. In general, farmers in North China, including Zoucheng, also use corn to feed poultry and other livestock. Therefore, any effects on these two agricultural products (corn and wheat) should influence the total agricultural outputs and incomes for rural households.

In the simulation model, the price of the representative farm product will influence the expected income from agriculture. However, there are lots of agricultural products excluding corn and wheat, which are normally considered as economically viable crops in China. These crops can bring extra income or benefits to rural households. Therefore, I took account of these planting preferences in the Zoucheng area (e.g. by obtaining total planting areas for relevant agricultural products in Zoucheng from the local Yearbooks 2000-2013) to create a price for an assumed, representative farm product. First, the representative farm product in the simulation model takes the market price of corn and the price of wheat as the basic reference price due to the widespread planting of these two crops locally. Based on the prices of corn and wheat, the simulation model also takes account of market prices for other crops such as strawberries, apples, peanuts, cabbages, spinach, and livestock such as chicken to increase the price of farm products in the simulation model until the crop profits are similar to actual levels in Zoucheng. Therefore, it is assumed that all farmers in the model prefer to sell their agricultural products at this "mixed price" as the initial price in the market simulation.

The growth rate of agricultural output is the most important part of the agricultural subsystem. Guan, Gao, & Fukahori (2010) considered the value of farming production, animal husbandry, forestry, and fisheries as original data for renewable resources in their renewable resource subsystem. In this study, the agricultural production module will follow Guan et al.'s (2010) general structure, i.e., logistic growth. The growth of agricultural production has been estimated using an intrinsic growth rate (Li et al., 1994). The intrinsic growth rate describes the natural growth in farm output. As farm products mature and reproduce themselves, the amount of land

and physical capital are assumed constant. Of the other variables, only labour input and pollution factors can change the growth rate of agricultural output.

The model in this study did not consider the effects of natural shocks (e.g., natural disasters) on farm production, in order to avoid the effects of uncertainties on simulation results. Ellis (2000) and Chambers & Conway (1992) define shocks as unpredictable, traumatic and sudden events in on people's lives. These events can directly destroy livelihood assets or other relevant capitals. Any natural disasters or shocks may create vulnerabilities for rural dwellers. Thus, one of the assumptions is that there are no natural shocks to any products during the simulation period.

By simulating agriculture in line with the constraints discussed above, the growth rate of agricultural output per household (Å, see Eq. (3)) is based on the intrinsic growth rate for agricultural crops (RG_A), agricultural output per household (A), potential or maximum agricultural output (\overline{A}), labour input (N_A), and coal mining pollution effects on farmland quality (PL_A). The dynamic process of agricultural production over time is that more growth will lead to more seeds and other inputs. More seeds or other inputs in the next planting season mean that agricultural output will be increased. Variations in the number of agricultural labourers will change the intrinsic growth rate of agricultural outputs and coal mining activity results in outcomes including farmland degradation, high soil pH, lack of microbial activity, and poor water quality (see Chapter Five), which reduce the growth rate of agricultural output. Taking these factors into consideration, agricultural output per household (see Eq. (4)) is generated. All of the relevant coefficients are discussed in Section 7.5.2.2.

The key simulation equations for the Agricultural Production Subsystem are given below:

(3). Growth rate of agricultural output (kg/household): $\dot{A} = RG_A \left(1 - \frac{A}{\overline{A}} - S_{PLA}PL_A + S_{NA}N_A\right)A$ (4). Agricultural output per household (kg/household):

 $A = \int \dot{A} dt$

As discussed in Chapters One, Three and Five, coal mining causes surface subsidence to increase, and degrades the farmland quality. As a result, there is a negative relationship between mining activities and agricultural output. In Equation (3), the effect of coal mining pollution on farmland quality is an important factor influencing the growth of agricultural output. In this simulation model, coal output represents mining activities, which impact on farmland quality and in turn affect the growth rate of agricultural output.

Farmland quality is represented by "effective" farmland size. For instance, coal mining brings dust waste into the area, which degrades the quality of farmland by, for example, altering soil pH. With this relationship between coal production and "effective" farmland size, coal pollution on farmland quality per household (PL_A) is presented in Equation (5). In the simulation model, Equation¹⁸ (5) is based on relevant data. All of these data were collected from local Yearbooks.

The key simulation equations for the Agricultural Production Subsystem is given below:

(5). Effect of coal mining pollution on farmland quality:

 $PL_A = \gamma(C)$, where $\gamma = LOOKUP$

¹⁸ The equation will use the LOOKUP function. In Vensim, the WITH LOOKUP function is a very convenient function to specify a nonlinear relationship between two different variables or to construct a situation with more than one breakpoint. Between the specified pairs of values, a range of straight lines can be drawn and the LOOKUP function calculates intermediate values (Kirkwood, 1998). Appendix B presents the specified graph of the values that were used in the LOOKUP function in this thesis.

7.3.3 Tourism Module

The tourism subsystem primarily simulates the growth of tourist numbers in Zoucheng city. The number of tourist arrivals or tourist population is the most common variable used to accurately measure and forecast tourism development (Song & Li, 2008; Nejad & Tularam, 2010). In addition, an accurate forecasting of tourist arrivals will help the government to establish and carry out appropriate tourism-related strategies for long-term success (Australian Bureau of Statistics, 2009).

This subsystem comprises three linked components: government investment in public infrastructure or facilities (INV); pollution effects (i.e., industrial dust discharged) on the quality of the natural environment (PL_T); and the effect of tourism operators (N_T) on tourism service quality.

As discussed in Chapter Two, government support is important in tourism operations and development in Zoucheng. In general, tourism models or relevant studies consider government investments such as infrastructure, transport, services offered by the tourist sites and the quality of the natural environment in defining the attributes of a tourist product (Hernandez & Leon, 2007; Casagrandi & Rinaldi, 2002; Nejad & Tularam, 2010). Government investment in public infrastructure, transportation, information guides and relevant facilities may increase the attractiveness of local tourist sites (Oliveira, 2003; Khadaroo & Seetanah, 2008; Wilson, et al., 2001; Lohmann, et al., 2009). For instance, major constraints and challenges in rural China include difficult access to the transport infrastructure, lack of public transport systems, and bad road conditions, which may limit the mobility of residents and even reduce the attraction of the location for visitors in rural China (Dalkmann, et al., 2008). Government investment is crucial for economic and tourist development in rural areas. For instance, the government helps to provide funds and develop tourism products, such as cultural festivals; this preservation of tradition contributes to the maintenance of rurality (Zou, et al., 2014).

132

In general, tourists tend to demand high quality in tourism services and the natural environment. The number of tourism operators (i.e., N_T) affects the development of service quality, which eventually influences visitors' long-term satisfaction and growth in tourist numbers (Hsiao & Yao, 2012). Poor environmental quality tends to reduce a region's attractiveness to tourists (Hu & Wall, 2008; Mihalic, 2000; Johnston & Tyrrell, 2005). The dust generated and discharged into the atmosphere and emission factors can be used to estimate the likely level of air pollution in mining areas (Ghose & Majee, 2000). Industrial dust discharge is a typical environmental issue in Zoucheng, which is relevant to coal production in the simulation model. It is estimated based on total coal output (C). With the growth of total coal output and thus industrial dust discharge (i.e., PL_T) in Zoucheng, the attraction of local tourism is reduced and the total tourist numbers per unit time will also be reduced.

Government investment in public infrastructure (INV on Eq. (6)) is predicted from total sectoral output value (Y) in three industries: mining (Y_C), agriculture (Y_A), and tourism (Y_T) and the ratio (0.0631, see Section 7.4.3.1 for the calculation) of public infrastructure to total generated industrial value. In the current situation, the government investment is purely maintenance. Thus, the simulation model uses the ratio of maintenance expenditure in terms of local GDP to avoid overestimating actual situation. Furthermore, overinvestment in public infrastructure may attract more tourists in the simulation model, which could generate a different scenario. Therefore, the ratio of total government investment in public infrastructure is assumed to be constant in Zoucheng. Eq. (6) also indicates that government investment in public infrastructure changes with the growth of total generated industrial value in the simulation model. With more profitable industry development such as mining, local government will invest more in the local public infrastructure system.

The key simulation equation for the Tourism Subsystem is given below.

(6). Government investment in public infrastructure (yuan):

 $INV=0.0062(Y_{C}+Y_{A}+Y_{T})$
This subsystem simulates the time period of tourist arrivals. It is important to understand the growth patterns in tourism. The resort or tourism area lifecycle and tourist product lifecycle has been empirically and theoretically discussed in Butler's lifecycle model (1980). Butler's lifecycle model remains the most enduring heuristic to understand the evolution of tourist destinations (Pearce, 2001; Butler, 2006), which has been widely applied and accepted as a conceptual framework for analyzing the historical progression of tourism area from a variety of perspectives (Buhalis, 2000; Cole, 2009; Lundtorp & Wanhill, 2001; Hovinen, 2002; Berry, 2006; Haywood, 2006). The model proposes a growth pattern for tourist destinations, which includes several stages-exploration, involvement, development, consolidation, maturity and stagnation (Butler, 1980; Lundtorp & Wanhill, 2001).

In the first stage (exploration), only a few tourists travel to the destination due to undeveloped facilities and a poorly developed related infrastructure system. The next stage involvement is characterised by an increase in tourist numbers. It is the first stage at which tourists start to become aware of the destination as a tourist base. New businesses such as food and catering, accommodation, leisure activities and site tours begin to spring up. Local residents begin to realize that tourism could be a profitable activity for their livelihood choice. However, initially inadequate infrastructure systems such as roads limit the development of the tourist industry.

In the development stage, the destination's tourism facilities become well developed and the tourist numbers steadily increase. The local tourism industry and economic activities require skilled personnel and general labourers such as activities related to the natural attractions of the site, cultural attractions, restaurants, and other physical attractions. Thus, government investment in detailed advertising and promotional campaigns is necessary to increase the level of tourist knowledge about such destinations.

In the consolidation stage, the tourists number continues to grow, but at a declining rate. The destination becomes well known and the novelty of the destination may decline over time for tourists who have already visited the destination a number of times (Nejad & Tularam, 2010). Therefore, these potential tourists do not list the destination as a priority. The highest cumulative number of tourists is achieved in the stagnation phase. However, the resort provides nothing new to be discovered. For instance, all tourists know about the destination and its facilities. The destination may face a number of issues involving the environment (e.g. pollution), culture (e.g. culture shocks), and social problems (e.g. crimes) (Nejad & Tularam, 2010). Local government still needs to control and/or regulate these issues. At this stage the destination has two possibilities, decline or rejuvenation. Using Butler's lifecycle theory and framework, many studies apply logistic growth to represent tourist numbers over time (Nejad & Tularam, 2010; Lundtorp & Wanhill, 2001; Cole, 2009; Casagrandi & Rinaldi, 2002; Hernandez & Leon, 2007). Therefore, the simulation model in this study also applies the logistic equation (see Eq. (7)).

Within the logistic equation, the intrinsic growth rate of tourist arrivals (RG_T) is directly influenced by government investment in public infrastructure in the simulation model. More rural tourism operators may improve the quality and service associated with the tourist destination. This leads to an increase in tourism: long-term satisfaction and loyalty enhances potential tourists' intentions to visit Zoucheng city. This consequently increases the number of tourists per season, creating positive feedback on tourist growth in the simulation model. Thus, the factor of tourism operators providing services in Zoucheng was included in Eq. (7). Hsiao et al., (2006; 2012) analysed and simulated a range of service quality attributes in a tourist attraction in Taiwan. Considering the similar traditional culture and behaviours in these two communities, the Vensim model for the tourism quality increment in this study was similar to Hsiao's coefficient of workers on service quality increment; therefore the coefficient relating to the effect of tourism operators on tourism service quality is assumed to be 0.0016. A more accurate coefficient for a county-level city may be derived in future studies. The key simulation equations for the Tourism Subsystem are given as below:

(7). Growth rate of tourist numbers:

$$\dot{T} = RG_{T} \left(1 - \frac{T}{T} + 0.0016S_{LT}N_{T} - S_{PLT}PL_{T} \right) T$$

where:

 $RG_T = S_{INVT}INV$

PL_T is mining pollution of the local atmosphere

N_T is the number of rural tourism operators

(8). Number of tourists

$$T = \int \dot{T} dt$$

7.3.4 Social Capital Module

The livelihood approach states that households' disposable assets may impact on the type of activity undertaken and the amount of income earned (Barrett et al., 2005; Brown et al., 2006). Access to livelihood assets is an important factor in determining the choice of household livelihood strategy. In the simulation model of this study, social capital is expected to support an alternative livelihood strategy for rural residents.

This subsystem simulates the factors that control the growth of social capital (SK). The growth of social capital takes one flow variable (i.e., residence period) as a determinant. In system dynamics, variation in the length of residence (RS) represents the characteristics of the dynamics changes in the social capital module, which is more suitable than using non-continual movement variables such as gender. The longer respondents have lived in Zoucheng, the more opportunities they have had to access a greater number of people, interact frequently, and build up long-term relationships in the local community, thus acquiring information through friends and their social circles (Gertler, et al., 2006; Onyx & Bullen, 2000). Policies could empower people to engage in new forms of bridging and social capital by

participating in networks such as livelihood development groups (Kah et al., 2005), which could enable people to participate in alternative livelihood choices (e.g. non-farm activities) (Anderson et al., 2002).

In the simulation model, social capital applications can widen household members' access to alternative livelihoods and incomes in tourism. This draws out some of the links between social capital and livelihood choices. Working and operating in tourism businesses is more complicated than traditional agriculture (He & Chen, 2012). With the growth of social capital, the simulation model considers the relationships and social networks needed to access information and skills about issues such as tourism services, tour guiding, and business management. These social circles can improve the ability of local people to serve outside visitors and consequently impact on tourism incomes in the simulation model. The whole process is consistent with DFID (1999): social capital will improve people's incomes indirectly.

The social capital subsystem captures the dynamics of changes in social capital. Social capital (SK) is linked with the length of residence (RS) in the simulation model. By following the coefficient of RS in the regression model (see Chapter Six) and the characteristics of survey participants such as average years of residence (see Chapter Four), social capital growth with RS is shown in Eq. (9). The initial value of social capital (SK₀) is based on the average length of residence in the city or surrounding rural area. With increasing RS, social networks are likely to expand and further enhance SK. However, local residents may have a limited ability to access additional social capital resources because homogeneous networks (e.g. living in a small village for a long time) reduce the effect of networking as a way of accessing social capital (Lin, 2004). Using a ten year RS in this simulation model, and the distribution of the calculated social capital scores from Chapter Four, we obtain reasonable simulated results (see Figures 8.5 & 8.15).

The key equation for the Social Capital Subsystem is given below.

(9). Social capital:

$$SK = SK_0 + 0.006RS$$
 (0 ≤ RS ≤ 30)

where SK₀ is the initial value of social capital.

7.3.5 Livelihood Outcome Module

This subsystem addresses incomes and employment in agricultural and tourism activities. It may provide some policy suggestions and planning strategies for local development at an equilibrium point by analyzing the effects on incomes and worker migration between agriculture and rural-tourism. These results also suggest possible alternative livelihood choices for mining household members in rural areas.

The income from agriculture (I_A) is determined by crop profits (π_A), which vary with changes either in total agricultural output per household (A), or in the price of agricultural outputs (P_A) and the cost of agricultural production (PC_A). Simulations of agricultural output were discussed in the Agricultural Production Module (Section 7.3.2). The values of initial costs and the initial price of agricultural output are discussed in Section 7.4.5.1. With respect to government influences on local agricultural development (discussed in Chapter Two) and the small area (a county level city) under study, variations in unit price and unit cost received from different amounts of outputs are assumed to be fairly small.

Excluding the effect of total agricultural output per household (A) on the price of agricultural output (P_A), tourism raises agricultural growth by impacting on the price of agricultural outputs in the simulation model. In the simulation model, it is a particularly important way of increasing the total income from farming activity when crop price is assumed to be relatively insensitive in the initial local market. Increased tourist demands will stimulate the price of representative agricultural outputs. In the simulation model, the effect of tourism on the price of agricultural product is based

on the tourism intensity rate (TIR). TIR ($TIR = \frac{Tourists}{Population size}$) compares the relationship between tourist arrivals and the population size (Toops, 1999) and is a way to measure the impact of tourism on an area (Zhang & Zhang, 2003; Keyim et al., 2005). Thus, we employ the TIR to define the tourist effect on agriculture in Zoucheng. In the simulation model, the price of agricultural outputs is varied with the TIR as well.

Equations (10) and (11) show the price and cost of agricultural outputs responding to changes in agricultural output respectively. Equation (12) shows the net profit from an agricultural livelihood.

The key equations for the Livelihood Outcome Subsystem are given below:

(10). Price of agricultural output (yuan/kg): $P_A = (5.13 - 0.0001A)(1 + 0.594TIR)$

(11). Non-wage cost of agricultural output (yuan/kg):

$$PC_A = 1.24 - 0.0002A$$

(12). Expected income from agriculture (yuan/kg):

$$I_A = \pi_A = A(P_A - PC_A)$$

(As noted above, the parameters of the equations either have been, or will be, discussed in earlier or subsequent sections.)

As explained in Chapters Two and Three, rural tourism operators can provide multiple farmland-based activities including picking fruit and other crops, rural homemade meals, rural natural environment entertainment and adventure to tourists. In the market, more tourists might bring impacts such as contributions to GDP, increasing the employment rate, and creating new sources of household income (Wei, Qu, & Ma, 2013; Nejad & Tularam, 2010). In the simulation model, the number of tourist arrivals is one of the factors determining the expected incomes from tourism. Each tourist is assumed to bring a constant net profit (π_T). Social capital (e.g., social networks) is another variable that stimulates the tourism incomes. Rural household members can access additional information about operating rural tourism ventures, customers' requirements, and service skills through their social capital. The coefficients of Equation (13) will be discussed in Section 7.5.5.1.

The key equation for the Livelihood Outcome Subsystem is:

(13). The expected income from tourism:

$$I_{\rm T} = T(1 + S_{\rm SKI}SK)\pi_{\rm T}$$

Grasso (1998) and Rohorua (2009) used a mathematic model to determine the optimal relationship between economic and the ecological development. Using their approach, labour distribution is based on the different incomes of the involved industries. As mentioned in Chapter Two, high incomes in mining leading to more labourers working in this industry is one of the most notable characteristics of local livelihoods in Zoucheng city. Therefore, the allocation of workers is dependent on their relative incomes at different times. For instance, at any given time, the more income the industry provides, the more workers it attracts.

Considering our model and analyses, some scenarios and assumptions need to be set in terms of the process of labour allocation. First, labour mobility is determined by the profitability of the activities. Second, there is no switching cost for farmers to switch between tourism and agriculture. Third, there is a fixed number of workers. Fourth, agriculture and tourism are independent or are separate activities. Finally, the simulation model avoids the issue of influx of labourers from other cities when alternative activity (i.e., tourism) is proven to be viable. With these assumptions on worker allocation, the simulation avoids "outside shocks" such as extra labourers travelling from other cities to influence the simulated results of livelihood outcomes. This module derives the distribution of workers as a result of livelihood choices. Different livelihood choices result in different incomes. Rural-tourism activity is an alternative livelihood for some rural households in the simulation model, which allows household members to become self-employed tourism operators. Therefore, the simulation model presents the results of people moving out of farming into tourism. Regarding relative income differences (D_I) between agricultural activity (I_A) and rural-tourism (I_T), the labour distribution between farming (L_A) and tourism (L_T) is distributed between these two activities.

The key equations for the Livelihood Outcome Subsystem are:

(14). Income difference between farming and rural-tourism activities:

$$D_I = \frac{I_T - I_A}{I_T + I_A} \qquad -1 \le D_I \le 1$$

(16). Rural tourism operators:

$$L_{\rm T} = \frac{L(1+D_{\rm I})}{2}$$

(17). Total farming employment:

$$L_{A} = \frac{L(1 - D_{I})}{2}$$

7.4 Specification of Model Parameters and Validation

There are three approaches to specifying the values and interrelationships in the simulation model: (1) Econometric approach: parameters are based on statistical analyses of survey data in this thesis; (2) Background: in some similar study areas in the literature, available historical data and information have been used to calibrate the model; (3) Mathematical programming: some values use in-built functions (e.g. 'IF THEN ELSE' loops, calculus, etc.). Most available historical data are based between 2006 and 2010 so as to correspond to the year (i.e., 2010) that the household survey was undertaken. Five years is the normal period for central and local government to implement their main policies and programmes in China. The scaling constant effects

in equations were estimated based on helping the model achieve sensible results that matched the actual environment. Appendix A evaluates and tests the sensitivity of model to determine its representativeness and authenticity.

7.4.1 Coal Mining Module

The values of parameters in the coal mining module are based on relevant data for coal production from local Yearbooks between 2006 and 2010 (Zoucheng Statistical Yearbook, 2006-2010) and other studies in Zoucheng (e.g. Yang et al., 2007; The conciliation commission of regional economies, 2006, Dec).

Based on the historical data¹⁹, the initial coal production (39.2 million tons) is configured to the mean value of coal production from 2006 to 2010. According to YanZhou Mining Group (2014), the maximum coal production is about 50.5 million tons in the simulation model. The general growth rate of coal production was calculated from the compound growth rate of coal production from 2006 to 2010. The general growth rate is about 0.11%.

7.4.2 Agricultural Production Module

The subsystem considers farmland quality and agricultural productivity as the source of total agricultural output per household.

7.4.2.1 Farmland

Farmland quality refers to 'effective' farmland size. Therefore, the farmland section is divided into two modules: farmland size per household and changes in farmland size per unit time. The initial set value of farmland size was obtained from the mean value of data collected from local official government statistics, the household survey conducted in this study and other published literature.

¹⁹ Appendix C provides summarized table data, some of which were used in this simulation.

The mean value of the farmland size per household in the household survey in this study is similar to that in other literature. From local Yearbooks (2006-2010), farmland size per household was around 4 mu. Another report indicated that the local farmland area was about 4.3 mu per household (The conciliation commission of regional economies, 2006 Dec). The mean area of farmland in Zoucheng was based on the household survey for this study, about 4.46 mu. According to other investigations of farmland area per person (National Bureau of Statistics of China (NBSC), 2000; Li, 2000), the area was about 4.64 mu per household in Shandong province, which is slightly higher than the survey data in the present study (i.e., 4.46 mu). Thus, a plausible initial farmland size for the simulation model was taken as 4.5 mu per household.

Pollution effects on farmland quality (i.e., PL_A, see Equation (5)) are assumed to follow total coal output. The relationship between coal mining activities and farmland quality is based on the raw data from the Zoucheng Statistical Yearbooks 2006-2010.

7.4.2.2 Agricultural output

Agricultural output is measured in kilograms per household. Using the local grain yield for the horizon of 2006-2010 (Zoucheng Yearbooks, 2006-2010), the average value of the grain yield was about 2,050.88 kg per household in Zoucheng. This result is based on the average value of the grain yield per mu between 2006-2010 and the initial farmland size per household discussed in the last section (455.75×4.5). According to several experiments in Zoucheng (Sun et al., 2009), the maximum agricultural outputs for wheat and corn were about 653 kg per mu and 924 kg per mu respectively. Considering this range of grain yields and the flexibility of the simulation model, potential or maximum agricultural output (\overline{A}) was set at 3,546 (788 $\times 4.5$) kg. The average grain yield (i.e., 2,050.88 kg) is considered as the initial value for agricultural production.

The growth of China's agricultural production per annum fluctuated from 0.5-4% based on annual grain yields across all provinces from the 1950s to the 1990s (Fan & Zhang, 2002; Liu & Wang, 2005; State Statistical Bureau, 1997). With regard to the main agricultural products (wheat and corn) in this simulation model, Carter, Chen, and Chu (2002) measured the growth of agricultural productivity in China by using national data and provincial aggregate data. The annual growth rate is wheat production was about 1.3% from 1978 to 1996. The annual growth rate in corn was production was about 2.5% in the same period.

In the simulation model, the intrinsic growth rate of agricultural output (RG_A) is calculated by using the data from local statistical yearbooks (2006-2010). During the period 2006 to 2010, the grain yield in 2009 was decreased to 459 kg per mu because of drought and freezing temperatures, which drove down the average growth rate in Zoucheng (Zoucheng Statistical Yearbook, 2006-2010). Based on the mean value of growth rate in the local grain yield, RG_A was about 0.7%.

7.4.3 Tourism Module

This module mainly simulates the number of tourist arrivals (T). This is influenced by government investment in public infrastructure (INV), the number of rural tourism operators (N_T), and the pollution factor's (PL_T) impact on tourist growth. The number of tourism operators (N_T) will be discussed in Section 7.4.5.2.

7.4.3.1 Government investment in public infrastructure

According to local government research (March 2008), if local tourism was expected to develop at a certain scale, the investment required was about 36.2 million yuan. As mentioned in the previous discussion in Section 7.3.3, government investment is a very important factor in stimulating tourism development. The simulation model is based on the current situation and does not focus on additional investment in local tourism. The main purpose of including government investment in this subsystem is

to help estimate an actual level of tourist growth and avoid overestimating local tourism development.

The model allows for the simulation of an aggregate value of production in three industries (agriculture, mining, and tourism). The total values are calculated by adding all of these generated industrial values together. Then, a proportion of total generated values are used in the field of local tourism support including maintenance of facilities in the local city, transportation systems, business promotion and information gathering in the market. All of these investments are generally named "government investment in public infrastructure (INV)" in the simulation model.

The sectoral output value of coal mining (Y_C) follows the total coal output (C) and coal price per ton. The coal price per ton was obtained from the official website of the Coal Industry Bureau of Shandong Province²⁰, which lists the products of the coal chemical industry in China, and China's coal resources, at about 600 yuan per ton on average in the local region in recent years (figures are from 2014). This price is similar to that of run-of-mine coal. Run-of-mine coal is the raw material, normally consisting of coal, rocks, middlings, minerals and contamination, which is delivered from the mine to the coal preparation plant. Normally, a mine can produce a large amount of run-of-mine coal.

The sectoral output value of agricultural production (Y_A) is calculated from the total agricultural output per household (A), the price of agricultural outputs (P_A), and the number of households.

The sectoral output value of tourism (Y_T) is based on total tourist spending. The local government has collected data on tourist spending, using random field surveys (observations=500) in local tourist attractions. Most tourists tend to spend less than

²⁰Zoucheng city belongs to Shandong province. There are some news reports and data about coal prices, supply, transport costs etc available at http://www.sdmtw.com/portal/home. Two other sites provide the coal prices in relevant months in China (http://www.sxcoal.com/sdcoal/16027/0/listnew.html; http://www.chemcp.com/channel/meitan/)

500 yuan/visit (in about 40% of total observations) at local tourist attractions (Zoucheng government, March 2008). The simulation model uses the average value that most tourists tend to spend (i.e., 250 yuan). From the official website of the National Bureau of Statistics of the People's Republic of China²¹, the average consumer price index (CPI) was about 102.5 from 2008 to 2014. In the simulation model, each tourist is assumed to spend 270 yuan per unit time²² (i.e., 4 month periods), which is the average value of 250 yuan including the CPI effect from 2008 to 2014.

Government investment in public infrastructure (INV) is calculated as a percentage of the city maintenance expenditure in terms of local GDP. Figure 7.2 presents the yearly percentages in Zoucheng city from 2000 to 2010. Each year, the local government allocated funds for the maintenance of public infrastructures. The highest was 0.782% in 2010. In this subsystem, the investment ratio was based on the mean value (about 0.63%) of these percentages from 2006 to 2010.



Figure 7.2: The proportion of *city maintenance expenditure in terms of local GDP from 2000 to 2010*

Source: Author's compilation from Zoucheng Statistical Yearbook, 2000-2010

²¹ The official Web site of the National Bureau of Statistics of the People's Republic of China provides the yearly consumer price index (CPI) in China (http://data.stats.gov.cn/index.htm).
²² The calculated result is based on

$$\frac{(250+250\times(1+2.5\%)+250\times(1+2.5\%)^{2}+250\times(1+2.5\%)^{3}+250\times(1+2.5\%)^{4}+250\times(1+2.5\%)^{5}+250\times(1+2.5\%)^{6})}{7}$$

7.4.3.2 Discharged industrial dust

In this subsystem, the amount of industrial dust discharged changes with total coal production. The model used local data (Statistical Yearbook 2006 - 2010) to simulate the relationship between growth of coal output and the growth in industrial dust discharge (see Appendix B).

7.4.3.3 Number of tourist arrivals per unit time

The dynamics of the model assume that the number of tourists varies according to a rate that depends on three variables: intrinsic growth rate of tourist arrivals (RG_T), tourism service quality (L_T), and the effect of air pollution (PL_T).

The initial value of tourists was referenced from the mean value of the two most important tourist attractions (Meng Temple and Yi Mountain) in Zoucheng. Due to the time steps in the simulation model, the mean value was based on four-month periods. Using the official statistics for Zoucheng (Zoucheng Tourist Administration, 2011), the average number of tourists per four-month period visiting the Meng Temple and Yi Mountain was about 46,100 between 2006 and 2010. The initial tourist numbers in the simulation model starts at about 46,000.

The carrying capacity for potential tourists is calculated on the capacity of local scenic spots. According to the estimated method for scenic spot capacity (Zoucheng government, March 2008), the land area of the Mencius Temple is about 23,250 m² and the lengths of the tourist routes on Yi Mountain is about 8,000m. Each tourist has 75 m² in the Mencius Temple and 5m on Yi Mountain in 120 working days²³. The simulation model assumes each current tourist is occupying nearly two tourist spaces based on the carrying capacity for potential tourists at the Mencius Temple and Yi Mountain that will guarantee a good tourism environment in these areas. In other

²³ The calculated result is based on a total of 360 working days and three four-monthly cycles per year (i.e., $\frac{360}{3}$).

words, the Mencius Temple and Yi Mountain can reduce their initial carrying capacity (assumed to be 55% of initial carrying capacity) to provide more tourist spaces for potential tourists. Thus, the carrying capacity of potential tourists²⁴ per four-month period is 125,000 tourists in the simulation model.

7.4.4 Social Capital Module

In the simulation model, all of the coefficients for social capital are from the previous regression analysis in Chapter Six. It assumes that the collected social capital data and information represent the household level due to the dominant role of mining in mining households, because of the high incomes provided in this sector. In a mining household, a miner's social capital is assumed to cover the range of the whole family's social networks and these social resources could help household members to improve their access to relevant information.

The coefficient of RS is about 0.018 using the previous analysis of factors associated with access to social capital (see Chapter Six). However, the simulation model is based on four-month periods. The coefficient of RS as related to social capital is therefore estimated at 0.006. The initial value of social capital uses the statistical distribution of social capital index (see Chapter Four) and average years of residence (26 years) in Zoucheng. Because the simulation model involves rural households, the coefficient for rural area (-0.409) from the regression analysis (see Chapter Six) is also applied.

7.4.5 Livelihood Outcomes Module

The livelihood outcomes subsystem simulates two processes: expected incomes and worker allocation between agriculture and rural-tourism.

²⁴ The calculated result is based on $\left(\frac{23250}{75} + \frac{8000}{5}\right) \times 120 \times 0.55$

7.4.5.1 Income sources

Agricultural output per household (A), the price of agricultural outputs (P_A), and the cost of agricultural outputs (PC_A) are driving forces in identifying the expected incomes in agriculture.

The agricultural output per household (A) was discussed in Section 7.4.2.2. The costs of planting agricultural products and the net profits from agriculture in Zoucheng are used as references to construct the price of the representative agricultural product in the model. According to the official website of the Commerce Department of the Chinese Government and the State Administration of Grain²⁵, the average price of maize was about 3.3 yuan/kg, and wheat was about 2.3 yuan/kg in North China in 2014. However, other farm products exhibited high prices, including strawberries at about 10 yuan/kg, spinach at 6 yuan/kg, and some livestock such as chicken at more than 13 yuan/kg. As previously discussed, the price of agricultural outputs (P_A) in this model is mainly referenced on the lowest prices (i.e., wheat and corn) and driven up by the prices of other agricultural outputs in the market until the initial crop profits in the simulation model are similar to practical levels in Zoucheng. According to Wang et al. (2012), the net profits of agricultural output were around 1,800 yuan per mu when farmers chose wheat and corn as their main agricultural products in Zoucheng. Viewed in connection with farming activities in the household survey for this study, the average net profit of agricultural output per rural household was about 7,953 yuan per season (about 1,767 yuan per mu). The average cost of agricultural outputs was about 546.16 yuan per unit area (mu). Given the initial values (e.g., grain yield) set in Section 7.4.2.2, the price of agricultural outputs (PA) was set at 5.1 yuan per kg and

²⁵ These two official Web sites provides the daily and weekly fluctuating prices of agricultural products in China

⁽http://www.chinagrain.gov.cn:81/lm/scnew.asp?dq=all&leibie=3m&biaoti=all3m; http://nc.mofcom.gov.cn/channel/gxdj/jghq/index.shtml)

the cost of agricultural outputs per unit (PC_A) was set at 1.2 yuan per kg in the simulation model.

The initial expected income from tourism was based on background data and information in Zoucheng. As discussed in Chapter Two, average monthly net profit is about 1,080 yuan. Regarding four months as the time step and 46,000 as the initial tourist number in the simulation model, the net profit per tourist²⁶ is set at about 0.094 yuan/tourist. This simulation model sets the tourism income per worker to be around 1,000 yuan/worker at the initial stage of local rural tourism development.

Mining wages are assumed to be constant, which is referenced on the household survey in Zoucheng. On average, the mean value of the monthly wage for rural miners is about 2,700 yuan in rural households. Thus, the mining wage was set at 10,800 Yuan $(2,700\times4)$ per unit time in the simulation model.

7.4.5.2 The number of rural households and labourers

The livelihood choices functions (i.e., allocation of labourers) are based on the income differences between agriculture and rural tourism. The alternative livelihood (rural tourism) shows some people moving out of farming into tourism businesses, based on the expected incomes of rural-tourism activity. Total mining households in the rural area are set at 10,000 in the simulation model, as there are more than 500 rural households in each village town in Zoucheng. These approximately 500 rural households in each village town are expected to be the experimental pioneers. They may lead other rural households to develop alternative livelihoods in future, if rural-tourism is a suitable livelihood choice. The total population size is assumed to be one million. In the simulation model, total labourers, number of households, mining workers, and village towns are all held constant in the simulation model. Each rural household provides two labourers in the simulation model. Each household can

²⁶ The calculated result is based on $\frac{1080 \times 4}{46000}$

provide one labourer to do a full-time mining work, which means it is recognized as a mining household. Another household member could move between farming and tourism.

Finally, the initial labour allocation needs to be discussed so as to simulate the current status. As discussed in Chapter Two, the average ratio between tourism employees and workers in relevant mining industries is approximately one to three. Therefore, the initial number of tourism operators was set at around 3,400 against a total of 10,000 miners and the rest of the labourers were initially allocated to agriculture in the simulation model.

CHAPTER EIGHT: SIMULATION RESULTS AND DISCUSSION

After all of the variables, linkages, and relevant parameters are specified based on historical time series data, background information, and relevant regression analysis, the simulation can then be processed and performed.

The model presented here is based on local environment issues, relevant livelihood assets (e.g. social capital, natural capital), characteristics of rural households (e.g. resident area, resident period, initial incomes), and interactions between the three economic activities of agriculture, rural tourism and mining. Four scenarios are simulated: current risks of mining activity such as degrading the quality of farmland; the potential impact of developing tourism on additional livelihood activities (i.e., farming and rural tourism); the potential impact of a change in social capital on rural tourism activities (e.g., incomes, number of workers); and comparing viable strategies for local communities, including mining, farming, and rural tourism. All of these scenarios are simulated in four-monthly time steps and ten-years time periods (see Chapter Seven). Thus, there are 30 time steps, which represent the x-axis on all graphs in this chapter, in the simulation model.

The first simulation addresses the effect of pollution issues, caused by coal mining, on rural livelihoods and the number of tourists. The second scenario simulates the effect of a change in government policy on promoting tourism, which uses the simulated results of agricultural output, number of tourists, expected incomes in agriculture and tourism and labour distribution between these two activities. The third simulation looks at the impact of expanded social capital (i.e., SK+1; also see Section 8.3) on livelihood creation, which in turn highlights the role of social capital in facilitating the transition to alternative economic activities. The last simulation moves from two expansions of livelihood activities (farming and rural tourism) to discuss all three viable strategies for local rural households (mining, farming, and rural tourism).

The strengthened social capital in this scenario is still applied to rural tourism activities. By comparing the different incomes of mining, farming and rural tourism that form the different livelihood choices for rural households, the aim is to determine potentially sustainable livelihoods that can take the place of mining activity eventually. Moreover, the last scenario also simulates the situation of pit closures (i.e., C=0; also see Section 8.4) so as to highlight the importance of developing alternative livelihoods before the local mineral resource is exhausted.

8.1 Current issues arising coal mining

As discussed in Chapters Two and Five, coal mining is the dominant sector in terms of local economic development and livelihoods, which creates environmental issues such as air pollution and loss of land quality. Therefore, the implementation of regulations and policies cannot ignore the current environmental issues generated by coal production. The simulation model in this study must firstly present the current environmental issues in Zoucheng city. It was therefore seen as necessary to model the effect of current environmental issues on agriculture and tourism activities.

Increased coal production puts more pressure on both farmland and air quality, which may influence agricultural output and the number of tourist arrivals. Figures 8.1 and 8.2 focus on these two environmental issues, illustrating the simulated results of the effect of pollution on farmland quality and tourist numbers.

Figure 8.1 illustrates the situation and results. With increased total coal output (C), excessive dust and coal residues could change the pH of the soil, microbial activity and a number of other factors that cause corresponding reduction in the quality of farmland (Q_F). In the simulation model, the farmland quality unit is defined as "unit quality". Figure 8.1 shows that farmland quality falls from an initial 4.5 unit quality per household to about 4.33 unit quality after approximately 10 years. The reduced soil fertility leads to less total agricultural output per household (A). The growth of agricultural income may be driven down by this trend. With limited job opportunities

in mining (the model maintains a constant level of employment in the mining sector) and fewer alternative livelihood choices (e.g. without rural tourism), the members of mining households in rural areas would face challenges in increasing their incomes via farming activities.





Without any other supportive factors in place to improve air quality and tourism development, mining activities create increasing pressures on air quality, because increased coal output will also mean an increase in industrial dust discharged to the atmosphere. Poor air quality will have a direct impact on the perceived environment in the city, which, in turn, is predicted to reduce tourist arrivals from 46,000 to about 43,128 over the whole simulation period (see Figure 8.2).



Figure 8.2: Industrial dust discharged and tourist numbers

8.2 The potential impact of developing the tourism sector

In the simulation model, the number of mine workers (10,000 miners) and individual miners' incomes (2,700 yuan/month) are held constant. Excluding these mining workers, other household members need more livelihood choices to diversify their livelihoods in a sustainable way. This section concentrates on simulated livelihood outcomes by making tourism available as a factor in local households' livelihood development. The model results focus on the incomes from farming activity and rural tourism activity and the number of workers moving out of farming into tourism. Furthermore, agricultural output per household, crop price, tourist arrivals, and social capital growth are included in the simulation model to allow us to explore their linkages and interactions with the movements of agricultural labour into the tourism sector.

a) Labour allocation between agriculture and tourism

Figure 8.3 presents the initial allocation of workers (labourers). The growth trend for workers in tourism indicates the potential advantages of rural tourism activity. At the beginning of the simulation period, most of the workers are engaged in farming and only about 3,400 are active in the tourism industry. However, once tourism activity picks up, as a result of more tourist arrivals and the development of relevant businesses as a consequence of local social network activity related to tourism, the number of rural workers in tourism increases from 3,430 at the end of the first time period to 4,741 at the end of the simulation period. Correspondingly, in the face of increased livelihood choices (i.e., rural tourism), the number of workers engaged in farming declines dramatically over the same period. In other words, increased tourism activity and the use of social capital results in a decline in the number of farm workers. At the end of the simulated period (ten years) of operation, there are still more household members engaged in farming, in total, but the gap between farming and tourism worker numbers has gradually reduced. Figure 8.3 shown that the difference in labour allocation between agriculture and tourism changes from 3,139 workers initially to 517.



Figure 8.3: Workers allocation between agriculture and tourism

b) Tourism sector

Figures 8.4 and 8.5, using the cause and effect flows in the simulation model, illustrate the proposition that business operations created in the tourism sector, together with the growth in tourist arrivals and the associated increase in social capital, could form an alternative livelihood choice for the Zoucheng community due to the transfer of labourers from agriculture to rural tourism.



Figure 8.4: Number of tourists

As discussed in Section 7.3.3, the growth rate of tourist numbers is based on the logistic equation (see Equation (7) in Chapter Seven). It is expected to have several stages, such as exploration, development, maturity and stagnation, and consequently influence the development pattern for the number of tourists (see Equation (8) in Chapter Seven), such as from rapid growth to a constant or stable number. However, Figure 8.4 presents a linear path. The main reason is that the simulation model is based on the actual historical data in Zoucheng and no other stimuluses such as extra investments from local government or business operators (Section 7.4.3). Therefore, the simulated results (Figure 8.4) conform to the actual tourism development situations (at the exploration or development stage) in Zoucheng (see Chapter Two).

In Figure 8.4, the number of tourists grows rapidly, which provides a new income source for rural labourers in the simulation model.



Figure 8.5: Social capital

As discussed in Section 7.3.4 of Chapter Seven, social capital grows with increasing time of residence (i.e., RS) in Zoucheng in the simulation model. The length of time that a person has lived in an area has a positive association with access to social capital. The initial value of social capital (i.e., SK₀) is based on the average length of residence. According to the household survey of this thesis, the average duration in Zoucheng is 26 years and the average age is 35 (see Table 4.4). Given a ten-year time of residence in the simulation model, the years resident in Zoucheng are increased from 26 to 36 and age is changed from 35 to 45, which is a stable and solid age group in people's life (e.g. having a family and a steady job). Regardless of some shocks (e.g. reeducation, lottery winning, etc.) on social networks, each resident period (e.g. one year resident period) in Zoucheng is assumed to contribute an equal value to people's social capital. Thus, Figure 8.5 presents a linear movement due to no other effects on social capital in the simulation. Though, of course, social capital is not possible to expand infinitely, it is limited by the simulated time period. This simulation model

uses 10 years time (i.e., $0 \le RS \le 30$; also see Equation (9) in Chapter Seven) as one limiting condition for the accumulation of social capital. In the absence of other factors (e.g., education), each worker is assumed to have a stable social circle in local community after 10 years, the scarcity of materials for network resources or homogeneous networks may reduce the accessing better social capital (Lin, 2004). With regard to social capital growth theory (Lin, 2001), the calculated distribution of social capital scores in Zoucheng (see Chapter Four), the social capital growth trend and range (from about 0.059 to about 0.23) are reasonable. With the growth of social capital, workers can learn relevant skills and obtain relevant information about the operation of tourism businesses to further improve their career prospects and expected incomes in tourism. The specific impact of social capital will be discussed in the next section.

Figure 8.6 shows that growth in the number of tourists and in social capital will increase the expected tourism income. Hence, workers will move from agriculture to tourism (see Figure 8.3). Figure 8.6 illustrates that the expected monthly tourism income increases from an initial level of about 1,091 yuan per month to around 2,150 yuan per month after 30 time units. The initial income in the simulation model is very close to the present average tourism income in Zoucheng (discussed in Chapter Two).

Figure 8.6: Expected tourism income (monthly wage)



c) Agricultural sector

The growth in tourist arrivals not only benefits expected tourism incomes but the profit from farm products. This happens because an expansion in the number of tourist arrivals will raise the price of representative farm products in the simulation model. As a result, farmers will earn more in the market.

The simulation model links all the causes and effects of the tourism sector to the agricultural sector. Figures 8.7 to 8.12 compare the simulated results for the agricultural sector with and without the effect of the tourism sector.

Without the effect of the tourism sector in the simulation model, the growth rate of farm products maintains an actual level. The current growth rate may be limited by farmland quality and therefore the potential capacity for agricultural output. However, farmland quality is negatively affected by local coal output in the simulation model. Thus, the growth rate of farm products shows a downward trend, falling from about 7.6 kg/unit area to around 7.14 kg/unit area (see Figure 8.7), while the agricultural output per household increases from 2,051 kg/unit area to almost 2,265 kg/unit area

(see Figure 8.8). The crop price and cost are driven down by slow increase of total agricultural output per household. As discussed in the initial assumptions (Section 7.3.5), the effect on unit price and cost of agricultural output per household is fairly small, moving from about 5.08 yuan/kg to about 5.078 yuan/kg for crop prices and from about 1.149 yuan/kg to about 1.135 yuan/kg for crop costs during the 10 year period of the simulation (see Figure 8.8). However, the growth in total agricultural output per household still boosts the crop profits per household (see Figure 8.7). The crop profits per household increase from an initial income of about 8,071 yuan to around 8,931 yuan after 30 time units (see Figure 8.9).



Figure 8.7: Agricultural output per household & growth rate of farm products





Figure 8.9: Crop profits per household



By adding tourism activity, the agricultural sector is likely to get some respite at first, despite the coal mining issues influencing farmland quality. Figure 8.10 shows that tourism cannot directly influence the growth rate for farm products and agricultural output per household. The agricultural output is very similar to that in Figure 8.7. The tiny change reflects the growth rate of agricultural outputs in Figure 8.10. The reason is that farmers will receive benefits (i.e., crop profits) from tourists, which slows down the shift of workers from farming to tourism. The labour input could positively

impact on the growth rate of agricultural output (see Eq. (3)). With the growth of tourist arrivals, the crop price increases from about 5.22 yuan/kg to around 5.35 yuan/kg (see Figure 8.11) after 30 time units. Compared to Figure 8.8, the price of agricultural outputs increases by about 0.3 yuan/kg after 10 years (i.e., 30 time units). The crop cost remains the same in Figure 8.8. The crop profits per household increase from about 8,356 yuan to almost 9,541 yuan (see Figure 8.12). Compared to Figure 8.9, the crop profits per household increase by almost 608 yuan after 30 time units.

Figure 8.10: Agricultural output per household & growth rate of farm products (tourist effect)





Figure 8.11 Price and cost of agricultural output (tourist effect)

Figure 8.12: Crop profits per household (tourist effect)



The expected income in rural tourism and the crop profit per household further affect inter-sectoral profit differences between agricultural and tourism. The relevant variables for expected incomes in agriculture and tourism in the simulation model allow us to explore the interactions between the movements of agricultural labour into the tourism sector.

While there is growth in the tourism sector, the opposite is happening in the agricultural sector. Agricultural employment falls from an initial labour allocation of 6,561 farmers to 5,258 farmers (see Figure 8.3), with a general increase in crop profit

per household from about 8,356 yuan (about 2,089 yuan/month) per household to about 9,539 yuan (about 2,384 yuan/month) per household (see Figure 8.12) and expected tourism income grow from 1,095 yuan per month to 2,150 yuan per month (see Figure 8.6).

From the simulation results above, increased tourism can produce more income for tourism operators, which attracts more workers from farming to tourism. In addition, more tourists raise the price of agricultural outputs, which eventually impacts on crop profits per households. Therefore, the application of tourism and social capital to livelihood development is a 'win-win' situation with benefits to both agriculture and tourism.

8.3 The potential impact of increased social capital

This section simulates the effects of different strengths of social capital on alternative livelihoods (rural tourism and farming). It provides the relevant livelihood results showing the effects of differential access to social capital on livelihood activities. In the simulation model, social capital plays an auxiliary role in rural tourism development. Rural labour is assumed to access more tourism business information and service skills through increased social capital resulting in higher tourism incomes.

Figure 8.13 shows that the expected tourism incomes in models with social capital compared to those without social capital gradually increased from a difference of 7 yuan/month (an increase of about 0.1%) in the first time period to a difference of 54 yuan/month (an increase of about 0.05%) in the final simulated period. Consequently, current social capital has a very limited ability to affect the number of tourism workers because in the simulation model, the initial social capital involves only the length of local residence. The social capital of labourers may limit labourers to a narrow social network in which to search for business information relating to rural tourism operation. Figure 8.14 shows that the numbers of rural tourism workers in models with social capital compared to those without social capital gradually increased from

a difference of 15 (increased by about 0.4%) at the first unit time to a difference of 64 (increased by about 1.4%) at the final simulated period.



Figure 8.13: Comparing expected tourism incomes with and without social capital (yuan/month)

Figure 8.14: Comparing number of rural tourism operators with and without social capital



The regression results from Chapter Six suggest that both kin ties and non-kin ties are equally likely to provide access to social capital. There is no statistically significant difference between the relationship between kin ties and social capital and that between non-kin ties and social capital. Thus, the simulation model ignores the specific impact of a change in kin and non-kin ties on social capital. Social capital (SK) is simply increased by one standard deviation (i.e., SK+1) to evaluate the effect of expanded social capital on all households. Given the distribution of predictive scores on social capital (see Chapter Four), the range of increased social capital is still reasonable (i.e., from about 1.059 to about 1.23).

The following comparisons include models with initial social capital (social capital based on length of local residence) and with expanded strength of relationships (SK+1). In this scenario, workers use different strengths of relationships to access relevant business information for rural tourism operation. The increased social capital enable the community to be better informed of the consequences of decisions relating to the tourism industry. In the simulation model, the initial social capital gains one extra unit with these associated factors (see Figure 8.15).





The outcomes show that strengthened social capital causes an approximately 11% increase in expected tourism income by compared to initial social capital in the simulation model (see Figure 8.16). With expanded social capital, the expected

tourism income increases from about 1,091 yuan/month to about 1,207 yuan/month at the first unit time (see Figure 8.16), with 232 extra workers moving from farming to tourism (see Figure 8.17).



Figure 8.16: Change in expected tourism incomes (yuan/month)

Figure 8.17: Change in number of rural tourism workers



The movement of workers from farming to tourism also impacts on tourism service quality, which in turn alters the number of tourist arrivals. In the simulation model, the increase in social capital has little effect on the number of tourist arrivals. The simulation result shown in Figure 8.18 illustrates the growth rate of tourist arrivals as related to strengthened social capital. On average, workers with high social capital can increase the growth rate of tourist arrivals by about 0.7% on average. Figure 8.19 shows the growth in number of tourist arrivals by applying either initial or strengthened social capital. Both social capital levels cause very small effects on the number of tourist arrivals. On average, the strengthened social capital stimulates about a 0.2% increase in tourist numbers.



Figure 8.18: Change in growth rate of tourist arrivals

Figure 8.19: Change in number of tourist arrivals


In previous discussions, tourism was likely to raise agricultural income by influencing the price of agricultural outputs (see Figure 8.11). However, the price of agricultural outputs and crop profits per household from farming seem almost unaffected by scenarios involving different strengths of social capital (see Figures 8.20 & 8.21). The main reason is the small variation in number of tourist arrivals with increased social capital (see Figure 8.19).



Figure 8.20: Change in price of agricultural output

Figure 8.21: Change in crop profits per household labourer in farming



Only workers in tourism show a marked response to the increase in social capital, with an increase in tourism incomes due to better social networks (see Figure 8.16). From Figure 8.22, it can be seen that high social capital causes movement of an extra 4.3% of farmers to tourism. In other words, social resources may significantly help rural labourers to access off-farm livelihood choices (rural tourism). As a result, more rural labourers leave to work in tourism, reducing the growth rate of farm products by about 0.6% on average (see Figure 8.23), which in turn decreases the crop profits. However, the negative effect on crop profits is very weak (less than 0.02%; see Figure 8.21). Compared to the expected increased tourism income (see Figure 8.16) for the local community, these losses in farming income can be ignored.



Figure 8.22: Change in number of farmers

Figure 8.23: Change in growth rate of farm products



8.4 Livelihoods development strategies-current livelihoods and alternative livelihoods

This section looks at three different livelihood strategies (mining, farming, and rural tourism) for rural households. In previous scenarios (Sections 8.2 and 8.3), the labour mobility was based on the different incomes from tourism and farming. Hence, the profitability of the livelihood is assumed to play an important role in determining livelihood choices in this section.

Due to the dominant role of mining activities, the number of jobs and the rate of pay in mining are assumed to be constant in the simulation model. Current income sources for rural households are from mining (constant) and agriculture (without tourism effects). However, tourism activities could be an expanding income source for mining households by using local tourism resources and social networks to develop rural tourism businesses. Thus, the alternative income sources for mining households are either from agriculture (with tourism effects) or tourism alone.

Running the system dynamics model for all three activities as simulations, the income changes can be seen in Figure 8.24, covering a 10-year period. All of the incomes were calculated monthly and social capital was strengthened for all households (i.e., SK+1) to create have a distinct contrast.

From Figure 8.24, it can be seen that mining creates a stable and high income for rural households, which is maintained at the same level throughout the simulation period (i.e., 2,700 yuan per month). However, pollution issues affect the growth rate of farm products and agricultural output (see Figure 8.1 and Figure 8.7). As a non-renewable natural resource, mining activity potentially threats the income sources for rural households in the long term. Without the effects of tourism and social capital on farming, the expected income in farming increased from about 2,017 yuan per month to about 2,233 yuan per month, which is a comparatively sluggish growth rate (see Figure 8.24).

Driven by the development of tourist demand together with social capital, the expected income in farming gradually increased from about 2,089 yuan per month to almost 2,385 yuan per month. The expected incomes in tourism show a remarkable growth, as can be seen in Figure 8.24. Given an increased level of social capital in tourism operations, the monthly income increased from about 1,207 yuan per month to around 2,385 yuan per month after 10 years. Compared to mining, the monthly income gap between mining and farming is reduced from about 611 yuan to about 315 yuan, while the gap between mining and tourism revealed the most powerful reduction, from about 1,493 yuan to 315 yuan. In general, mining activity is potentially replaceable by alternative livelihood choices (tourism and farming) after ten years in this simulation model, particularly as alternative livelihoods develop with strong social capital contributions.



Figure 8.24: Comparing livelihood development options for rural households

If this simulated alternative livelihood provides a good outlook for rural households, could the local community break out of the mining sector and set its sights on rural tourism initially? The analysis suggests that tourism is a worthwhile alternative to mining for producing acceptable incomes in rural households. However, mining is

closely related to local environmental problems, which then influence the development of agriculture and tourism. Thus, the next model looks at the impact of the Zoucheng government deciding to shut down the local pits (i.e., C=0) in order to encourage alternative livelihood development. This simulation can also be used to estimate the situation after all the existing mineral resources have been exhausted. In other words, there are no coal mining effects in this simulation model. This model focuses on the profitability of farming and tourism activities only, in order to determine the prospects for alternative livelihoods for the local community. In the simulation period, this scenario may cause unemployment amongst miners due to mine closures. It is assumed that local government institutes a ten-year subsidy policy and settles all of the laid-off miners in other relevant departments or industries (e.g., transportation industry), to remain consistent with the initial assumption (i.e., constant "miners") and avoid unemployment acting as a potential cause of social unrest.

When comparing farming activity between coal mining impacts (see Figure 8.10) and pit closure (see Figure 8.25), the growth rate of farm products generally increased slightly by around 0.23 kg/unit area on average after pit closure. As a result, agricultural output per household gradually increased by about 3 kg on average after pit closure. Compared to Figures 8.7 and 8.10, the growth rate and total output of farm products increased (see Figure 8.25) due to fewer pollution issues affecting farmland quality.



Figure 8.25: Agricultural output per household & growth rate of farm products (pit closures)

While there is a growth in the agricultural sector (see Figure 8.25), crop prices slightly decreased from about 5.22 yuan/kg to around 5.2195 yuan/kg (see Figure 8.26). In this scenario, the effect of agricultural output per household on crop price has a larger effect than tourist arrivals. However, the crop profits grow from about 8,356 yuan (about 2,089 yuan/month) to about 9,273 yuan (2,318 yuan/month) (see Figure 8.27) due to the increase in agricultural output per household. However, the crop profits seem lower than scenarios including the effects of tourism and coal mining (see Figure 8.12).



Figure 8.26: The price of agricultural output (pit closures)

Figure 8.27: Crop profits per household (pit closures)



For the tourism sector, the effect of pit closures on the growth rate of tourist arrivals is unexpected. The initial expectation was for pit closures to reduce the effects of environmental issues on tourism and farming developments. However, the growth rate of tourist arrivals per unit time barely increased, only from 14 tourists initially to 15 tourists (see Figure 8.28). (It should be noted that the growth line is not smooth because this model uses integers as tourist units). Consequently, the slow growth rate per unit time dragged the increase of the number of tourist arrivals down. Total tourists increased from 46,000 to approximately 46,420 tourists over the simulated time period (see Figure 8.29). The number of tourist arrivals further affects the expected tourism incomes, which rise from just over 1,091 yuan per month to about 1,121 yuan per month after ten years (see Figure 8.30). The small increase in the number of tourist arrivals can also explain the crop price trends in Figure 8.26.



Figure 8.28: *The growth of tourist arrivals per unit time (pit closures)*

Compared to scenarios including the coal mining sector (see Figure 8.18 & 8.19), the growth of the tourism sector (the growth rate of tourist arrivals and the number of tourist arrivals) is significantly slower with pit closures. The main reason is that pit closures mean one of the main income resources is lost and this also means a loss of revenue to the Zoucheng government. Less income for the government will impact on public investment in local infrastructure, which in turn reduces the growth rate of tourist arrivals. In the simulation model, the government investment in tourist growth decreased from about 4.3% to about 0.05% after pit closures (see Figure 8.31), which contributed to the very small rise in the number of tourist arrivals. Consequently, agricultural sector factors such as crop price and crop profit also responded to the tourism sector. In this scenario, financial support for tourism development has a more powerful influence than environmental issues from mining. This simulation also presents the reality of the situation with respect to the dominant role of mining in Zoucheng.

Figure 8.29: Number of tourist arrivals (pit closures)



Figure 8.30: Expected tourism income (pit closures)





Figure 8.31: The effect of government investment on tourist growth

8.5 Summary

Table 8.1 summarises the values of all of the 'outcome' variables for each scenario. Each 'outcome' variable has both an initial value and an end value to show the changes during the simulation period.

The first scenario simulates local environmental issues, presenting two negative relationships (i.e., total coal output vs. farmland quality, and industrial dust discharged vs. the initial tourist number).

The second model is sensitive to the number of workers and incomes in the tourism and agricultural sectors. In the simulation model, tourism presents a brighter outlook. With an increase in the number of tourists, the available livelihood choices (i.e., rural tourism) and income earnings could reduce the vulnerability of household livelihoods (i.e., more income, more diverse livelihoods). This is also seen as contributing towards the enhancement of quality of life in the community. The Zoucheng government might further modify the implementing channel of the developing strategy once tourism becomes large-scale.

For the third model, the increase in social capital can further develop alternative livelihoods. The accumulation of increased social capital seems able to help local

workers to access new careers in tourism businesses. For instance, local workers could access more social resources and networks to obtain tourism information for their rural tourism business operations in the local market. Given a one standard deviation increase in social capital on average for all households (i.e., SK+1), the expected tourism income was increased to a higher level throughout the simulation period compared to using only initial social capital, which also leads to more workers transferring from agriculture to tourism in comparison to the model with initial social capital. However, losing farm labourers does not significantly reduce farming income. For instance, the percentage change for crop profits is similar in scenarios two and three (see Table 8.1).

The final simulation model looks at three livelihood activities (mining, farming, and tourism). In the simulation model, tourism is an alternative livelihood choice for rural households. Regarding the current risks of mining activity (e.g. degrading farmland quality and increasing air pollution), mining households in rural areas can take advantage of the tourist market. Mining households can provide agricultural outputs from their farmlands and gain benefits from tourist demand causing high crop prices and profits. By accessing social networks, workers can more easily access information about rural tourism operations for their own commercial advantage. As a result, the incomes of rural tourism operators increase. The expectation of high incomes in tourism businesses attracts more operators, which can provide more jobs in the local community. This is a "win-win" result for the development of both agriculture and tourism. Thus, there is a good prospect for rural households aiming to develop rural tourism businesses. Therefore, this simulation displays a self-sustaining cycle of growth, which is more sustainable than mining activity.

The final scenario also simulates the impact of local pit closures on the development of the alternative livelihoods. Local government was assumed to incur more burdens (e.g. subsidy policy, resettlement, providing new work for laid-off miners) so as to maintain social stability. As a result, the increase in expected incomes in alternative livelihood strategies (e.g. rural-tourism) in this scenario (pit closures) is slower than in the model with three industries (coal mining, agriculture, and tourism). The simulation results indicate that industrial transformation and the availability of alternative livelihoods need a long time frame to establish. Therefore, a simulation model is necessary in order to analyse the possible policies. Local government can then implement compatible policies before mineral resource exhaustion occurs.

Scenario	Outcomes				
	Variables	1 st unit time	30 th unit time	%	Figures
				Chang	
				e	
	Total coal	39.15 million	39.43 million	0.7%	Figure 8.1
	output	tons	tons		
8.1. Current	Farmland	4.5 mu	4.33 mu	-3.8%	Figure 8.1
8.1: Current	quality				
issues	Industrial dust	4,424 ton	4,459 ton	0.8%	Figure 8.2
issues	discharged				
	The initial	46,000 tourists	43,128 tourists	-6.2%	Figure 8.2
	tourist number				
	Number of	46,000 tourists	89,035 tourists	94%	Figure 8.4
	tourists				
	Expected	1,091	2,150	97%	Figure 8.6
	tourism income	yuan/month	yuan/month		
	Agricultural	2,051	2,265	10%	Figure 8.10
9.2. Detential	output	kg/household	kg/household		
impact of	Growth rate of	7.64 kg/unit	7.16 kg/unit	-6.2%	Figure 8.10
developing tourism sector	farm products	area	area		
	Crop price	5.22 yuan/kg	5.35 yuan/kg	2.34%	Figure 8.11
	Crop cost	1.15 yuan/kg	1.14 yuan/kg	-1%	Figure 8.11
	Crop profits	8,356	9,541	14.2%	Figure 8.12
		yuan/household	yuan/household		
	Number of	6,569	5,258	-20%	Figure 8.3
	labourers in				
	farming				

Table 8.1: Summary of the simulated resu	ults of relevant variables for each scenario
--	--

	Number of	3,430	4,741	38%	Figure 8.3
	rural tourism				
	operators				
	Social capital	SK+1	SK+1	-	Figure 8.15
	Expected	1.208	2,385	97%	Figure 8.16
	tourism income	yuan/month	yuan/month		
	Growth rate of	1,369 tourists	1,413 tourists	3%	Figure 8.18
	tourist arrivals				
	Number of	46,000 tourists	89,337 tourists	94%	Figure 8.19
8.3: Potential	tourist arrivals				
impact of	Crop price	5.2 yuan/kg	5.35 yuan/kg	2.4%	Figure 8.20
increased	Crop profits	8,356 yuan	9,538 yuan	14%	Figure 8.21
social capital	Growth rate of	7.6 kg/unit area	7.12 kg/unit	-6%	Figure 8.23
	farm products		area		
	Number of	6,337	5,000	-21%	Figure 8.22
	farmers				
	Number of	3,662	4,999	37%	Figure 8.17
	rural tourism				
	operators				
	Mining income	2,700	2,700	0%	Figure 8.24
		yuan/month	yuan/month		
	Farming	2,018	2,333	11%	Figure 8.24
	income	yuan/month	yuan/month		
	(Without T and				
	SK)				
	Farming	2,089	2,385	14%	Figure 8.24
8 1.	income (With	yuan/month	yuan/month		
0.4. Livelihood	T and SK)				
development	Tourism	1,207	2,384	98%	Figure 8.24
strategies	income (With	yuan/month	yuan/month		
	SK)				
	Agricultural	2,051	2,273	11%	Figure 8.25
	output (pit	kilo/household	kilo/household		
	closures)				
	Crop price (pit	5.2 yuan/kg	5.2195 yuan/kg	-1%	Figure 8.26
	closures)				
	Crop profits	8,356 yuan	9,373 yuan	12.6%	Figure 8.27
	(pit closures)				

	Growth rate of tourist arrivals (pit closures)	14 tourists	15 tourists	0.7%	Figure 8.28
	Number of tourist arrivals (pit closures)	46,000 tourists	46,420 tourists	0.9%	Figure 8.29
	Expected tourism income (pit closures)	1,091 yuan/month	1,121 yuan/month	3%	Figure 8.30
	Government investment in tourist growth	About 4%	Around 0.05%	-99%	Figure 8.31

The simulation variables are based on recent data and information from Zoucheng city. There are no extra policies to stimulate tourism development in the simulation model. The simulated results for three industries (mining, tourism, and agriculture) in four scenarios did not indicate any exaggeration at variance with the historical facts. The results of the simulations in this thesis provide a possible perspective on the development of alternative livelihoods by using social capital. They also bring some insights for future policy implementation in the area of industry transformation when local mineral resources are exhausted.

Although the dynamic simulation model generated useful results and some possible policy insights for policy-makers, in future there is a need to fill data gaps and use more relevant variables to improve the understanding of several feedback mechanisms within the system. Some other factors with dynamic changes could be found to provide more information about tourist demand, the proportion of breeding livestock in crop-livestock agricultural systems, and other continuous variables relating to social capital and pollution effects so as to increase the integrity and accuracy of the whole simulation model. While the simulation model is currently not fully calibrated to all realistic and specific relationships for all sectors, it can be used to investigate and test livelihood development. The results of the simulation are indicative of trends. Although Zoucheng government has not started to notice the social capital role during the economic development, some of the local policies and cases potentially revealed its important position. Based on Zoucheng's experiences in rural economic development in recent years, local rural policies have moved to support social networks and relationships in villages to strengthen the villagers' abilities to access resources (e.g., financial capital) and information (e.g., new business operations) (Zong, 2016; Zoucheng Government, 2015). For instance, Xiangcheng village used social networks among villagers and a joint liability for a loan to establish a borrower group to reduce the entry barriers for rural people who want to improve their livelihoods, but have a credit constraint. By the end of 2012, 55 local households joined this borrower group and the loan amount was above 4 million yuan. These loans solved the problem cf money shortages in the developing of alternative activities for rural people (Zoucheng, 2012). In Yishan village, the nearby rural tourism households and local farmers worked together to expand their networks to share business information, cooking materials, and even extra guests. This cooperative society in the local village avoids vicious competition and increases the households' revenues (Zoucheng Tourist Administration, 2013).

CHAPTER NINE: SUGGESTED POLICY IMPLICATIONS AND CONCLUDING STATEMENT

9.1 Discussion and Implications

With regard to the role of social capital in sustainable livelihood development, this study discussed two questions: 1. Can different types of social ties (kin and non-kin ties) lead to differential access to social capital in a small community in China? 2. Does the strength of social capital alter local livelihood patterns by diversifying household livelihoods, or providing new livelihoods in industries with the potential for development?

Chapter Six used a regression model to explore the factors associated with social capital. The results argued that both kin ties and weak ties are important factors associated with social capital for mining sector residents in Zoucheng. Chapters Seven and Eight established a simulation model, which applied different intensities of social capital to alternative livelihood development (i.e., rural tourism). The simulated results suggested that increased social capital worked better than lower levels of social capital in the process of transition to an alternative livelihood. By enhancing social capital by one standard deviation (i.e., social capital+1), rural tourism operators received more income (increasing by 97%; see Table 8.1) and attracted more self-employed workers (increased by 37%, see Table 8.1) in an alternative livelihood choice. The analyses and findings of this thesis suggest several important conclusions regarding the role of social capital as a support mechanism for sustainable development strategies that can be drawn upon in the process of developing alternative their livelihoods.

Although the simulation model in Chapter Seven simplified the process of increasing social capital, different types of social ties (kin and non-kin ties) may be important indicators for the implementation of policies on community development. In Confucian and Taoist understanding, everything exists in a state of relatedness and connectedness (Carlisle & Flynn, 2005). The huge population size in China provides the potential to establish complex social networks and social resources. In China, both culture and population size create difficulties for policy-makers wanting to invest in social capital and increase it efficiently. As discussed in Chapter Three, social capital is hard to identify and assess. However, the answer to the first question in this study provides some hints for governments with respect to implementing relevant policies. The regression analysis in Chapter Six tested associated relationships. Both kin ties and non-kin ties are significantly related to access to general social capital. It may be easier for governments to find certain groups of people with different social ties (i.e., kin & non-kin ties) rather than observing the least tangible capital assets (i.e., social capital) in a household, which may be useful for community policy formulation and execution.

The findings of this thesis provide a perspective on the role of social capital in improving livelihood strategies in family systems, communities, or villages. Putnam (1993; 1995) suggests that the level of social capital (e.g., traditional personal networking or family ties) has been declining with the industrialization and modernization of the US since the 1970s. However, China is not likely to follow a similar development path to the US. A traditional East Asian agrarian society is normally focused on family relationships (Pham, 1999). The family is the basis of society and is a historic component of Confucian societies in East Asia (Pham 1999; Dalton, 2001). In China, many villages are named after the largest family living there, emphasising the characteristics of kinship. In these villages, Chinese people customarily use family relationships to communicate with each other before doing any 'business'. Relatives help each other in certain events (e.g. death, birth, or marriage) (Chu, June 2001). Measuring and collecting information about people's kinship may be more efficient and easier than measuring other relationships such as friendships in a village or community. In the process of implementing livelihood development policies, governments could investigate the potential effects on the largest kin-related households in villages so as to increase the effectiveness of the implementing the policy. These target households might have strong social capital and social networks, considering their kinships and reputations in a village, and could play leading roles to guide the others and increase the overall development in a community. However, the special "guanxi elite" may influence other residents' involvement and allow only a few or limited residents into the "inner circle" of tourism development (Li et al., 2007). Thus, the influence of weak ties is also considered in the process of tourism development. Weak ties may diversify available information and strengthen social capital (Granovetter, 1973, 1974; Bian & Ang, 1997), which could be applied to extend residents' involvement after the early stages of tourism development.

Investment in social capital plays an important role in the process of tourism development. As discussed in Section 3.4.2, social capital can influence residents' collective actions and attitudes to tourism development. Through community members' involvement, a higher satisfaction with tourism development and a higher tolerance for tourists can be achieved (Timothy & Tosun, 2003). For instance, local knowledge and experience in understanding tourism development can be used to empower local residents to participate in tourism effectively (Tosun, 2000), or reduce potential environmental or traditional cultural conflicts between tourists and local aspirations (Aas et al., 2005; Timothy & Tosum, 2003; Stronza, 2000). In the Chinese context, increasing attention has been given to guanxi in tourism development. Based on a comprehensive review of guanxi and relationship marketing, guanxi can influence resident involvement and bring core stakeholders together in tourism development (Li et al., 2007; Lew & Wang, 2004; Geddie et al., 2002). Social capital support (e.g. government policies involving community participation) may provide a more targeted approach to local tourism development in rural communities than "blind" spending by governments on tourism promotion.

Due to the dominant role of mining in household livelihoods in Zoucheng, the collected social capital information from miners was assumed to represent the household level in the simulation model (see Section 7.4.4). Household members with high-ranking positions in mining employment could help other household members to move into alternative livelihood choices or lower-ranking jobs in the mining industry by using their current social resources. The present situation in Zoucheng involves the risks of natural resource dependency. The residents of Zoucheng are over-reliant on mining activity, but the available job opportunities in mining are limited as represented by the constant number of miners in the simulation model (see Chapters Seven and Eight), and cannot satisfy the total job demands from rural households (as discussed in Chapter Two). Regardless of the environmental considerations, the limited job opportunities in mining are another important reason to develop alternative livelihoods for rural households in Zoucheng. One implication for policy decision-making is the relative advantage existing in different job positions. In terms of social ties, people in high-ranking positions are likely to access better resources (discussed in Chapter Three). According to Table 4.2, the Prestige score for a worker/miner is 45 and for a farmer it is only 30 (See Chapter Four). In mining, miners' job positions tend to be reasonably homogeneous in status. In rural households, a household member in farming may take advantage of the status and pay improvements of a mining job. Looking at the simulated results in Chapter Eight, miners' social resources could help other household members to have more alternative livelihood choices (rural tourism).

The superior rates of pay in mining almost preclude the possibility of shifting from mining to tourism in Zoucheng. However, developing the competitive advantage of rural tourism may attract other family members in mining households to transition from farming to tourism, which has the potential to increase the overall benefits in rural mining households. It might provide a good start to developing a more sustainable livelihood, particularly in this case, where there are more than two existing resources (farmland, mineral resource, and tourist resource) in Zoucheng and the local economy is presently relying heavily on only one natural resource (mineral resource).

By considering social capital, the focus of economic development could change from purely business or monetary-oriented to the context of community interest such as neighbourhoods and service programmes for disadvantaged groups (Lejano & Wessells 2006; Blair & Carroll 2008). In China, the government has implemented many programmes with goals that include the reduction of poverty, but which focus on a narrow range of methods such as allocating food or other daily necessities. However, because social capital assumes such an important role for some households, future policies might consider ways to enhance social capital and thus alleviate poverty, using the analysis provided in Chapter Six, which illustrates clearly the factors that have the greatest relationships with residents' social connections and also discusses their motivational characteristics. The results from the regression model in Chapter Six indicated, that length of residence (i.e., years resident in Zoucheng) and higher educational levels had significant positive relationships with social capital. Therefore, government could invest in education (e.g. hospitality, tourism mangement, or other subjects related to rural economic development). Also, more government welfare policies would benefit local people and encourage longer residence in Zoucheng. Both of these government investments might assist pertain tourism communities by improving local experience and promoting educated knowledge about rural tourism development, which could increase the understanding of tourism businesses and local residents' involvement in effective tourism development processes.

The simulation results in Chapter Eight showed that increased of social capital can further develop alternative livelihoods (i.e., rural tourism). This has implications for policy makers who want to develop tourism in rural areas. The concept of social capital could be applied in establishing and operating rural tourism businesses, particularly in developing areas (Zhao et al., 2011). As the results from the regression model in Chapter Six show, kin ties are important factors associated with social capital. Members of strong social networks, such as family members, can be encouraged to enter tourism employment, decision-making, sharing rural tourism benefits, or local cultural concerns, which can further increase the involvement of social capital in tourism development. (Ying & Zhou, 2007). For example, it is common to hire family members for the daily operation of small tourism business in China. In the long run, all profits from small tourism businesses are more relevant to the business owner's family as household income rather than any form of individual salary (Zhao, 2009). All of the involved household members could be benefited in operating tourism businesses, which in turn pushes forward tourism development.

The model in Chapter Eight also simulated the scenario of conducting pit closures before making any attempt at establishing a well-developed alternative industry in Zoucheng. This idea proved to be an intellectual dead end, but it does demonstrate some of the risks of economic transformation in the absence of well-developed alternative livelihoods. It is always very difficult to find an equilibrium solution when a developing economy meets issues between environmental costs and monetary costs. Short-term behaviours and blind pursuit of short-term gains are dangers for economic development. The results of the simulation models in Chapter Eight indicate that it may take some time to establish an alternative livelihood choice and replace the current system. It is particularly important for the future viability of Zoucheng livelihoods that the local economy shifts towards a tourist economy, away from the present mineral resource-dependent base in a rural area.

9.2 Conclusion

The main objective of this thesis was to determine whether the strength of social capital can alter household livelihoods and their sustainability. The evidence supports the main research question (Research Question (2)). The empirical results from the simulation model indicate that social capital can generally enhance the effects of

policy implementations in certain cultural contexts. This finding accords with the conclusion of Lin & Si (2010).

This thesis used the Position Generator instrument to measure the social capital of workers involved in the mining industry and residing in rural and urban neighbourhoods of Zoucheng city. Lin's (2001; 2004) theory of social capital in the Chinese context was used for this analysis. The analysis then took the generated social capital scores as the dependent variable, and demonstrated that kin and non-kin ties are equally effective ways to access social capital in a small community. Based on regression analysis results from this study and collected background information available in Zoucheng, simulation models were used to determine the role of social capital in alternative livelihood development. By implementing relevant policies on social capital and rural tourism, the results of these simulation models supported the possibility of replacing the current development strategy (mining). This study also provided a new perspective on the nature of Chinese social capital and its societal consequences during a transition economy in a small city.

The thesis fills a gap in the social capital literature by measuring social networks and the role of kin and non-kin ties in terms of their relationship to social capital in a county-level city, which has been largely ignored to date.

This research can facilitate the application of social capital to promote the implementation of appropriate regulations and policies at the local government level. For instance, in small and medium-sized cities, to solve economic restructuring issues, local policymakers could promote sustainable development of people's livelihoods, and diversify living strategies for communities in rural areas. This would involve selecting the most suitable of the available resources, which can enhance the efficiency of the policies. During the implementation process, access to social capital can increase the likelihood of getting access to more resources (information), which will further expand the variety of livelihood strategies available and reduce vulnerability. The findings have implications for government policy, the personal

career development of individuals in transitional economies, and other researchers analyzing similar areas such as social capital applications with sustainable livelihood frameworks in other countries influenced by eastern Asian culture. This study also provides a foundation for developing social networks to accrue the necessary social capital for the success of other livelihood strategies in rural areas of China.

9.3 Limitations

The analysis and models in this thesis have several limitations. First, although the causality was discussed in Chapter Three, it was not established in the regression model for this study. The interpretation of the regression results for this study on social capital was limited to associations rather than causation. For instance, while human capital may increase social capital, the converse may also be true. With different research purposes, follow-up data in future surveys will provide a longitudinal dataset more conducive to making causal inferences. Second, the measurements were also susceptible to omitted variable bias. For instance, selfreported measures of social networks in current jobs involve subjective assessment by participants. Third, the simulation model assumed no transition cost for workers shifting between different industries. In practice, governments often use preferential policies such as subsidies for specific activities so as to promote economic restructuring. Considering these transition costs in the model could produce different results. Fourth, social capital was directly increased by one standard deviation in simulation model. However, this did not indicate the driving values for this growth. Although the regression model in Chapter Six found that educational level and the strength of social ties (i.e., kin and non-kin ties) are positively associated with access to social capital, the specific "cost" of increasing social capital by one unit is omitted in the simulation model. In addition, the simulation model ignored new entrants to the local labour market such as migrants and workers graduating from school. Finally, some coefficients of the relationships and interlinkages such as the relationship between number of rural tourism operators and tourism service quality in the simulation model are based on relevant theories and similar study areas, which lack empirical data. This gap may be filled in a future study.

9.4 Possibilities for future research

As far as the research participants in this study were concerned, the mining group tended to be younger than the population average and were more likely to be male. In future, participants could be selected from broader areas, which may be more generalizable to all rural households in China.

There are higher growth and upside potentials in the simulation models. The risks of a natural resource-dependent economy in the simulation model provide the possibility of analysing the environmental and monetary costs involved in the development of multiple industries, as for example in the scenario of pit closure in Chapter Eight. In practice, agricultural outputs could be diversified in the agricultural model. The coefficients for crop price and cost could be calculated from several specific surveys and experiments. More potential factors associated with access to social capital could be used in the simulation model, which needs further work in terms of social capital analysis. The driving values for the growth of social capital corresponding to different livelihood outcomes could be further analysed. As local rural tourism develops in the future, more practical relationships such as local pollution effects on numbers of tourist arrivals, number of tourist arrivals and incomes in rural tourism could be studied in order to refine the simulation model.

The analyses conducted in this study provide some good supplements to existing social capital and sustainable livelihood development literature, particularly in a relation to small cities in China. The study provides a base for future investigations of the role of social capital during the transition to a sustainable livelihood in the Chinese context.

REFERENCES

- Aas, C., Ladkin, A., & Fletcher, J. (2005). Stakeholder Collaboration and Heritage Management. Annals of Tourism Research, 32(1), 28-48.
- Abdulai, A. & CroleRees, A. (2001). Determinants of income diversification amongst rural households in Sourthern Mali. *Food Policy*, 26(4), 437-452.

Abramson, N. R., & Ai, J. X. (1999). Canadian Companies Doing Business in China: Key Success Factors. *Management International Review*, 39(1), 17-27.

- Acquaah, M. (2007). Managerial social capital, strategic orientation, and organizational performance in an emerging economy. *Strategic Management Journal*, 28, 1235-1255.
- Adam, F., & Roncevic, B. (2003). Social capital: Recent debates and research trends. *Social Science Information*, *42*(2), 155-183.
- Adhikari, K.P., & Goldey, P. (2009). Social capital and its "Downside": The impact on sustainability of induced community-based organizations in Nepal. *World Development*, *38*(2), 184-194.
- Adler, P. S., & Kwon, S. W. (2002). Social capital: Prospects for a new concept. *Academy of Management Review*, 27(1), 17-40.
- Aguilera, M. B., & Massey, D. S. (2003). Social capital and the wages of Mexican Migrants: new hypothesis and tests. *Social Forces*, 82(2), 671-701.
- Ahmed, N., Troell, M., Allison, E. H., & Muir, J. F. (2010). Prawn postlarvae fishing in coastal Bangladesh: Challenges for sustainable livelihoods. *Marine Policy*, 34(2), 218-227.
- Algina, J., & Olejinik, S. (2003). Sample size tables for correlation analysis with applications in partial correlation and multiple regression analysis. *Multivariate Behavioral Research*, *38*(3), 309-323.
- Allison, E. H., & Ellis, F. (2001). The livelihood approach and management of small-scale fisheries. *Marine Policy*, 25(5), 377-388.
- Allison, E. H., & Horemans, B. (2006). Putting the principles of the Sustainable Livelihoods Approach into fisheries development policy and practice. *Marine Policy*, 30(6), 757-766.
- Ambler, T. (1994). Marketing's Third Paradigm: Guanxi, *Business Strategy Review*, 5(4), 69-80.
- Amijaya, J., & Littke, R. (2005). Microfacies and depositional environment of Tertiary Tanjun Enim low rank coal, South Sumatra Basin, Indonesia. *International Journal of Coal Geology*, 61(3-4), 197-221.
- Amuedo-Dorantes, C., & Mundra, K. (2007). Social networks and their impact on the earnings of Mexican immigrants. *Demography*, 44(4), 849-863.
- Anderson, C. L., Locker, L., & Nugent, R. (2002). Microcredit, Social Capital, and Common Pool Resources. *World Development*, *30*(1), 95-105.
- Annen, K. (2001). Inclusive and exclusive social capital in the small-firm sector in developing countries. *Journal of Institutional and Theoretical Economics*, 157(2), 319-330.

- Archer, B. H. & Fletcher, J. E. (1991). The Development and Application of Multiplier Analysis. In C.P. Cooper (Ed.), *Progress in Tourism, Recreation and Hospitality Management*, Vol. 1, (pp.28-46). London. UK: Belhaven.
- Archer, B. H. (1977). *Tourism Multiplier: The State of the Art*, Occasional Papers in Economies, No 11, Bangor, Wales: University of Wales Press.
- Asafu-Adjaye, J. (2005). *Environmental Economics for Non-Economists*. Singapore: World Scientific Publishing.
- Ateljevic, J., & Doorne, S. (2004). Diseconomies of scale: A study of development constraints in small tourism firms in Central New Zealand. *Tourism and Hospitality Research*, 5(1), 5-24.
- Atkinson, R., & Flint, J. (2001). Accessing hidden and hard-to-reach population: snowball research strategies. *Social Research Update*, *Issue 33*. Retrieved from http://sru.soc.surrey.ac.uk/SRU33.html
- Australian Bureau of Statistics. (2009). Australian National Accounts: Tourism Satellite Account 2008-2009, 5249.0 Canberra ACT: ABS.
- Babulo, B., Muys, B., Nega, F., Tollens, E., Nyssen, J., Deckers, J., & Mathijs, E. (2008). Household livelihood strategies and forest dependence in the highlands of Tigray, Northern Ethiopia. *Agricultural Systems*, 98(2), 147-155.
- Bagozzi, R. P., & Yi, Y. (1988). On the evaluation of structural equation model. *Journal of Academy of Marketing Science*, *16*(1), 74-94.
- Ballesteros, E. R., & Ramirez, M. H. (2007). Identity and community-Reflections on the development of mining heritage tourism in Southern Spain. *Tourism Management*, 28(3), 677-687.
- Bannock, G., Baxter, R. E., & Davis, E. (2003). *The Penguin Dictionary of Economics* (7th edition). London, UK: Penguin Group.
- Barrett, C. B., Clark, M. B., Clay, D. C., & Reardon, T. (2005). Heterogeneous constraints, incentives and income diversification strategies in rural Africa. *Quarterly Journal of International Agriculture*, 44(1), 37-60.
- Barrett, C. B., Reardon, T., & Webb, P. (2001). Nonfarm income diversification and household livelihood strategies in rural Africa: concepts, dynamics, and policy implications. *Food Policy* 26(4), 315-331.
- Barrett, C., Bezuneh, M., & Abdillahi, A. (2001). Income diversification, poverty traps and policy shocks in Cote d'Ivoire and Kenya. *Food Policy*, *26*(4), 367-384.
- Barrett, C., Reardon, T., & Webb, P. (2001). Nonfarm income diversification and household livelihood strategies in rural Africa: Concepts, issues, and policy implications. *Food Policy*, 26(4), 315-331.
- Bateman, I. J., Carson, R. T., Day, B., Hanemann, M., Hanleys, N., Jones-Lee, M., Swanson, J. (2002). *Economic Valuation with Stated Preference Techniques*. A *Manual*. Cheltenham, UK: Edward Elgar.
- Batjaral, B. (2007). Internet entrepreneurship: Social capital, human capital, and performance of Internet ventures in China. *Research Policy*, *36* (5), 605-618.
- Batjargal, B. (2006). The dynamics of entrepreneurs' networks in a transition economy: the case of Russia. *Entrepreneurship and Regional Development*, *18*(4), 305-320.

Batjargal, B. (2007). Internet entrepreneurship: Social capital, human capital, and performance of Internet ventures in China. *Research Policy*, *36*(5), 605-618.

Baum, T. (1995). *Managing Human Resources in the European Tourism and Hospitality Industry: A Strategic Approach*. London, UK: Chapman & Hall.

Bebbington, A. J. (1999). Capitals and capabilities: A framework for analyzing peasant viability, rural livelihoods and poverty. *World Development*, *27*(12), 2021-2044.

Becton, S., & Graetz, B. (2001). Small business-small minded? Training attitudes and needs of the tourism and hospitality industry. *International Journal of Tourism Research*, 3(2), 105-113.

Belisle, F. J. (1983). Tourism and Food Production in the Caribbean. *Annuals of Tourism Research*, 10(4), 497-513.

Berry, S. (1989). Social institutions and access to resources. Africa, 59(1), 41-55.

Berry, T. (2006). The Predictive Potential of the TALC Model. In R. Butler (Ed.)

The tourism area life cycle Volume 1 Applications and modifications (pp. 254-280). Clevedon, UK: Channel View Publications. Retrieved from

http://books.google.co.uk/books?id=AicuoXTVrU0C&pg=PP1&dq=The+Tourism+

Area+Life+Cycle+Volume+2+Conceptual+and+Theoretical+Issues&hl=en&ei=ZO

NvTI-YNtjU4wbm4PTeCA&sa=X&oi=book_result

&ct = result & resnum = 1 & ved = 0 CC4Q6AEwAA # v = onepage & q & f = false

- Bi, Z. W., & Ding, D. X. (2003). Surface damage caused underground mining and prevention. *Safe Environment Engineering*, 10, 54-57.
- Bian, Y. (1994) *Work and Inequality in Urban China*. Albany, NY: State University of New York Press.

Bian, Y. (1997). Bringing Strong Ties Back In: Indirect Connection, Bridges, and Job Search in China. *American Sociological Review*, 62(3), 36-385.

Bian, Y. (2005/2006). Guanxi. In J. Beckert & M. Zafirovski (Eds.), International Encyclopedia of Economic Sociology (pp. 312-314). New York, NY: Routledge.

Bian, Y. (2008). The formation of social capital among Chinese urbanites: theoretical explanation and empirical evidence. In N. Lin, & B. Erickson (Eds.), *Social Capital: An International Research Program* (pp. 81-104). New York, NY: Oxford University Press.

Bian, Y., & Ang, S. (1997). Guanxi Networks and Job Mobility in China and Singapore. *Social Forces*, *75*, 981-1006.

Bian, Y., & Huang, X. (2009). Network resources and job mobility in China's transitional economy. *Research in the Sociology of Work, 19*, 255-282.

Bian, Y., & Li, Y. (2001). Social Network Capital of Chinese Urban Families. *Tsinghua Sociological Review*, 2, 1-18.

Bian, Y., Shu, X., & Logan, J. (2001). Communist Party membership and regime dynamics in China, *Social Forces*, *79*(3), 805-841.

Bian, Z., Inyang, H. I., Daniels, J. L., Otto, F., & Struthers, S. (2010). Environmental issues from coal mining and their solutions. *Mining Science and Technology* 20(2), 215-223.

- Biernacki, P., & Dan, W. (1981). Snowball Sampling: Problem and Techniques of Chain Referral Sampling. *Sociological Methods and Research*, *10*(2), 141-163.
- Blair, J. P., & Carroll, M. C. (2008). Social Capital. *Economic Development Journal*, 7(3), 42-49.
- Bontkes, T. S., & Keulen, H. (2003). Modelling the dynamics of agricultural development at farm and regional level. *Agricultural Systems*, *76* (1), 379-396.
- Bourdieu, P. (1983). The Forms of Capital. In J. G. Richardson (Ed.), *Handbook of Theory and Research for the Sociology of Education* (pp. 241-258).Westport, CT: Greenwood Press.
- Bourdieu, P., & Wacquant, L. J. D. (1992). *An invitation to reflexive sociology*. Chicago, IL: University of Chicago Press.
- Boxman, E. A. W., de Graaf, P. M., & Flap, H. D. (1991). The impact of social and human capital on the income attainment of Dutch managers. *Social Networks*, *13*(1), 51-73.
- Bramwell, B., & Lane, B. (1994). Rural tourism and sustainable rural development. Proceedings from the Second International School of Rural Development. London, UK: Channel View Books.
- Bretschger, L. & Smulders, S. (2011). Technologies, preferences, and policies for a sustainable use of natural resources. [Editorial]. *Resource and Energy Economics*, *33*, 881-892.
- Brown, R., Stephens, C., Ouma, J., Murithi, M., & Barrett, C. (2006). Livelihood strategies in the rural Kenyan highlands. *African Journal of Agricultural and Resource Economics*, *1*(1), 21-36.
- Brown, W., & Churchill, C. (1999). Providing insurance to low-income households Part 1: A primer on insurance principles and products. Bethesda, MD: Microenterprise Best Practices.
- Bruntland Commission Report. (1987). *Report of the World Commission on Environment and Development: Our common future*, United Nations WCED: Oxford University Press. Retrieved from http://www.un-documents.net/ourcommon-future.pdf
- Bryden, J. M. (1973). *Tourism and Development: A Case Study of the Commonwealth Caribbean*. Cambridge, UK: Cambridge University Press.
- Buhalis, D. (2000). Marketing the Competitive Destination of the Future. *Tourism Management*, 21(1), 97-116.
- Bureau of City Planning of Zoucheng City. (2014). *The master plans in Zoucheng*. Zoucheng: Zoucheng government information. Retrieved from http://ghj.zcw.gov.cn/index.php/cms/item-view-id-269.shtml
- Bureau of Statistics of Zoucheng City. (2014). Statistical communiqué of social and economic development in Zoucheng: 2013. Zoucheng: The Bureau of Statistics of Zoucheng City.
- Burt, R. S. (1984). Network items and the general social survey. *Social Networks*, *6*(4), 293-339.
- Burt, R. S. (1992). *Structural holes: The social structure of competition*. Cambridge, MA: Harvard University Press.

- Burt, R. S. (1997). The Contingent Value of Social Capital. *Administrative Science Quarterly*, 42(2), 339-365.
- Burt, R. S. (2001). Structural holes versus network closure as social capital. In N.Lin, K. S. Cook & R. S. Burt (Eds.), *Social capital: Theory and research* (pp. 31-56). Hawthorne, NY: Aldine de Gruyter.
- Bury, J. (2004). Livelihoods in transition: transnational gold mining operations and local change in Cajamarca, Peru. *Geographical Journal*, *170*(1), 78-91.
- Butler, R. (1980). The Concept of a Tourist Area Cycle of Evolution: Implications for Management of Resources. *Canadian Geographer*, 24(1), 5-12.
- Butler, R. (2006). *The Tourism Area Life Cycle Model, Vol 1, Applications and Modifications*. Clevedon, UK: Channel View Publications.
- Cai, F., Park, A., & Zhao, Y. (2008). The Chinese labour market in the reform era. In L. Brandt & T. Rawski (Eds.), *China's Great Economic Transformation*. Cambridge, UK: Cambridge University Press.
- Cai, Q. (2003). Migrant Remittances and Family Ties: A case Study in China. *International Journal of Population Geography*, 9(6), 471-483.
- Cai, Y. (2003). Collective Ownership or Cadres' Ownership? The Non-Agricultural Use of Farmland in China. *China Quarterly*, *175*, 662-680.
- Carlisle, E., & Flynn, D. (2005). Small business survival in China: Guanxi, Legitimacy, and Social capital. *Journal of Development Entrepreneurship*, 10(1), 79-96.
- Carney, D. (ed.) (1998). Sustainable Rural Livelihoods What contribution can we make? London, UK: Department for International Development.
- Carroll, M. C., & Stanfield, J. R. (2003). Social capital, Karl Polanyi and American Social and Institutional Economics. *Journal of Economic Issues*, *37*(2), 387-395.
- Carter, C. A., Chen, J., & Chu, B. (2002). Agricultural productivity growth in China: farm level versus aggregate measurement. *China Economic Review*, *14*(1), 53-71.
- Casagrandi, R., & Rinaldi, S. (2002). A theoretical approach to tourism sustainability. *Conservation Ecology*, *6*(1), 13-27.
- Castelan-Ortega, O.A., Fawcett, R.H., Arriaga-Jordan, C., Herrero, M. (2003). A decision support system for smallholder campesino maize-cattle production systems of the Toluca Valley in Central Mexico. Part 1-integrating biological and socio-economic models into a holistic system. *Agricultural Systems*, 75(1), 1-21.
- Castillo Canalejo, A. M., Lopez-Guzman Lopez, T. J., & Millan Vasquez de la Torre, G. (2010): Industrial Mining Tourism Activities, *Anatolia: An International Journal of Tourism and Hospitality Research*, 21(2), 379-383.
- Cawley, M., & Gillmor, D.A. (2008). Integrated rural tourism: concepts and practice. *Annals of Tourism Research*, *35*(2), 316-337.
- Chambers, R., & Conway, G. (1992). Sustainable Rural livelihoods: Practical Concepts for the 21st Century. Institute for Development Studies, DP 296. Brighton, UK: University of Sussex.
- Chan, K. W., & Hu, Y. (2003). Urbanization in China in the 1990s: New definition, different series, and revised trends. *The China Review*, *3*(2): 49-71.

- Chan, K. W., Henderson, J. V., & Tsui, K. Y. (2008). Spatial dimensions of Chinese economic development. In L. Brandt & T. Rawski (Eds.), *China's Great Economic Transformation*, New York, NY: Cambridge University Press.
- Chaves, R., Lopez, D., Macias, F., Casares, J., & Monterroso, C. (2013). Application of System Dynamics technique to simulate the fate of persistent organic pollutants in soils. *Chemosphere*, 90(9), 2428-2434.
- Chen, C., Liu, W., Liaw, S., & Yu, C. (2005). Development of a dynamic strategy planning theory and system for sustainable river basin land use management. *Science of the Total Environment*, *346*, 17-37.
- Chen, F. (2009). Union Power in China: Source, Operation, and Constraints. *Modern China 35*(6), 662-689.
- Chen, J. (2007). Rapid urbanization in China: a real challenge to soil protection and food security. *Catena*, 69(1), 1-15.
- Chen, M. (1995). Asian Management Systems. New York, NY: Routledge.
- Chen, Y. (2012). The causal effect of social capital in the labour market identification challenges and strategies. *Chinese Sociological Review*, 44(2), 76-100.
- Chen, Z., Huffman, W. E., & Rozelle, S. (2009). Farm technology and technical efficiency: Evidence from four regions in China. *China Economic Review*, 20(2), 153-161.
- Cheng, E.J. (1998). *Household heads, non-economic factors and grain production in China in the 1990s.* Working Paper #98/5, Center for Asian Studies, Chinese Economy Research Unit. Adelaide, Australia: University of South Australia.
- Cheng, M.Y., & Luo, J.D. (2001). The downside of organizational social capital in group knowledge creation: A study of Chinese high-tech organizations. *African Journal of Business Management*, 5(30), 12114-12125.
- China Daily. (2006, May 29). *Rural Labour Shortage Beginning to Be Felt*. Retrieved from http://www.chinadaily.com.cn/bizchina/2006-05/29/content_602449.htm
- China Daily. (2007, May 16). *Do we need our own Mother's Day?* Retrieved from http://www.chinadaily.com.cn/ezine/2007-05/16/content_873529.htm
- Choi, H. C., & Murray, I. (2010). Resident attitudes toward sustainable community tourism. *Journal of Sustainable Tourism*, 18(4), 575-594.
- Chow, G. C. (2002). China's Population Problems and Policy. Unpublished document. Department of Economics, Princeton University. Princeton, NJ.
- Chow, M., & Thompson, S.K. (2003). Estimation with link-tracing sampling designs-a Bayesian approach. *Survey Methodology*, 29(2), 197-205.
- Christakis, N. & Fowler, J. (2010). *Connected: the amazing power of social networks and how they shape our lives*. London, UK: HarperPress.
- Chu, J. (2001). Prenatal Sex Determination and Sex-Selective Abortion in Rural Central China. *Population And Development Review*, 27(2), 259-281.
- Chwe, M. S. Y. (1999). Structure and strategy in collective action. *American Journal of Sociology*, 105(1), 128-156.

- Cockcroft, A., Barry, G., Cotes, J. E., & Lyons, J. P. (1982). Shape of small opacities and lung function in coal workers. *Thorax 37*, 365-369.
- Cole, S. (2009). A Logistic Tourism Model Resort Cycles, Globalization, and Chaos. *Annals of Tourism Research*, *36*(4), 689-714.
- Coleman, J. S. (1988). Social capital in the creation of human capital. *American Journal of Sociology*, *94*, S95-S120.
- Coleman, J. S. (1990). *Foundations of Social Theory*. Cambridge, MA: Harvard University Press.
- Combs, J.G. (2010). Big samples and small effects: Lets not trade relevance and rigor for power. *Acedemy of Management Journal*, *53*, 9-13.
- Conceicao, P., Ricardo, F., Sebastian, L. (2011). Managing Natural Resources for Human Development in low-income Countries. UNDP Regional Bureau for Africa Policy Notes 2011-002.
- Congressional Budget Office (1990). *Federal taxation of tobacco, alcoholic beverages and motor fuels*, Washington, DC: US Government Printing Office.

Conrad, J. M. (1995). Bioeconomic models of the fishery. In D. Bromley (Ed.), Handbook of Environmental Economics (pp. 405-432). Oxford, UK: Blackwell.

- Cook, K. S., Emerson, R. M., & Gillmore, M. R. (1983). The distribution of power in exchange networks: Theory and experimental results. *American Journal of Sociology*, 89(2), 275-305.
- Coomes, O. T., Barham, B. L., Takasaki, Y. (2004). Targeting conservationdevelopment initiatives in tropical forests: insights from analyses of rain forest use and economic reliance among Amazonian peasants. *Agricultural Economics*, 51, 47-64.
- Cordes, S., Allen, J., Bishop, R.C., Lynne, G.D., Robison, L.J., Ryan, V.D., & Shaffner, R. (2003). Social capital, attachment value, and rural development: A conceptual framework and application of contingent valuation. *American Journal* of Agricultural Economics, 85(5), 1201-1207.
- Costanza, R., & Ruth, M. (1998). Using dynamic modeling to scope environmental problems and build consensus. *Environmental Management*, 22(2), 183-195.
- Costanza, R., & Voinov, A. (2001). Modeling ecological and economic systems with STELLA: Part III. *Ecological Modeling*, *143*(1-2), 1-7.
- Cox, L. J., Fox, M., & Bowen, R. L. (1995). Does Tourism Destroy Agriculture? *Annals of Tourism Research*, 22(1), 210-213.
- Crookes, D. J., Blignaut, J. N., de Wit, M. P., Esler, K. J., Le Maitre, D. C., Milton, S. J., ... Rebelo, A. (2013). System dynamic modeling to assess economic viability and risk trade-offs for ecological restoration in South Africa. *Journal of Environmental Management*, 120, 138-147.
- Dalkmann, H., Hutfilter, S., Vogelpohl, K., & Schnabel, P. (2008). Sustainable mobility in rural China. *Journal of Environmental Management*, 87, 249-261.
- Dalton, R. J., & Ong, N. N. T. (2005). *Civil society and social capital in Vietnam. Modernization and Social Change in Vietnam.* Hamburg, Institut for Asienkunde.
- Dalton, R. J., Hac, P. M., Nghi, P. T., & Ong, N. T. (2001). Social Relations and Social Capital in Vietnam: The 2001 World Value Survey. Stockholm, Sweden:

WVS. Retrieved from

http://www.democracy.uci.edu/files/docs/vietnam/vietnam02.pdf

- Dasgupta, P. (2000). Economic progress and the idea of social capital. In P.Dasgupta & I. Serageldin (Eds.), *Social capital: A Multifaceted Perspective*.Washington, DC: World Bank.
- Dasgupta, P. (2005). Economics of social capital. *The Economic Record*, 81(225), S2-S21.
- Davis, B., Winters, P. Carletto, G. Covarrubias, K. Quinones, E., Zezza, A., DiGiuseppe, S. (2007). *Rural Income Generating Activities: A Cross Country Comparison.* ESA Working Paper No 07-16. Rome, Italy: FAO.
- Davis, J. & Whittington, D. (1997). A comparison of participatory group meetings and contingent valuation surveys: a case study of Lugazi, Uganda. *Economic Development and Cultural Change*, 46(1), 73-94.
- De Brauw, A., Huang, J., Rozelle, S., Zhang, L., & Zhang, Y. (2002). The Evolution of China's Rural Labor Markets during the Reforms. *Journal of Comparative Economics*, 30(2), 329-353.
- De Weerdt, J. (2002). *Risk-sharing and endogenous network formation*. WIDER Discussion Paper 57/2002. Helsinki, Finland: World Institute for Development Economics Research (WIDER).

Deardorff, A. V. (2004). *Local Comparative Advantage: Trade Costs and the Pattern of Trade*. DP No. 500. Ann Arbor, MI: University of Michigan.

- Demoi, L. A. (1991). About rural and farm tourism. *Tourism Recreation Research*, *16*(1), 3-6.
- Demurger, S., Fournier, M., & Yang, W. (2010). Rural households' decisions towards income diversification: Evidence from a township in northern China, *China Economic Review*, 21, 532-544.
- Deng, X., Huang, J. Rozelle, S. & Uchida, E. (2006). Cultivated land conversion and potential agricultural productivity in China. *Land Use Policy* 23(4), 327-384.
- Deng, X., Luo, Y., Dong, S., & Yang, X. (2005). Impact of resources and technology on farm production in northwestern China. *Agricultural Systems*, 84(2), 155-169.
- Department for International Development (DFID) (1999). Sustainable livelihoods guidance sheets. London, UK: DFID.
- Department of Culture, Radio, Film, TV, Press and Publication of Zoucheng (DPZ). (2015). *The research report of cultural development in Zoucheng during the 13rd five-year plan.* Zoucheng: DPZ. Retrieved from http://whjgk.zoucheng.gov.cn/index.php/cms/item-view-id-441.shtml
- Dercon, S. (2002). *Income risk, coping strategies and safety nets*. WIDER Discussion Paper 22/2002. Helsinki, Finland: United Nations University, World Institute for Development Economics Research (WIDER).
- Ding, H., Wang, R., Wu, J., Zhou, B., Shi, Z., & Ding, L. (2007). Quantifying land use change in Zhejiang coastal region, China using muti-temporal landsat TM/ETM+ Images. *Pedosphere*, 17(6), 712-720.

- Domar, E. (1946). Capital expansion, rate of growth and employment. *Econometrica*, *14*(2), 137–147.
- Dowla, A. (2006). In credit we trust: Building social capital by Grameen Bank in Bangladesh. *The Journal of Socio-Economics*, *35*, 102-122.
- Du Cros, H., Bauer, T. Lo, C., & Rui, S. (2005) Cultural Heritage Assets in China as Sustainable Tourism Products: Case Studies of the Hutongs and the Huanghua Section of the Great Wall. *Journal of Sustainable Tourism*, *13*(2), 171-194.
- Du, B. S. (2010). The development of ecotourism in Zoucheng. Zoucheng Government News. Retrieved from http://www.mencius.gov.cn/news01/03/10191.html
- Dufhues, T., Buchenrieder, G., & Fischer, I. (2006). Social capital and rural development: literature review and current state of the art. Leibniz Institute of Agricultural Development in Central and Eastern Europe Discussion paper No. 96. Halle, Germany: IAMO.
- Dupont, W. D., & Plummer, W. D. Jr. (2003). Density distribution sunflower plots. *Journal of Statistical Software*, 8, 1-11.
- Durlauf, S.N. & Quah, D.T.(1998). The new empirics of economic growth. NBER Working Paper No. 6422. Cambridge, MA: National Bureau of Economic Research.
- Dziuban, C.D. & Shirkey, E.S. (1974). When is a correlation matrix appropriate for factor analysis? Some decision rules. *Psychological Bulletin*, *81*(6), 358-361.
- Elliot, J. A. (1999). *An Introduction to Sustainable Development* (2nd Edition). London, UK: Routledge.
- Ellis, F. (1999). Rural Livelihood Diversity in Developing Countries: Evidence and Policy Implications, *Natural Resource Perspectives*, 40 (April, 1999). London, UK: Overseas Development Institute (ODI). Retrieved from http://www.odi.org.uk/nrp/40.html
- Ellis, F. (2000). *Rural Livelihoods and Diversity in Developing Countries*, Oxford, UK: Oxford University Press.
- Ellis, F., & Bahiigwa, G. (2003). Livelihoods and Rural Poverty Reduction in Uganda. *World Development*, *31*(6), 997-1013.
- Ellis, F. & Freeman, H.A. (2004). Rural livelihoods and poverty reduction strategies in four African countries. *Journal of Development Studies*, 40(4), 1-30.
- Ellis, F., Kutengule, M., & Nyasulu, A. (2003). Livelihoods and Rural Poverty Reduction in Malawi. *World Development*, *31*(9), 1495-1510.
- Erzurumlu, S. S., & Erzurumlu, Y. O. (2015). Sustainable mining development with community using design thinking and multi-criteria decision analysis. *Resources Policy*, *46*, 6-14.
- Fafchamps, M., & Minten, B. (2002). Property rights in a flea market economy. *Economic Development and Cultural Change*, 49(2), 229-267.
- Fan, S. (1991). Effects of technological change and institutional reform on production growth in Chinese agriculture. *American Journal of Agricultural Economics*, 73(2), 266-275.

- Fan, S., & Zhang, X. (2002). Production and productivity growth in Chinese agriculture: new national and regional measure. *Economic Development and Cultural Change*, 50(4), 819-838.
- Fan, S., Wailes, E J., & Gramer, G L. (1995). Household Demand in Rural China: A Two-Stage LES-AIDS Model. *American Journal of Agricultural Economics*, 77(1), 54-62.
- Fan, S., Zhang, X., & Robinson, S. (2003). Structural change and economic growth in China. *Review of Development Economics*, 7(3), 360-377.
- Fang, Y. D., Zhang, C. Q., & Sheng, C. (2012). An alternative livelihood choice for rural households in Zoucheng-rural tourism. *Peasant Daily*. Retrieved from http://www.mencius.gov.cn/news01/system/22298.html
- Farmer, A. M. (1993). The effect of dust on vegetation–a review. *Environmental Pollution*, 79(1), 63-75.
- Fearniside, M. P. (2001). Saving tropical forests as a global warming countermeasure: an issue that divides the environmental movement. *Ecological Economics* 39(2), 167-184.
- Fei, X. (1947/1992). *From the Soil: the foundations of Chinese society*. Berkeley, CA: University of California Press.
- Ferrary, M. (2003). Trust and social capital in the regulation of lending activities. *Journal of Socio-Economics*, *31*(6), 673-699.
- Ferrol-Schulte, D., Gorris, P., Baitoningsih, W., Adhuri, D. S., & Ferse, S. C. A. (2015). Coastal livelihood vulnerability to marine resource degradation: A review of the Indonesian national coastal and marine policy framework. *Marine Policy*, 52,163-171.
- Field, J. (2003). Social capital. London: Routledge.
- Filmer, D., & Pritchett, L. H. (2001). Estimating wealth effect without expenditure data - or tears: An application to educational enrolments in states of India. *Demography*, 38(1), 115-132.
- Fishbein, M. and Ajzen, I. (1975) *Belief, Attitude, Intention and Behaviour: an introduction to theory and research.* Reading, MA: Addison-Wesley.
- Flap, H. (2002). No man is an island. In E. Lazega & O. Favereau (Eds.), *Conventions and Structures*. Oxford, UK: Oxford University Press.
- Fleury, P., Plagnes, V., & Bakalowicz, M. (2007). Modelling of the functioning of karst aquifers with a reservoir model: Application to Fontaine de Vaucluse (South of France). *Journal of Hydrology*, 345(1-2), 38-49.
- Florida, R. (2002, May). The rise of the creative class: why cities without gays and rock bands are losing the economic development race. *Washington Monthly*. Retrieved from

http://www.washingtonmonthly.com/features/2001/0205.florida.html

- Fock, K. Y. & Woo, K. (1998). The China Market: Strategic Implications of Guanxi. *Business Strategy Review*, 7(4), 33-44.
- Foley, M. W., & Edwards, B. (1999). Is it time to disinvest in social capital? *Journal of Public Policy*, *19*(2), 141-173.

- Fong, E., & Ooka, E. (2002). The social consequences of participating in the ethnic economy. *International Migration Review*, *36*(1), 125-146.
- Ford, A. (1999). *Modeling the Environment: An Introduction to System Dynamics Modeling of Environmental Systems*. Washington, DC: Island Press.
- Ford, A., & Flynn, H. (2005). Statistical screening of system dynamics models. *System Dynamics Review*, 21(4), 273-302.
- Forrester, J. W., & Senge, P. (1980). Tests for building confidence in System Dynamics Models, *TIMS Studies in the Management Sciences*, *14*, 209-228.
- Franke, S. (2005). *Measurement of Social Capital: Reference Document for Public Policy Research, Development and Evaluation*. Ottawa, ON: Government of Canada.
- Fraser, E. D. G., Dougill, A. J., & Reed, M. S. (2010). Anticipating vulnerability in food systems to global environmental change: challenges of using dynamic systems approaches and the case of pastoral systems in Botswana. *Ecology and Society*, 15(2), 17-48.
- Fu, J. (2009). Tourism in mountainous areas-an investigation into the farmhouse tourism in Muyang, Nanjiang County, Sichuan Province. *Market Modernization*, 567(6), 228-229.
- Fu, Y. (2005). Measuring personal networks with daily contacts: A single-item survey question and the contact diary. *Social Networks*, 27(3), 169-186.
- Fung, K., Wu, Y., & Zhuo, S. (2013). Re-examination of the surplus agricultural labour in China. *China Agricultural Economic Review*, 5(2), 197-212.
- Gao, S., Xu, K., & Yang, J. (2008). Managerial ties, absorptive capacity, and innovation. *Asia Pacific Journal of Management*, 25(3), 395-412.
- Garcia-Barrios, L. E., Speelman, E. N., & Pimm, M. S. (2008). An educational simulation tool for negotiating sustainable natural resource management strategies among stakeholders with conflicting interests. *Ecological Modelling*, 210(1-2), 115-126.
- Gargiulo, M., & Benassi, M. (1999). The dark side of social capital. In R. Leenders, & S. Gabbay (Eds.), *Gorporate social capital and liability* (pp.298-322). Boston: Kluwer.
- Garrod, B., Wornell, R., & Youell, R. (2006). Re-conceptualising rural resources as countryside capital: the case of rural tourism. *Journal of Rural Studies*, 22,117-128.
- Geddie, M. W., DeFranco, A. L., & Geddie, M. F. (2002). From guanxi to customer relationship marketing: how the constructs of guanxi can strengthen CRM in the hospitality industry. *Journal of Travel & Tourism Marketing*, *13*(3), 19-33.
- General Statistics Office (GSO). (2006). *Questionnaire on Household Living Standard Survey 2006 (VHLSS-2006)*. Hanoi, Vietnam: General Statistics Office.
- Gertler, P., Levine, D.I., Moretti, E. (2006). Is Social Capital the Capital of the Poor? The Role of Family and Community in Helping Insure Living Standards against Health Shocks. *CESifo Economic Studies*, *52*(3), 455-499.

- Ghose, M. K., & Majee, S. R. (2000). Sources of air pollution due to coal mining and their impacts in Jharia coalfield. *Environment International 26*(1-2), 81-85.
- Giannanasi, F., Lovett, P., & Godwin, A. N. (2001). Enhancing confidence in discrete event simulations. *Computers in Industry*, 44, 141-157.
- Gibson, J., Huang, J., & Rozelle, S. (2003). Improving estimates of inequality and poverty from urban china's household income and expenditure survey. *Review* of *Income and Wealth*, 49(1), 53-68.
- Gilchrist, A., & Kyprianou, P. (2011). *Social networks, poverty and ethnicity*. JRF programme paper: Poverty and ethnicity. York, UK: Joseph Rowntree Foundation.
- Gile, K. J., & Handcock, M. S. (2011). Improved Inference for Respondent-Driven Sampling Data with Application to HIV Prevalence Estimation. *Journal of the American Statistical Association*, *106*(493), 135-146.
- Glaeser, E. L., Laibson, D. I., & Sacerdote, B. S. (2002). An economic approach to social capital. *The Economic Journal*, *112*(483), F437-F458.
- Golley, J., & Meng, X. (2011). Has China Run Out of Surplus Labour? *China Economic Review*, 22(4), 555-572.
- Gomes, A. (1993). Integrating Tourism and Agricultural Development. In D. J. Gayle & J. N. Goodrich (eds), *Tourism Marketing and Management in the Caribbean* (pp. 155-156). New York, NY: Routledge.
- Gong, Y. L., Koplan, J. P., et al. (1995). Cigarette-smoking in China-prevalence, characteristics, and attitudes in Minhang District. *JAMA*, 274(15), 1232-1234.
- Goodman, L. A. (1961). Snowball sampling. *Annals of Mathematical Statistics*, 32(1), 148-170.
- Gradiz, L., Sugimoto, A., Ujihara, K., Fukuhara, S., Kahi, A.K., Hirooka, H. (2007).
 Beef cow-calf production system integrated with sugarcane production: simulation model development and application in Japan. *Agricultural Systems*, 94(3), 750-762.
- Graham, John L., & Lam, N. Mark. (2003). The Chinese Negotiation. *Harvard Business Review*, 81(10), 82-91.
- Granovetter, M. (1973). The strength of weak ties. *American Journal of Sociology*, 78(6), 1360-1380.
- Granovetter, M. (1974). *Getting a Job: a study of contracts and careers*. Cambridge, MA: Harvard University Press.
- Granovetter, M. (1985). Economic action and social structure: the problem of embeddedness. *American Journal of Sociology*, *91*(3), 481-510.
- Granovetter, M. (1995). *Getting a job: A study of contacts and careers (2nd edition)*. University of Chicago Press, Chicago.
- Granovetter, M. S. (2005). The impact of social structure on economic outcomes. *Journal of Economic Perspectives*, 19(1), 33-50.
- Grasso, M. (1998). Ecological-economic model for optimal mangrove trade-off between forestry and fishery production: Comparing a dynamic optimization and simulation model. *Ecological Modelling*, *112*(2-3), 131-150.
- Green, M. L., Espinal, L., Traversa, E., & Amis, E. J. (2012). Materials for
sustainable development. MRS Bulletin, 37(4), 303-309.

- Green, S.B. (1991). How many subjects does it take to do a regression analysis? *Multivariate Behavioral Research 26*, 499-510.
- Grootaert, C. (1998). *Social capital, household welfare and poverty in Indonesia*. Washington, DC: World Bank.
- Guan, D., Gao, W., & Fukahori, H. (2010). Modeling the relationship between economic growth, resource consumption and environment pollution by system dynamics model. *Journal of Architecture and Planning*, 75(647), 165-174.
- Guiso, L., Sapienza, P., & Zingales, L. (2000). *The role of social capital in financial development*, Working Paper Series No. 7563. Cambridge, MA: National Bureau of Economic Research,.
- Gujarati, D. N. (2003). *Basic Econometrics* (4th edition). New York, NY: McGraw-Hill.
- Gunson, A. J., & Yue, J. (2001). *Artisanal Mining in the People's Republic of China*. Report for MMSD. London, UK: IIED
- Guo, J. X. (2007). The Estimation of Surplus Labour in Chinese Agriculture in 1996-2005: A Stochastic Frontier Approach. *Nankai Economic Studies*, *4*, 72-81.
- Guo, L., & Liu, J. (2012). Regional tourism to regional transformation: the case of Jining area of Shandong Province in China. 6th Conference of the International Forum of Urbanism: Tourbanism, Barcelona (pp. 1-10). Barcelona, Spain: IFoU.
- Gupta, D. K., Rai, U. N., Tripathi, R. D., & Inouhe, M. (2002). Impacts of fly-ash on soil and plant responses, *Journal of Plant Research*, *115*(6), 401-409.
- Gursoy, D., Chi, C. G., & Dyer, P. (2010). Locals' attitudes toward mass and alternative tourism: The case of Sunshine Coast, Australia. *Journal of Travel Research*, 49(3), 381-394.
- Habibah, A., Hamzah, J., & Mushrifah, I. (2010). Sustainable Livelihood of the Community in Tasik Chini Biosphere Reserve. *Journal of Sustainable Development*, 3(3), 184-196.
- Haggblade, S., Hazell, P. B. R. & Reardon, T. (Eds.) (2007). Transforming the rural nonfarm economy: opportunities and threats in the developing world. Baltimore, MD: Johns Hopkins University Press.
- Hair, J. F., Blac,, W. C., Babin, B. J., Anderson, R. E., & Tatham, R. L. (2006). *Multivariate data analysis*. Upper Saddle River, NJ: Pearson International Edition.
- Hal, J. L., Hills, M. J., Huang, T., (2007). A simulation model using system dynamic method for construction and demolition waste management in Hong Kong. *Journal of Construction Innovation*, 7(1), 7-21.
- Han, Y., Wang, G., Cravotta III, C. A., Hu, W., Bian, Y., Zhang, Z., & Liu, Y. (2013). Hydrogeochemical evolution of Ordovician limestone groundwater in Yanzhou, North China. *Hydrological Processes*, 27(16), 2247-2257.
- Handcock, M. S., & Gile, K. J. (2011). Comment: On the Concept of Snowball Sampling. *Sociological Methodology*, *41*(1), 367-371.

- Hannon, B., & Ruth, M. (1994). *Dynamic modelling*. New York, NY: Springer-Verlag.
- Hannum, E., & Adams, J. (2007). Beyond Cost: Rural Perspectives on Barriers to Education. In D. Davis & W. Feng (Eds.), *Creating Wealth and Poverty in China* (pp. 156-171). Redwood City, CA: Stanford University Press.
- Harpham, T., Grant, E., & Thomas, E. (2002). Measuring social capital within health surveys: Key issues. *Health Policy and Planning*, *17*(1), 106-111.
- Harris, R.J. (1985). *A primer of multivariate statistics* (2nd ed). New York: Academic Press.
- Harrod, R. F. (1939). An essay in dynamic theory. *Economic Journal*, 49(193), 14-33.
- Haynes, R. J. (2009). Reclamation and revegetation of fly ash disposal sites -Challenges and research needs. *Journal of Environmental Management*, 90(1), 43-53.
- Haywood, M. (2006). Evolution of Tourism Areas and the Tourism Industry. In R.Butler (Ed.), *The tourism life cycle Volume 1: Applications and modifications* (pp. 51-69). Clevedon, UK: Channel View Publishers.
- He, G., & Chen, M. (2012). Policy Assessment: A Case Study of Rural Tourism, Recreation and Leisure Development in Suburb of Chengdu, China. *Studies in Sociology of Science*, 3(2), 1-5.
- He, J. M. (2005). On the development of rural tourism in China's urban suburbs: a case study on the evolution of *Nongjiale* in Chengdu. *Tourism Tribune*, 20(6), 71-74.
- He, J. M., & Li H. (2003). Systematic analysis on factors affecting the development of rural tourism: a case study of Chengdu. Wuhan University Journal of Natural Sciences, 8(3B), 919-925.
- He, L., & Qi, Y. (2012). Research on Rural Tourism and Development of New Rural Construction Based on Coupling Theory. *Institute of Interdisciplinary Business Research*, 3(10), 65-75.
- He, S. J., & Su, G. Q. (2002). Utilization of waste land in mining area of China. *Resources Science*, 24(2), 17-21.
- Heilig, G. K. (2000). Can China feed itself? An analysis of China's food prospects with special reference to water resource. *International Journal of Sustainable Development and World Ecology*, 7(3), 153-172.
- Hekimoglu, M., & Yaman, B. (2010). Sensitivity Analysis of System Dynamics Models by Behavior Pattern Measures. In T-H. Moon (Ed), Proceedings of the 28th International Conference of the System Dynamics Society, July 25-29, Seoul, Korea. Albany, NY: System Dynamics Society. Retrieved from http://www.systemdynamics.org/conferences/2010/proceed/papers/P1260.pdf
- Henderson J. V. (2009). *Urbanization in China: Policy Issues and Options*. Report prepared for CERAP (China Economic Research and Advisory Programme).
- Hendryx, M., O'Donnell, K., & Horn, K. (2008). Lung cancer mortality is elevated in coal-mining areas of Appalachia. *Lung Cancer*, 62(1), 1-7.

- Hentschel, T., Hruschka, F., & Priester, M. (2002). *Global Report on Artisanal & Small-Scale Mining*. Report No. 70, January 2002, prepared for MMSD project.London, UK: International Institute for Environment and Development (IIED)
- Hermans, D. (1981). The Encounter of Agriculture and Tourism: A Catalan Case. *Annals of Tourism Research*, 8(3), 462-479.
- Hernandez, J. M., & Leon, C. J. (2007). The interactions between natural and physical capitals in the tourist lifecycle model. *Ecological Economics*, 62(1), 184-193.
- Herreo, M., Gonzales-Estrada, E., Thornton, P. K., Quiros, C., Waithaka, M. M., Ruiz, R., & Hoogenboom, G. (2007). IMPACT: generic household-level databases and diagnostics tools for integrated crop-livestock systems analysis. *Agricultural Systems*, 92, 240-265.
- Hesketh, S, R., & Everitt, B, S. (2004). *A Handbook of Statistical Analyses using Stata* (3rd), Boca Raton, FL: CRC Press.
- Higuchi, S., Matsushita, S., Muramatsu, T., Murayama, M., & Hayashida, M. (1996). Alcohol and aldehyde dehydrogenase genotypes and drinking behavior in Japanese. *Alcoholism Clin Exp Res* 20, 493-497.
- Hilson, G. (2002). An overview of land use conflicts in mining communities. *Land Use Policy*, *19*(1), 65-73.
- Hitt, M. A., Lee, H. U., & Yucel, E. (2002). The importance of social capital to the management of multinational enterprise: Relational networks among Asian and Western firms. *Asia Pacific Journal of Management*, 19(2-3), 353-372.
- Hobbs, N. T., Schimel, D. S., Owensby, C. E., Ojima, D. S. (1991). Fire and grazing in the tall grass prairie: contingent effects on nitrogen budgets. *Ecology*, 72, 1374-1382.
- Holling, C. S., Berkes, F., & Folke, C. (1998). Science, sustainability and resource management. In F. Berkes, C. Folke, & J. Colding (Eds.), *Linking Social and Ecological Systems: Management Practices and Social Mechanisms for Building Resilience* (pp.342-363). Cambridge, UK: Cambridge University Press.
- Holzmann, R., & Jorgensen, S. (1999). Social protection as social risk management: Conceptual underpinnings for the Social Protection Sector Strategy Paper. *Journal of International Development*, 11(7), 1005-1027.
- Hoogeveen, J. G. M. (2001). Income risk, consumption security and the poor. Oxford *Development Studies*, *30*(1), 105-121.
- Hook, M., Zittel, W., Schindler, J., & Aleklett, K. (2010). Global coal production outlooks based on a logistic model. *Fuel*, 89(1), 3546-3558.
- Horsley, J., Prout, S., Tonts, M., & Ali, S. H. (2015). Sustainable livelihoods and indicators for regional development in mining economies. *Extractive Industries and Society*, 2(2), 368-380.
- Hospers, G. (2002). Industrial Heritage Tourism and Regional Restructuring in the European Union. *European Planning Studies*, *10*(3), 397-404.
- Houghton, J., Siegel, M., Wirsch, A., Moulton, A., Madnick, S., & Goldsmith, D. (2014). A survey of methods for data inclusion in System Dynamics models:

methods, tools and applications. Composite Information Systems Laboratory (CISL) Working Paper CISL#2014-03. Cambridge, MA:Massachusetts Institute of Technology. Retrieved from http://web.mit.edu/smadnick/www/wp/2014-03.pdf

- Houthakker, H. S. (1957). An international comparison of household expenditure patterns, commemorating the centenary of Engel's Law. *Econometrica*, 25(4), 532-551.
- Hovinen, G. (2002). Revisiting the Destination Life Cycle. *Annals of Tourism Research*, 29(1), 209-230.
- Hsiao, C. T., Liao, W. Y., & Chen, C. W. (2006). Causal model of the relationship between service quality, satisfaction, perceived value and loyalty: A case study of the National Museum of Natural Science, *The Museology Quarterly*, 20(2), 81-96.
- Hsiao, C., & Yao, M. (2012). System Dynamics approach to visitors' long-term satisfaction with museum: a case study of the national museum of natural science. *International Journal of Electronic Business Management*, 10(2), 113-121.
- Hu, J. L., & Lee, Y. C. (2008). Efficient three industrial waste abatements for regions in China. *International Journal of Sustainable Development and World Ecology*, 15(1), 132-144.
- Hu, W., & Wall, G. (2008). Environmental Management, Environmental Image and the Competitive Tourist Attraction. *Journal of Sustainable Tourism*, 13(6), 617-635.
- Hu, Y.H. (2008). A study on the participation of community in rural tourism. *Business Times*, *30*, 89-90.
- Hu, Z., & Zhao, Y. (2007). *Report of Yangzhou Coal Mining Subsidence*. pp. 20-43. Beijing: China University of Mining & Technology.
- Hu, Z., Xu, X., & Zhao, Y. (2012). Dynamic monitoring of land subsidence in mining area from multi-source remote-sensing data-a case study at Yanzhou, China. *International Journal of Remote Sensing*, 33(17), 5528-5545.
- Huang, G., Zhou, W., & Ali, S. (2011). Spatial patterns and economic contributions of mining and tourism in biodiversity hotspots: A case study in China. *Ecological Economics*, 70(8), 1492-1498.
- Huang, J., & Rozelle, S. (2006). The emergence of agricultural commodity markets in China. *China Economic Review*, *17*(3), 266-280.
- Hubbert, M. K. (1956). *Nuclear energy and the fossil fuels*. Report presented before the Spring Meeting of the Southern District. Plaza Hotel, San Antonio, Texas: American Petroleum Institute; March 7–9, 1956. Publication No 95. Houston, TX: Shell Development Company. Retrieved from http://www.hubbertpeak.com/hubbert/1956/1956.pdf
- Hurlbert, J. S., Haines, V. A., & Beggs, J. J. (2000). Core networks and tie activation: What kind of routine networks allocate resources in non-routine situations? *American Sociological Review*, 65(4), 598-618.

- Illenberger, J., & Flotterod, G. (2012). Estimating network properties from snowball sampled data. *Social Networks*, *34*(4), 701-711.
- ILO (International Labor Organization). (1994). *Household Income and Expenditure Surveys*, Yearbook of Labour Statistics, Sources and Methods, Volume 6. Geneva, Switzerland: ILO.
- ILRI. (2000). *Livestock strategy to 2010: making the livestock revolution work for the poor*. Nairobi, Kenya: ILRI.
- Inkeles, A. (1983). *Exploring Individual Modernity*. New York, NY: Columbia University Press.
- Inkeles, A., & Baker, W. (2000). Modernization, cultural change, and the persistence of traditional values. *American Sociological Review*, 65(1), 19-51.
- Inkeles, A., & Smith, D. (1974). *Becoming Modern: Individual Change in Six Developing Countries*. Cambridge, MA: Harvard University Press.
- Jamal, T., & Getz, D. (1995). Collaboration theory and community tourism planning. *Annals of Tourism Research*, *21*(1), 121-139.
- Jankowski, A. T, & Molenda, T. (2007). Anthropogenic water bodies in Upper Silesia. Cz. 4.Środowiska powierzchniowe – zbiorniki zapadliskowe i w nieckach osiadań. Przyr. Górn. Śl., 48, 10-11. [in Polish]
- Jansen, G. P., Pender, J., Damon, A., Wielemaker, W., & Schipper, R. (2006). Policies for sustainable development in the hillside areas of Honduras: a quantitative livelihood approach. *Agricultural Economics*, *34*(2), 141-153.
- Jansen, H., Pender, J., Damon, A., & Schipper, R. (2006). Rural development policies and sustainable land use in the hillside areas of Honduras: a quantitative livelihoods approach. (Research Report). Washington, DC: International Food Policy Research Institute. Retrieved from http://www.ifpri.org/publication/rural-development-policies-and-sustainableland-use-hillside-areas-honduras
- Janvry, A., Sadoulet, E., & Zhu, N. (March 2005). *The Role of Non-Farm Incomes in Reducing Rural Poverty and Inequality in China*. Retrieved from http://are.berkeley.edu/~sadoulet/papers/RNF-Nong.pdf
- Jenkins, C. L., & Henry, B. M. (1982). Government involvement in tourism in developing countries. *Annals of Tourism Research*, 9(4), 499-519.
- Jennings, N. S. (1999).Small-scale Gold Mining: Examples from Bolivia, Philippines and Zimbabwe. Sectoral Activities Programme Working Paper, SAP 2.76/WP.130, Geneva, Switzerland: ILO.
- Jia, R., & Ding, R. (2002). System dynamics complex analysis of dynamic feedback. Beijing: Higher Education Press.
- Jogo, W., & Hassan, R. (2010). Balancing the use of wetlands for economic wellbeing and ecological security: The case of the Limpopo wetland in southern Africa. *Ecological Economics*, 69(7), 1569-1579.
- John, G. W. (1998). A simulation model to illustrate feedbacks among resource consumption, production, and factors of production in ecological-economic systems. *Ecological Modelling*, *112*, 227-247.

- Johnson, D. G. (1997). Richard T. Ely lecture: agriculture and the wealth of nations. *American Economic Review*, 97, 1-11.
- Johnston, R. J., & Tyrrell, T. J. (2005). A Dynamic Model of Sustainable Tourism. *Journal of Travel Research*, 44(2), 124-134.
- Ju, K. Y., Zhou, D. Q., Wang, Q. W., & Wu, J. M. (2010). Relationships between the energy consumption structure and energy security in China. *Resources Science*, 32(9), 1692-1697.
- Kah, J. M. L., Olds, D. L., & Kah, M. M. O. (2005). Microcredit, Social Capital, and Politics: The Case of a Small Rural Town – Gossas, Senegal. *Journal of Microfinance*, 7(1), 121-152.

Kaiser, H.F. (1974). An index of factorial simplicity. Psychometrika, 39(1), 31-36.

- Kang, C., & Wang, Y. (2003) Sex ratio at birth. In *Theses Collection of 2001* National Family Planning and Reproductive Health Survey (pp.88-98).
 Beijing: China Population Publishing House.
- Kara, A., & Zaim, S. (2012). Technology and Job Competence in the Turkish Labour Market: A Model and Simulations. *Social and Behavioral Sciences*, 58, 820-829.
- Karavezyris, V., Timpe, K., & Marzi, R. (2002). Application of system dynamics and fuzzy logic to forecasting of municipal solid waste. *Mathematics and Computers in Simulation* 60(3), 149-158.
- Keating, B. A., Carberry, P. S., Hammer, G. L., Probert, M. E., Robertson, M. J., Holzworth, D., ... Bristow, K. L. (2003). An overview of APSIM, a model designed for farming systems simulation. *European Journal of Agronomy 18*, 267-288.
- Keyim, P., Yang, D., & Zhang, X. (2005). Study of rural tourism in Turpan, China. *Chinese Geographical Science*, *15*(4), 377-282.
- Khadaroo, J., & Seetanah, B. (2008). The role of transport infrastructure in international tourism development: A gravity model approach. *Tourism Management*, 29(5), 831-840.
- Kipnis, A. B. (1997). *Producing Guanxi: sentiment, self and subculture in a North China village.* Durham, NC: Duke University Press.
- Kirkwood, C. W. (1998). *System dynamics methods: a quick introduction*. Retrieved from http://www.public.asu.edu/~kirkwood/sysdyn/SDIntro/SDIntro.htm
- Knack, S. & Keefer, P. (1997). Does social capital have an economic payoff? A cross-country investigation, *Quarterly Journal of Economics*, 112(4), 1251-1288.
- Knight, J., & Song, L. (1999). *The Rural-Urban Divide: Economic Disparities and Interactions in China*, Oxford, UK: Oxford University Press.
- Knight, J., & Song, L. (2003). Chinese Peasant Choices: Migration, Rural Industry or Farming. *Oxford Development Studies*, *31*(2), 123-147.
- Knight, J., & Song, L. (2005). *Towards a Labour Market in China*, Oxford, UK: Oxford University Press.
- Knight, J., & Yueh, L. (2008). The role of social capital in the labor market in China. *Economics of Transition*, *16*(3), 389-414.

- Knight, J., Heng, Q., & Li, S. (2011). The Puzzle of Migrant labour Shortage and Rural Labour Surplus in China. *China Economic Review*, 22(4), 585-600.
- Knofczynski, G.T., & Mundfrom, D. (2007). Sample sizes when using multiple linear regression for predition. Educational and Psychological Measurement 2008, 68(431), 431-442.
- Kollmair, M., & Gamper, S. (2002). *The sustainable livelihoods approach*. (Input Paper for the Integrated Training Course of NCCR North-South, Aeschiried, Switzerland). Development Study Group, University of Zurich. Retrieved from: http://www.nccr-

pakistan.org/publications_pdf/General/SLA_Gamper_Kollmair.pdf

- Komarek, A. M., Waldron, S. A., & Brown, C. G. (2012). An exploration of livestock-development policies in western China. *Food Policy*, *37*(1), 12-20.
- Kong, L. G. (2007). Research on hydrogeological characteristics and classification of coal of the mining subsidence land. *Coalmining*, *12*, 15-16.
- Kong, T. (2007). A Selective Review of Recent Developments in the Economic Growth Literature. *Asian-Pacific Economic Literature*, 21(1), 1-33.
- Krishna, A. (2001). Moving from the stock of social capital to the flow of benefits: the role of agency. *World Development, 29*(6), 925-943.
- Kwan, F. (2009). Agricultural labour and the Incidence of Surplus Labour: Experience from China during Reform. *Journal of Chinese Economy and Business Studies*, 7(3), 585-600.
- Lam, K. J. (2003). Earnings advantages of Party members in urban China. Business Research Centre Working Paper, Department of Economics, Hong Kong, China: Hong Kong Baptist University.
- Lam, T. H., He, Y., Li, L. S., He, S. F., & Liang, B. Q. (1997). Mortality attributable to cigarette smoking in China. *JAMA*, 278(18), 1505-1508.
- Lambini, C. K., & Nguyen, T. T. 2014). A comparative analysis of the effects of institutional property rights on forest livelihoods and forest conditions: evidence from Ghana and Vietnam. *Forest Policy and Economics*, 38, 178-190.
- Latimer, H. (1985). Developing Island Economies: Tourism V. Agriculture. *Tourism Management*, *6*, 32-42.
- Lau, R. S. (2002). Competitive factors and their relative importance in the US electronics and computer industries. *International Journal of Operations & Production Management*, 22(1), 125-135.
- Leana, C., & Van Buren, III HJ. (1999). Organizational social capital and employment practices. *Acad. Manage. Rev.*, 24(3), 538-555.
- Lee, C. K. (1998). *Gender and the South China Miracle: Two Worlds of Factory Women*, Berkeley, CA: University of California Press.
- Lejano, R. P., & Wessel, A. T. (2006). Community and economic development: seeking common ground in discourse and in practice. *Urban Studies*, *43*(9), 1469-1489.
- Leung, J. C. B. (2003). Social security reforms in China issues and prospects. *International Journal of Social Welfare*, *12*(2), 73-85.

- Lew, A. A., & Wang, A. (2004). Sojourners, guanxi and clan associations: social capital and overseas Chinese tourism to China. In T.E. Coles & D.J. Timothy (Eds.), *Tourism, diasporas and space* (pp.202-214). London, UK: Routledge.
- Li, C., Frolking, S., Harriss, R.H., (1994). Modeling carbon biochemistry in agricultural soils. *Global Biogeochemical Cycles*, 8(3), 237-254.
- Li, H., & Zhang, J. (2004). *Fines, Limited Liability and Fertility*. Unpublished document. Department of Economics, The Chinese University of Hong Kong.
- Li, H., Dong, L., & Ren, J. (2015). Industrial symbiosis as a countermeasure for resource dependent city: a case study of Guiyang, China. *Journal of Cleaner Production*, 107, 252-266.
- Li, H., Zhang, J., & Zhu, Y. (Feb 2008). The quantity-quality trade-off of children in a developing country: identification using Chinese twins. *Demography*, 45(1), 223-243.
- Li, L., Hesketh, T., & Zhu, W. X. (2005). The effect of China's one-child family policy after 25 years. *Health Policy Reports*, *353*(11), 1171-1176.
- Li, X. (1997). Driving forces of arable land conversion in China. Interim Report IR-97-076/Sept. Laxenburg, Austria: International Institute for Applied Systems Analysis.
- Li, Y. (2000). China Land Resources. Beijing: China Land Publishing House.
- Li, Y., Lai, K., & Feng, X. (2007). The problem of 'Guanxi' for actualizing community tourism: a case study of relationship networking in China. *Tourism Geographies*, 9(2), 115-138.
- Li, Y-F., Liu, Y., Du, Z., & Chen, J. (2009). Effect of coal resources development and compensation for damage to cultivated land in mining areas, *Mining Science and Technology*, 19(5), 620-625.
- Li, Y-P. (2000). Ethnic tourism: a Canadian experience. *Annals of Tourism Research*, 27(1), 115-131.
- Li, Z. (2010). Quantitative analysis of sustainable energy strategies in China. *Energy Policy*, *38*, 2149-2160.
- Li. K. (2012). The economy transformation of coal resource-based city: a case study of Zoucheng, Shandong Province. (MBA dissertation, Shandong Normal University). *Dissertation Abstracts*, 10445, F293.1.
- Lin, B., & Liu, J. (2010). Estimating coal production peak and trends of coal imports in China. *Energy Policy*, *38*(1), 512-519.
- Lin, G. C. S., & Ho, S. P. S. (2003). China's land resources and land-use change: insights from the 1996 land survey. *Land Use Policy*, *20*, 87-107.
- Lin, J., Si, X. S. (2010). Can guanxi be a problem? Contexts, ties, and some unfavorable consequences of social capital in China. Asia Pacific Journal of Management, 27(3), 561-581.
- Lin, N. (1982). Social resources and instrumental action. In P. V. Marsden & N. Lin (Eds.) Social structure and network analysis (pp. 131-145). Beverly Hills, CA: Sage.
- Lin, N. (1989). Chinese Family Structure and Chinese Society. *Bulletin of the Institute of Ethnology*, 65,382-399.

- Lin, N. (1999a). Building a network theory of social capital. *Connections*, 22(1), 28-51.
- Lin, N. (1999b). Social networks and status attainment. *Annual Review of Sociology*, 25(1), 467-487.
- Lin, N. (2001). *Social Capital: A Theory of Social Structure and Action*. New York, NY: Cambridge University Press.
- Lin, N. (2001a). Guanxi: a conceptual analysis. In A. So., N. Lin., & D. Poston (Eds.), *The Chinese Triangle of Mainland, Taiwan, and Hong Kong: Comparative Institutional Analysis* (pp.153-166). Westport, CT: Greenwood.
- Lin, N. (2001b). Building a network theory of social capital. In N. Lin., K. S. Cook., & R. S. Burt (Eds.), *Social capital: Theory and research* (pp. 3-29). Hawthorne, NY: Aldine de Gruyter.
- Lin, N. (2005). *A Network Theory of Social Capital*. New York, NY: Oxford University Press.
- Lin, N., & Duming, M. (1986). Access to Occupations through Social Ties. Social Networks 8,365-385.
- Lin, N., & Erickson, B. H. (Eds.). (2008). *Social Capital: An International Research Program*. New York, NY: Oxford University Press.
- Lin, N., Ensel, W., & Vaughn, J. (1981). Social Resources and the Strength of Weak Ties: Structural Factors on Occupational Status Attainment. *American Sociological Review*, 46(4), 393-405.
- Lin, N., Fu, Y-C. & Hsung, R-M. (2001). The position generator: Measurement Techniques for Investigations of Social Capital. In N. Lin., K. Cook, & R. S. Burt (Eds.), *Social Capital: Theory and Research* (pp.57-81). New Brunswick, NJ: Transaction Publishers
- Lin, N., Ye, X. & Ensel, W. (2000). Revisiting Social Support: Integration of Its Dimensions. Formosa Journal of Mental Health, 13(2), 37-64.
- Lin, Y. (1992). Rural reforms and agricultural growth in China. *The American Economic Review*, 82(1), 34-51.
- Liu, A. & Wall, G. (2006). Planning tourism employment: A developing country perspective. *Tourism Management*, 27(1), 159-170.
- Liu, G. (2000). *The beautiful views of Zoucheng city*. The Zoucheng post office: Zoucheng
- Liu, J., Zhan, J., & Deng, X. (Aug, 2005). Spatio-temporal patterns and driving forces of urban land expansion in China during economic reform era. *Royal Swedish Academy of Sciences*, 34(6), 450-455.
- Liu, Q. S. & Liu, G. Z. (2000). *Reform and Development in a County economy: Zoucheng Model*. (S.Y. Liu, Trans.). Beijing: China Economic Publishing House.
- Liu, W. (2000). Theoretic reflections of community participation in tourism development. *Journal of Tourism, 1*, 47-52.
- Liu, Y., & Wang, X. (2005). Technological progress and Chinese agricultural growth in the 1990s. *China Economic Review*, *16*(4), 419-440.

Liu, Z. (2003). The economic impact and determinants of investment in human andpolitical capital in China. *Economic Development and Cultural Change*, 51, 823-849.

Livestock Environment and Development Initiative (LEAD). (2007). *Livestock and environment toolbox. FAQ.* Rome, Italy: FAO. Retrieved from http://www.fao.org/agriculture/lead/tools/toolbox/en/

Loheide, S. P. (2011). Effects of changing channel morphology on vegetation, groundwater, and soil moisture regimes in ground-water-dependent ecosystems. *Geomorphology*, *126*(3), 364-376.

Lohmann, G., Albers, S., Koch, B., & Pavlovich, K. (2009). From hub to tourist destination- An explorative study of Singapore and Dubi's aviation-based transformation. *Journal of Air Transport Management*, *15*(5), 205-211.

Long, H., Tang, G., Li, X., & Heilig G. K. (2007). Socio-economic driving forces of land-use change in Kunshan, the Yangtze River Delta economic area of China. *Journal of Environmental Management*, 83, 351-364.

Lu, Y., Ruan, D., & Lai, G. (2013). Social capital and economic integration of migrants in urban China. *Social Networks*, *35*(3), 357-369.

Lucas, R. E. J., (1988). On the mechanics of economic development. *Journal of Monetary Economics*, 22, 3-42.

Ludi, E., & Slater, R. (2008). Using the Sustainable Livelihoods Framework to understand and tackle poverty (Briefing Note). London, UK: ODI.

Lundtorp, S., & Wanhill, S. (2001). The Resort Life Cycle Theory: Generating Processes and Estimation. *Annals of Tourism Research*, 28(4), 947-964.

Luo, B., & Fu, B. (2009). The farmland property rights deformity: the history, reality and reform. *China Agricultural Economic Review*, *1*(4), 435-458.

Lyon, F. (2000). Trust, Networks, and Norms: The Creation of Social Capital in Agricultural Economies in Ghana. *World Development*, *28*(4), 663-681.

Lyons, J. R., Ryder, R. C., Campbell, H., Clark, W. G., & Gough, J. (1974). Significance of irregular opacities in the radiology of coal workers' pneumoconiosis. *British Journal of Industrial Medicine*, *31*, 36-44.

Ma,Q. R. & Duan, X. D. (2010). The characteristics of agriculture in Daisu, Zoucheng. Zoucheng Government News. Retrieved from http://www.mencius.gov.cn/news01/04/8276.html

Maani, K. E, & Cavana, R. Y.(2007). Systems Thinking, System Dynamics: Managing Change and Complexity. (2nd ed). Auckland, New Zealand: Pearson Education NZ (Prentice Hall).

Macbeth, J., Carson, D., & Northcote, J. (2004). Social capital, tourism and regional development: SPCC as a basis for innovation and sustainability. *Current Issues in Tourism*, 7(6), 502-522.

MacDonald, R., & Jolliffe, L. (2003). Cultural rural tourism: evidence from Canada. *Annals of Tourism Research*, *30*(2), 307-322.

Mai, Y., & Peng, X. (2012). Estimating China's Rural Labor Surplus: A dynamic General Equilibrium Analysis. *The Chinese Economy*, *45*(6), 38-59.

Mantel, R.R., Rego, J.C., Goyena, P. (1994). *Optimal Control Modeling with Vensim: Applications to Public Finance*. Paper presented at the 12th International System Dynamics Conference, Sterling, Scotland, 1994 (pp. 118-129). Retrieved from http://www.systemdynamics.org/conferences/1994/proceed/papers_vol_2/mant

http://www.systemdynamics.org/conferences/1994/proceed/papers_vol_2/mant e118.pdf

- Marsden, P., & Hurlbert, J. S. (1988). Social Resources and Mobility Outcomes: A Replication and Extension. *Social Forces*, *66*(4), 1038-1059.
- Mayerthaler, A., Haller, R., & Emberger, G. (2009). A land-use/transport interaction model for Austria. *In Proceeding of the 27th International Conference of The System Dynamics*, Albuquerque/USA, pp.1-21. Retrieved from

http://www.systemdynamics.org/conferences/2009/proceed/papers/P1239.pdf

- Mayoux, L. (2001). Tackling the Down Side: Social Capital, Women's Empowerment and Micro-Finance in Cameroon. *Development and Change*, *32*(3), 435-464.
- Mbaiwa, J., & Stronza, A. L. (2010). The effects of tourism development on rural livelihoods in the Okavango Delta, Botswana. *Journal of Sustainable Tourism*, *18*(5), 635-656.
- McCallister, L., & Fischer, C. (1978). A procedure for surveying personal networks. *Sociological Methods and Research*, 7(2), 131-148.
- McCarty, C., Bernard, R. H., Killworth, P. D., Shelley, G. A., & Johnson, E. C. (1997). Eliciting representative samples of personal networks. *Social Networks*, *19*(4), 303-323.
- McIntosh, R. W. (1977). *Tourism: Principles, Practices, Philosophies. (Second ed.)*. Columbus, OH: Grid.
- McKenzie, D. J. (2005). Measuring inequality with asset indicators. *Journal of Population Economics*, 18(2), 229-260.
- Medina, D. O. (2010; 2012). *Is cropland-dominance in landscapes an alternative social-ecological regime? An empirical exploration of patterns in global cropland cover data.* Unpublished Master's thesis. Stockholm University, Sweden.
- Meng, L., Feng, Q., Zhou, L., Lu, P., & Meng, Q. (2009). Environmental cumulative effects of coal underground mining. *Procedia Earth and Planetary Science*, 1(1), 1280-1284.
- Meng, Q., Xu, L., Zhang, Y., Qian, J., Xin, Y., Guo, J., ... Barber, S. L. (2012).
 Trends in access to health services and financial protection in China between 2003 and 2011: a cross sectional study. *Lancet*, *379*(9818), 805-814.
- Meng, X., & Wu, H. (1997). The direct impact of the relocation of farm labour on Chinese grain production. *China Economic Review*, 7(2), 105-122.
- Mi, H. & Ji, G. L. (1999). Sustainable Development analysis: population, natural resources, environment and economic management in Chinese Counties (S.Y. Liu, Trans.). *Population and Economics*, 6, 17-24. Retrieved from http://WWW.CQVIP.COM/QK/95683X/199906/3755457.html

Mihalic, T. (2000). Environmental management of a tourist destination: A factor of tourism competitiveness. *Tourism Management*, 21(1), 65-78.

- Ministry of Land and Resources P.R.C (TMLR). (1999). China Land & Resources Almanac: 1999. Ministry of Land and Resources P.R.C.
- Mitchell, R. C. and Carson, R. T. (1989) Using Surveys to Value Public Goods: The Contingent Valuation Method, Washington, DC: Resources for the Future.

Montomery, M. R., Gragnolati, M., Burke, K. A., & Paredes, E. (2000). Measuring living standards with proxy variables. *Demography*, *37*(2), 155-174.

- Moore, A. D., Holzworth, D. P., Herrmann, N. I., Huth, N. I., & Robertson, M. J. (2007). The common modeling protocol: a hierarchical framework for simulation of agricultural and environmental systems. *Agricultural Systems*, 95(1), 37-48.
- Morgan, W. K., & Lapp, N. L. (1976). Respiratory disease in coal miners. *American Review of Respiratory Disease*, 113(4), 531-559.
- Moser, C. (1998). The asset vulnerability framework: Reassessing urban poverty reduction strategies. *World Development*, 26(1), 1-19.
- Moxnes, E. (2005). Policy sensitivity analysis: simple versus complex fishery models. *System Dynamics Review*, *21*(2), 123-145.
- Muetzelfeldt, R., & Massheder, J. (2003). The Simile visual modelling environment. *European Journal of Agronomy*, *18*, 345-358.
- Musk, A. W., Cotes, J. E., Bevan, C., & Campbell, M. J. (1981). Relationship between type of simple coal workers' pneumoconiosis and lung function. A nine year follow-up study of subjects with small rounded opacities. *British Journal of Industrial Medicine*, 38(4), 313-320.
- Mutz, G., & Klump, R. (Eds.). (2005). Modernization and social transformation in Vietnam: Social capital formation and institutional building. Hamburg, Germany: Institut f
 ür Asienkunde.
- Nair, P. K., & Sinha, J. K. (1990). Haul road dust consolidation in opencast mines a new approach. *Indian Journal of Environmental Protection*, 10(1), 35-39.
- Narayand, D. (1999). *Bonds and bridges: Social capital and poverty*. Policy Research Working Paper 2167/1999. Washington DC: World Bank.
- Narayand, D., & Pritchett, L. (1999). Cents and sociability: Household income and social capital in rural Tanzania. *Economic Development and Cultural Change*, 47(4), 871-897.
- National Bureau of Statistics of China (NBSC). (2000). *China Statistical Year Book* 2000. Beijing: China Statistical Publishing House.
- National Bureau of Statistics of China (NBSC). (2007). Statistical Yearbook of China. Beijing: China Statistical Press.
- National Bureau of Statistics of China (NBSC). (2016). National economy and society developed statistical bulletin 2015. Retrieved from http://www.ce.cn/xwzx/gnsz/gdxw/201602/29/t20160229_9163077.shtml
- Nejad, S. A. H. S., & Tularam, G.A. (2010). Modeling Tourist Arrivals in Destination Countries: An Application to Australian Tourism. *Journal of Mathematics and Statistics*, 6(4), 431-441.

Neto, A. C. L., Legey, L. F. L., Araya, M. C. G., & Jablonski, S. (2006). A system dynamics model for the environmental management of the Sepetiba Bay Watershed, Brazil. *Environmental Management*, *38*(5), 879-888.

New China News Agency (NCNA). (2008). Text of Chinese premier's speech at climate change technology seminar. Xinhua news agency domestic service, Bejing. Retrieved from

```
http://www.ccchina.gov.cn/WebSite/CCChina/UpFile/File367.pdf
```

Ng, M. K. (2002). Sustainable urban development issues in Chinese transitional cities: HongKong and Shenzhen. *International Planning Studies*, 7(1), 7-36.

Nguyen, T. T., Bauer, S., & Uibrig, H. (2010). Land privatization and afforestation incentive of rural farms in the Northern Uplands of Vietnam, *Forest Policy and Economics*, *12*(7), 518-526.

Nguyen, T. T., Do, T. L. Buhler, D., Hartje, R., & Grote, U. (2015). Rural livelihoods and environmental resource dependence in Cambodia. *Ecological Economics*, *120*(12), 282-295.

Nunkoo, R., & Ramkissoon, H. (2012). Power, trust, social exchange and community support. *Annals of Tourism Research*, *39*(2), 997-1023.

- Oi, J. C. (1999). *Rural China Takes Off: Institutional Foundations of Economic Reform*, Berkeley, CA: University of California Press.
- Oliveira, J. A. P. (2003). Governmental responses to tourism development: three Brazilian case studies. *Tourism Management*, 24(1), 97-110.
- Olson, E. (1992). *Crafting institutions for self-governing irrigation systems*. San Francisco, CA: Institute for Contemporary Studies Press.
- Olson, M. (1996). Big bills left on the sidewalk: why some nations are rich and others poor. *Journal of Economic Perspectives*, *10*(2), 3-24.

Onyx, J., & Bullen, P. (2000). Measuring Social Capital in Five Communities. *Journal of Applied Behavioral Science*, *36*(1), 23-42.

Padam, S., Arvind, P., & Abha, A. (April 2007). House-to-house survey vs. snowball technique for capturing maternal deaths in India: A search for a costeffective method. *Indian Journal of Medical Research*, 125(4), 550-556.

Page, S. J., Forber, P., & Lawton, G. R. (1999). Small business development and tourism. *Terra incognita? Tourism Management, 20*, 435-459.

Paldam, M. (2000). Social capital: One or many? Definition and measurement. *Journal of Economic Surveys*, 14(5), 629-653.

Pan, Y. J. (2014). The ecological tourism development in Zhongxin. Zoucheng News, Retrieved from http://zczww.xinzc.cn/index.php/cms/item-view-id-4635.shtml

- Pandey, V. C., Singh, N. (2010). Impact of fly ash incorporation in soil systems. *Agriculture, Ecosystems & Environment, 136*(1-2), 16-27.
- Park, D. B., Lee, K., Choi, H., & Yoon, Y. (2012). Factors influencing social capital in rural tourism communities in South Korea. *Tourism Management*, 33(6), 1511-1520.
- Park, S., & Luo, Y. (2001). Guanxi and organizational dynamics: organizational networking in Chinese firms. *Strategic Management Journal*, 22(5), 455-477.

- Parsons, D., Nicholson, C. F., Blake, R. W., Ketterings, Q. M., Ramirez-Aviles, L., Fox, D. G., ... Cherney, J. H. (2011). Development and evaluation of an integrated simulation model for assessing smallholder crop-livestock production in Yucatan, Mexico. *Agricultural Systems 104*, 1-12.
- Paul, B. G. & Vogl, C. R. (2013). Organic shrimp aquaculture for sustainable households in Bangladesh. *Ocean & Coastal Management*, 71, 1-12.
- Pearce, D. (2001). An Integrative Framework for Urban Tourism Research. *Annals* of *Tourism Research*, 28(4), 926-946.
- Pedercini, M. (2004). *Evaluation of alternative development strategies for Papua, Indonesia.* Unpublished doctoral thesis. University of Bergen, Norway.
- Pegg, S. (2006). Mining and poverty reduction: transforming rhetoric into reality. *Journal of Cleaner Production*, *14*(3-4), 376-387.
- Pend, M. (2001). How Entrepreneurs Create Wealth in Transitional Economies. *Academy of Management Executive*, 15(1), 95-108.
- Peng, M. W., & Quan, M. J. (2009). A micro-macro link during institutional transitions. In L. Keister (Ed.), *Work and organization in China*. Oxford, UK: Emerald Press.
- Pham, V. B. (1999). *The Vietnamese Family in Change: The Case of the Red River Delta*. Richmond, UK: Curzon Press.
- Portes, A. (1998). Social Capital: Its Origins and Applications in Modern Sociology. *Annual Review of Sociology*, 24, 1-24.
- Portes, A. (2000). The two meanings of social capital. *Sociological Forum*, 15(1), 1-12.
- Portes, A., & Landolt, P. (1996). Unsolved mysteries-The Toqueville files II : The downside of social capital. *The American Prospect*, 7(26), 18-21.
- Portes, A., & Sensenbrenner, J. (1993). Embeddedness and Immigration: Notes on the Social Determinants of Economic Action. *American Journal of Sociology* 98(6), 1320-1350.
- Poverty-wellbeing.net. (2007, Mar). *Working with a Sustainable livelihoods Approach (Version 1.1.)*. Retrieved from https://www.shareweb.ch/site/Poverty-Wellbeing/currentpovertyissues/Documents/sla.pdf
- Powell, D. R., Fair, J., LeClaire, R. J., Moore, L. M., & Thompson, D. (2005). Sensitivity Analysis of an Infectious Disease Model. 2005 International System Dynamics Conference, July 17-21, Boston, MA. Retrieved from http://www.systemdynamics.org/conferences/2005/proceed/papers/LECLA330. pdf
- Putnam, R. D (1993b). The Prosperous Community: Social Capital and Public Life. *The American Prospect*, 13, 35-42.
- Putnam, R. D. (1995) Bowling Alone: American's Declining Social Capital. *Journal of Democracy*, 6 (1), 65-78.
- Putnam, R.D. (1998). Foreword to social capital: Its importance to housing and community development. *Housing Policy Debate*, 9(1), v-viii.
- Putnam, R. D. (2000). *Bowling alone: the collapse and revival of American community*. New York, NY: Simon & Schuster.

Putnam, R. D., Leonardi, R., & Nanetti, R. Y. (1993a). Making Democracy Work: Civic Traditions in Modern Italy. Princeton, NJ: Princeton University Press.

- Putzel, J. (1997). Policy arena: Accounting for the 'dark side' of social capital: Reading Robert Putnam on democracy. *Journal of International Development*, 9(7), 939-949.
- Qian, Z. (1997). Progression to Second Birth in China: A Study of Four Rural Counties. *Population Studies*, *51*(2), 221-228.
- Rae, A. N., & Zhang, X. (2009). China's booming livestock industry: household income, specialization, and exit. *Agricultural Economics*, 40(6), 603-616.
- Rakodi, C. (1999). A capital Assets Framework for Analysing Household Livelihood Strategies: Implications for Policy. *Development Policy Review*, 17(3), 315-342.
- Ramankutty, N., Foley, J. A., & Olejnjczak, N. J. (2002). People on the land: changes in global population and croplands during the 20th Century. *Ambio*, *31*(3), 251-257.
- Rankin, K. N. (2002). Social capital, microfinance and the politics of development. *Feminist Economics*, 8(1), 1-24.
- Raskin, P. (2000). Bending the curve: Towards global sustainability. *Development*, 43(4), 67-74.
- Rebick, M. E. (2000). The Importance of Networks in the Market for University Graduates in Japan: A Longitudinal Analysis of Hiring Patterns. *Oxford Economic Papers*, 52(3), 471-496.
- Reichel, A., Lowengart, O., & Milman, A. (2000). Rural tourism in Israel: service quality and orientation. *Tourism Management*, 21, 451-459.
- Reinke, J. (1998). How to lend like mad and make a profit. *Journal of Development Studies*, *34*(3), 44-58.
- Richards, G. (1972). Tourism and Economy: an Examination of Methods for Evaluating the Contribution and Effects of Tourism in the Economy. Summary Report based on a PhD. Thesis, University of Surrey, Guildford, UK.
- Richards, J. (2006). "Precious" metals: The case for treating metals as irreplaceable. *Journal of Cleaner Production, 14*(3), 324-333.
- Richards, J.F. (1990). Land transformation. In B. L. Turner II, et al. (Eds.), *The Earth as Transformed by Human Action* (pp.163-178). New York, NY: Cambridge University Press.
- Robison, L. J., Schmid, A. A., & Siles, M. E. (2002). Is social capital really capital? *Review of Social Economy*, *60*(1), 1-21.
- Rohorua, H. T. S. (2009) *Economic Issues in the Fisheries Sector of Tonga*. Unpublished doctoral thesis. University of Waikato, Hamilton, New Zealand.
- Romer, P. (1986). Increasing returns and long run growth. *Journal of Political Economy*, *94*(5), 1002-1037.
- Romo, F. P., & Schwartz, M. (1995). Structural Embeddedness of Business Decisions: the migration of manufacturing plants in New York State, 1960 to 1985. American Sociological Review, 60(6), 874-907.

- Rozelle, S., Taylor, E.J., & de Brauw. A. (1999). Migration, Remittances and Agricultural Productivity in China. *American Economic Review Papers and Proceedings*, 89(2), 287-291.
- Ruan, D. (1998). The Content of the General Social Survey Discussion Networks: An Exploration of General Social Survey Discussion Name Generator in a Chinese Context. *Social Networks*, 20(3), 247-264.
- Ruffini, M. (2006). The Revival of Zambia's copper mining sector-rising copper prices renew interest in the Copper belt. *Engineering and Mining Journal*, 207(2), 39-43.
- Rufino, M. C., Tittonell, P., van Wijk, M. T., Castellanos-Navarrete, A., Delve, R. J., de Ridder, N., & Giller, K. E. (2007). Manure as a key resource within smallholder farming systems: analysing farm-scale nutrient cycling efficiencies with the NUANCES framework. *Livestock Science*, 112, 273-287.
- Ruhanen, L. (2009). *The tourism labour market in the Asia Pacific region*. Fifth UNWTO International Conference on Tourism Statistics. Tourism: an engine for employment creation. Bali, Indonesia, 30 March-2 April 2009.
- Rupasingha, A., Goetz, S. J., & Freshwater, D. (2000). Social capital and Economic Growth: A County-Level Analysis. *Journal of Agricultural and Applied Economics*, *32*(3), 565-572.
- Rus, A. (1995). Access and Mobilization- Dual Character of Social Capital: Managerial Networks and Privatization in Eastern Europe. Unpublished manuscript, Columbia University, New York.
- Ryan, C., Gu, H., & Fang, M. (2009a). Community participation and social impacts of tourism. In C. Ryan & H. Gu (Eds.), *Tourism in China: Destination, cultures and communities* (pp. 239-258). New York, NY: Routledge.
- Ryan, C., Gu, H., & Fang, W. (2009b). The context of Chinese tourism: an overview and implications for research. In C. Ryan & H. Gu (Eds.), *Tourism in China: Destination, cultures and communities* (pp. 327-336). New York, NY: Routledge.
- Saeed, K. (1994). *Development Planning and Policy Design: A system dynamics approach*. Aldershot, UK: Avebury.
- Samuels, W. J. (Ed.). (1992). *Essays on the Economic Role of Government*, vol. 2. New York, NY: New York University Press.
- Sanders, J., Nee, V., & Sernau, S. (2002). Asian immigrants' reliance on social ties in a multiethnic labour market. *Social Forces*, *81*(1), 281-314.
- Schuller, T., Baron, S., & Field, J. (2000). Social capital: A review and critique. In S. Baron, J. Field & T. Schuller (Eds.), *Social Capital: Critical perspectives* (pp. 1-38). Oxford, UK: Oxford University Press.
- Schwaninger, M., & Grosser, S. (2009). System Dynamics Modeling: Validation for Quality Assurance. In R. A. Meyers (Ed.), *Encyclopedia of Complexity and Systems Science*. New York, NY: Springer.
- Scoones, I. (1998). *Sustainable Rural Livelihoods: A Framework for Analysis*. IDS Working Paper No. 72. Brighton, UK: Institute of Development Studies.

Scott, C. (1992). Estimation of Annual Expenditure from One-month Crosssectional Data in a Household Survey, *Inter-Stat Bulletin*, *8*, 57-65.

Scott, J. (1991). Social Network Analysis: A Handbook. London, UK: Sage.

Shandong Tourism Administration Information (STAI). (2015). "Happy farmers in Spring Festival" in Tianhuang, Zoucheng. DaZhongNews, Retrieved from http://tour.dzwww.com/lvnews/201501/t20150115_11734171.htm

Shandong Urban and Rural Planning Design Institute (SURPDI). (2012). *The economic developmental strategy and the overall plan for Zoucheng city*, 2011-2030. Jinan: SURPDI. Retrieved from

http://www.sdghy.com/index.php?c=content&a=show&id=842

Shao, Q. W. (2007). Developing rural tourism and promoting the construction of new countryside. *Qiushi Magazine*, *1*, 42-44.

- Sharpley, R. (2003). Rural tourism and sustainability a critique. In D. Hall, L. Roberts & M. Mitchell (Eds.), *New directions in rural tourism* (pp. 38-53). Aldershot, UK: Ashgate.
- Sharpley, R., & Vass, A. (2006). Tourism, farming and diversification: an attitudinal study. *Tourism Management*, 27(5), 1040-1052.
- Sheldon, P. J. (1990). A Review of Tourism Expenditure Research, In C.P. Cooper (Ed.), *Progress in Tourism, Recreation and Hospitality Management*, Volume Two, London, UK: Belhaven.
- Shen, J. (2002). A study of the temporary population in Chinese cities. *Habitat International*, *26*(3), 363-377.
- Shen, J. (2007). Population growth, ecological degradation and construction in the western region of China. *Journal of Contemporary China*, *13*(41), 637-661.
- Shen, M., Rozelle, S., Zhang, L., & Huang, J. (2005). Farmer's professional associations in rural China: State dominated or new state-society partnerships?
 In B.H. Sonntag., J. Huang., S. Rozelle, & J. H. Skerritt (Eds.), *China's agricultural and rural development in the early 21st century* (pp. 197-228).
 Canberra, ACT: Australian Centre for International Agricultural Research.
- Shi, X. M. (2015). Factors influencing the environmental satisfaction of local residents in the coal mining area, China. Social Indicators Research, 120, 67-77.
- Shi, X. M., & He, F. (2012). Th Environmental pollution perception of residents in coal mining areas: A case study in the Hancheng Mine Area, Shaanxi Province, China. *Environmental Management*, 50(4), 505-513.
- Shi, X. M., & Liao, W. G. (2012). Residents' perception differences on environmental problems in coal mining areas: A case study of Tongchuan Mine Area, Shaanxi Province. *Scientia Geographica Sinica*, 32(9), 1087-1092.
- Siegel, P. (2005). Using an asset-based approach to identify drivers of sustainable rural growth and poverty reduction in Central America: A conceptual framework. World Bank Policy Research Working Paper 3475. Washington, DC: World Bank.

- Sierka, E., Molenda, T., & Chmura, A. (2009). Environmental repercussions of subsidence reservoirs reclamation. *Journal of Water and Land Development*, 13a (1), 1-41.
- Simith, F. P., Holzworth, D. P., & Robertson, M. J. (2005). Linking icon-based models to code-based models: a case study with the agricultural production systems simulator. *Agricultural Systems*, *83*, 135-151.
- Simmons, L. C., & Much, J. M. (1996). Is Relationship Marketing Culturally Bound: A Look at Guanxi in China. Advances in Consumer Research, 23, 92-96.
- Sinclair, M. T. (1998). Tourism and economic development: a survey. *Journal of Development Studies*, *34*(5), 1-51.
- Smith, D. R., Gordon, A., Meadows, K., & Zwick, K. (2001). Livelihood diversification in Uganda: Patterns and determinants of change across two rural districts. *Food Policy*, 26(4), 421-435.
- Snijders, T. A. B. (1999). Prologue to the measurement of social capital. *La Revue Tocqueville 20*(1), 27-44.
- Sobel, J. (March 2002). Can We Trust Social Capital? *Journal of Economic Literature*, 40(1), 139-154.
- Socher, K., & Tschurtschenthaler, P. (1994). Tourism and Agriculture in Alpine Regions. *Tourist Review*, 50, 35-41.
- Solinger, D. (1999). *Contesting Citizenship in Urban China: Peasant Migrants, the State and the Logic of the Market*. Oakland, CA: University of California Press.
- Solow, R. (1956). A contribution to the theory of economic growth. *Quarterly Journal of Economics*, 70(1), 65-94.
- Song, H. & Li, G. (2008). Tourism Demand Modeling and Forecasting. *Tourism Management*, 29(2), 203-220.
- Soriano, C. R. R. (2007). Exploring the ICT and Rural Poverty Reduction Link: Community Telecenters and Rural Livelihoods in Wu'an, China. *Electronic Journal of Information Systems in Developing Countries, 32*(1), 1-15.
- Sovacool, B. K. (2015). Countering a corrupt oil boom: Energy justice, Natural Resource Funds, and Sao Tome e Principe's Oil Revenue Management Law. *Environmental Science & Policy*, *55*, 196-207.
- State Statistical Bureau (1997). *Statistical Yearbook of China*. Bejing: China Statistical Publishing House.
- Stave, K. A. (2003). A system dynamics model to facilitate public understanding of water management options in Las Vegas, Nevada. *Journal of Environmental Management* 67(4), 303-313.
- Sterman, J. D. (2000). *Business Dynamics: Systems Thinking and Modeling for a Complex World*. New York, NY: McGraw-Hill.
- Stiglitz, J. E. (1999). Formal and informal institutions. In P. Dasgupta & I.Serageldin (Eds.), *Social Capital: A multifaceted perspective* (59-68. Washington DC: World Bank.
- Stokes, C. S. (2002). *Measuring impacts of HIV/AIDS on rural livelihoods and food security*. Rome, Italy: Gender and Population Division, FAO.

Stronza, A. L. (2000). *Because it is ours: Community-based ecotourism in the Peruvian Amazon.* PhD Thesis. University of Florida, Gainsville, FL.

- Su, B. (2011). Rural Tourism in China. Tourism Management, 32(6), 1438-1441.
- Su, M., & Wall, G. (2014). Community Participation in Tourism at a World Heritage Site: Mutianyu Great Wall, Beijing, China. *International Journal of Tourism Research*, 16, 146-156.
- Su, X. L. (2011). Study on the development of the Yamen culture tourism. *Economic Geography*, 2011-01.
- Sun, J. L., & Du, B. S. (2010). Modern agriculture brings more benefits to farmers in Zoucheng. *Jining Daily News*. Retrieved from http://www.mencius.gov.cn/news01/system/21975.html
- Sun, J. L., & Shao, M. K. (2010, January, 04). The development of rural land transferring in Zoucheng. *Jining Daily News*. Retrieved from http://www.sd.xinhuanet.com/wq/2010-01/04/content_18668068.htm
- Sun, J. L., Niu, Y. N., & Du, B. S. (2009). The agricultural output in experimental plots in Zoucheng, *Jining Daily News*. Retrieved from http://www.mencius.gov.cn/news01/system/21828.html
- Sun, X. (2014). Kanzhuang villagetown: Agricultural products brand is important. *Zoucheng Information*. Retrieved from http://www.mencius.gov.cn/news01/04/2014-06-12/36079.html
- Sun, Z. G. (2013). Modernization leads agricultural development. Zoucheng Government Information. Retrieved from http://zczww.xinzc.cn/index.php/cms/item-view-id-1305.shtml
- Swan, T. (1956). Economic growth and capital accumulation. *Economic Record*, *32*(2), 334-361.
- Tabachnick, B.G., & Fidell, L.S. (1989). *Using multivariate statistics* (2nd ed.). Cambridge, MA: Harper & Row.
- Tan, M. H., Li, X. B., Xie, H., & Lu, C. H. (2005). Urban land expansion and arable land loss in China: a case study of Bejing-Tianjin-Hebei region. *Land Use Policy*, 22, 187-196.
- Tan, S. Heerink, N., Kruseman, G., & Qu, F. (2008). Do fragmented landholdings have higher production costs? Evidence from rice farmers in Northeastern Jiangxi province, P.R. China. *China Economic Review*, 19(3), 347-358.
- Tang, S. (1998). Localization of tourism destination and community tourism research. *Geography Research*, *17*(2), 145-149.
- Tao, Z., & Li, M. (2007). What is the limit of Chinese coal supplies A STELLA model of Hubbert Peak. *Energy Policy*, *35*(6), 3145-3154.
- Taylor, B. E., Morison, J. B., & Fleming, E. M. (1991). The Economic Impact of Food Import Substitution in the Bahamas. *Social and Economic Studies*, 40(2), 45-62.
- Taylor, G. (1995). The community approach: does it really work? *Tourism Management*, *16*(7), 487-489.
- Taylor, J., & Dalziel, P. (2002). *Macroeconomics: New Zealand Edition*. Auckland, New Zealand: Wiley.

- Taylor, T., Ford, D., & Ford, A. (2010). Improving model understanding using statistical screening, *System Dynamics Review 26*(1), 73-87.
- TED. (2011, Jan). Martin Jacques: Understanding the rise of China. *Speaking at a TED Salon in London*. London, UK: TED. Retrieved from https://www.ted.com/talks/martin_jacques_understanding_the_rise_of_china?la nguage=en
- The conciliation commission of regional economies. (2006, Dec). *The report of regional economic research*. (S.Y. Liu, Trans.) ZouCheng City.
- Thirion, A. P., Macias, E.F., Hurley, J., & Vermeylen, G. (2007). *Fourth European Working Conditions Survey*. Dublin, Ireland: European Foundation for the Improvement of Living and Working Conditions.
- Thompson, S. K. (2002). Sampling (2nd edition). New York, NY: Wiley.
- Thompson, S. K. (2006). Adaptive web sampling. Biometrics, 62(4), 1224-1234.
- Thompson, S. K., & Frank, O. (2000). Model-based estimation with link-tracing sampling designs. *Survey Methodology*, *26*(1), 87-98.
- Timothy, D. J., & Tosun, C. (2003). Appropriate planning for tourism in destination communities: participation, incremental growth and collaboration. In S. Singh, D. J. Timothy, & R. K. Dowling (Eds.), *Tourism in Destination Communities* (pp. 181-204). Wallingford, UK: CABI Publishing.
- Tittonell, P., Zingore, S., van Wijk, M.T., Corbeels, M., Giller, K.E. (2007). Nutrient use efficiencies and crop responses to N.P and manure applications in Zimbabwean Soils: exploring management strategies across soil fertility gradients. *Field Crops Research*, 100(2-3), 348-368.
- Todaro, M. P. (1994). *Economic Development (5th Ed)*. New York, *NY*: Longman Publishing.
- Tosun, C. (2000). Limits to Community Participation in the Tourism Development Process in Developing Countries. *Tourism Management*, *21*(6), 613-633.
- Tout, H. (1938). The Standard of Living in Bristol. Bristol UK: Arrowsmith.
- Trigilia, C. (2001). Social capital and local development. *European Journal of Social Theory*, *4*(4), 437-442.
- Tsang, M. C., & Xiao, J. (1999). Human capital development in an emerging economy: The experience of Shenzhen, China. *China Quarterly*, *157* (March), 72-114.
- Tsang, W. K. (1998). Can Guanxi Be a Source of Sustained Competitive Advantages for Doing Business in China? *The Academy of Management Executive*, 12(2), 64-73.
- Tshimpanga, J. M. (2012). A Regime Shift Analysis of Poverty Traps in sub-Saharan Africa: Identifying Key Feedbacks and Leverage Points for Change. Master's Thesis, Uppsala University, Sweden.
- Tsui, A., Farh, L., Xin, K. (2000). Guanxi in the Chinese context. In J. T. Li, A. Tsui, & E. Weldon (Eds.), *Management and Organization in the Chinese Context* (pp. 225-244). London, UK; Macmillan.

- Turner, B. L. II (1989). The human causes of global environmental change. In R. S. Defires & T. Malone (Eds.), *Global Change and Our Common Future: Papers from a forum* (pp. 90-99). Washington, DC: National Academy Press.
- Tyrakowski, K. (1986). The Role of Tourism in Land Utilization Conflicts on the Spanish Mediterranean Coast. *GeoJournal*, *13*(1), 19-26.
- United Nations World Tourism Organization (UNWTO) (2004). *Tourism and poverty alleviation: recommendations for action.* Madrid, Spain: UNWTO.
- Uphoff, N. (1999). Understanding social capital: Learning from the analysis and experience of participation. In P. Dasgupta & I. Serageldin (Eds.), *Social capital: A multifaceted perspective* (pp. 215-249). Washington, DC: World Bank.
- Ursache, M. (2014). Sustainable development between recent experiences and future challenges. *Procedia Economics and Finance*, *15*, 1316-1323.
- Uzzi, B. (1997). Social Structure and Competition in Interfirm Networks: The Paradox of Embeddedness. *Administrative Science Quarterly*, 42(1), 35-67.
- Uzzi, B. (1999). Embeddedness in the Making of Financial Capital: How Social Relations and Networks Benefit Firms Seeking Financing. *American Sociological Review*, 64(4), 481-505.
- Van den Bergh, J. C. J. M. (1991). *Dynamic Models for Sustainable Development*. Amsterdam, Netherlands: Thesis Publishers.
- Van Der Gaag, M. P. J., & Snijders, T. A. B. (2005). The Resource Generator: measurement of individual social capital with concrete items. *Social Networks* 27, 1-29.
- Van Ittersum, M. K., Ewett, F., Heckelei, T., Wery, J., Olsson, J. A., Anderson, E., ... Wolf, J. (2008). Integrated assessment of agricultural systems-A component-based framework for the European Union (SEAMLESS). *Agricultural systems*, 96(1), 150-165.
- Van Wijk, M. T., Tittonell, P., Rufino, M. C., Herrero, M., Pacini, C., de Ridder, N., Giller, K. E. (2009). Identifying key entry-points for strategic management of smallholder farming systems in sub-Saharan Africa using the dynamic farmscale simulation model NUANCES-FARMSIM. *Agricultural Systems*, 102(1-3), 89-101.
- Vargas, S., Anchez, A., Plaza-Mejia, M., & Porras-Bueno, N. (2009). Understanding Resident's Attitudes toward the Development of Industrial Tourism in a Former Mining Community. *Journal of Travel Research*, 47(3), 373-387.
- Vennix, J. (2001). *Group Model Building, Faciliting Team Learning using System Dynamics*. Chichester, UK: Wiley.
- Ventana Systems, Inc. (2006). *User's Guide for Vensim*. Harvard, MA: Ventana Systems, Inc.
- Ventana Systems, Inc. (2010). *Vensim. PLE Plus for Windows Version 5.11 A.* Harvard, MA: Ventana Systems, Inc.

- Vlachos, D., Georgiadis, P., Iakovou, E. (2007). A system dynamics model for dynamic capacity planning of remanufacturing in closed-loop supply chains. *Computers & Operations Research*, 34(2), 367-394.
- Wack, P. (1985). Scenarios: shooting the rapids. *Harvard Business Review*, (November-December), 139-150.
- Walker, P., Lee, D., Goddard, R., Kelly, G., & Pedersen, J. (2005). *Regional tourism modelling: the South West Tapestry*. Southport, Australia: Cooperative Research Centre for Sustainable Tourism.
- Wampold, B.E., & Freund, R.D. (1987). Use of multiple regression in counselling psychology research: A flexible data-analytic strategy. *Journal of Counseling Psychology*. 34, 372-382.
- Wan, G. H. & Cheng, E. J. (2001). Effects of land fragmentation and returns to scale in the Chinese farming sector. *Applied Economics*, *33*(2), 183-194.
- Wang, H., Sindelar, J. L., & Busch, S. H. (2005). The impact of tobacco expenditure on household consumption patterns in rural China. *Social Science & Medicine*, 62(6), 1414-1426.
- Wang, J., Brown, D. G., & Agrawal, A. (2013). Climate adaptation, local institutions, and rural livelihoods: A comparative study of herder communities in Mongolia and Inner Mongolia, China. *Global Environmental Change*, 23(6), 1673-1683.
- Wang, J., Liu, H., & Peng, H. (2008). System dynamics model of urban transportation system and its application. *Journal of Transportation Systems Engineering and Information Technology*, 8(3), 83-89.
- Wang, L. (2013). The new rural economy in Kanzhuang, Zoucheng. Zoucheng Information. Retrieved from http://www.mencius.gov.cn/news01/04/2013-12-16/32904.html
- Wang, L.E, Cheng, S.K, Zhong, L.S., Mu, S.L., Dhruba, B.G., & Ren, G.Z. (2013). Rural Tourism Development in China: Principles, Models and the Future. *J.Mt. Sci. 10*, 116-129.
- Wang, M., Ye, Y., Qian, H., Song, Z., Zhang, Z., Zhou, J., & Ni, C. (2010). Common genetic variants in pre-micro RNAs are associated with risk of coal workers pneumoconiosis. *Journal of Human Genetics*, 55(1), 13-17.
- Wang, X., Huang, J., Zhang, L., & Rozelle, S. (2011). The rise of migration and the fall of self employment in rural China's labour market. *China Economic Review*, 22(4), 573-584.
- Wang, Y., & Xu, J. F. (2012). The prospect of agricultural development in Kanzhuang village town, Zoucheng. *Zoucheng Information Network*. Retrieved from http://www.mencius.gov.cn/news01/04/9852.html
- Wang, Y., & Yao, Y. (2003). Sources of China's economic growth 1952-1999: incorporating human capital accumulation. *China Economic Review*, 14(1), 32-52.
- Wang, Y., Zhang, D., Lian, D., Li, Y., & Wang, X. (2009). Environment cumulative effects of coal exploitation and its assessment. *Procedia Earth and Planetary Science 1*, 1072-1080.

- Wang, Y.M., & Tang, Y.Y. (2013). The quality featured agriculture in Zhongxin village town, Zoucheng city. *Jining Daily*. Retrieved from http://www.mencius.gov.cn/news01/system/31922.html
- WCED (1987). Report of the World Commission on Environment and Development: Our Common Future (Annex to Document A/42/427, 1987). Oxford, UK: Oxford University Press.
- Wei, X., Qu, H., & Ma, E. (2013). Modelling tourism employment in China. *Tourism Economics*, 19(5), 1123-1138.
- Wellman, B., & Wortley, S. (1990). Different strokes from different folks: Community ties and social support. *American Journal of Sociology*, 96(3), 558-588.
- Wellman, B., Chen, W., & Dong, W. (2002). Networking guanxi. In T. Gold, D. Guthrie, & D. Wank (Eds.), Social Connections in China: Institutions, Culture, and the Changing Nature of Guanxi (pp. 221-241). New York, NY: Cambridge University Press.
- Wen, Z. (1997). China's domestic tourism: impetus, development and trends. *Tourism Management*, 18(8), 565-571.
- Weng, Q. H. (2002). Land use change analysis in the Zhujiang Delta of China using satellite remote sensing, GID and stochastic modeling. *Journal of Environmental Management*, 64 (3), 273-284.
- Whiteley, P. F. (2002). Economic growth and social capital. *Political Studies*, 48(3), 443-466.
- Wilson, S., Fesenmier, D. R., Fesenmaier, J., & Van Es, J. C. (2001). Factors for Success in Rural Tourism Development. *Journal of Travel Research*, 40(2), 132-138.
- Winkels, A., & Adger, N. W. N. (2002). Sustainable livelihoods and migration in Vietnam: The importance of social capital as access to resources. Paper presented at the International Symposium, "Sustaining food security and managing natural resources in Southeast Asia – Challenges for the 21stcentury", January 8-11, 2002, Chiang Mai, Thailand. Retrieved from https://www.uni-

hohenheim.de/fileadmin/einrichtungen/sfb564/events/uplands2002/Full-Pap-S2-2_Winkels.pdf

- Wiranatha, A. & Smith, P. (2000). A conceptual framework for a dynamic model for regional planning: towards sustainable development for Bali, Indonesia. Paper presented at the 1st International Conference on Systems Thinking in Management, ICSTM2000. Retrieved from https://www.research.esta.pst/publication/221502175. A Conceptual Eremework.
 - https://www.researchgate.net/publication/221503175_A_Conceptual_Framework _for_a_Dynamic_Model_for_Regional_Planning_Towards_Sustainable_Develo pment_for_Bali_Indonesia
- Woldenhanna, T., & Oskam, A. (2001). Income diversification and entry barriers: Evidence from the Tigray Region of Northern Ethiopia. *Food Policy*, 26(4), 351-365.

Wolstenholme, E. (1997). System dynamics in the elevator (SD1163), e-mail communication, system-dynamics@world.std.com

Wong, O., & Lai, G. (2002). The tie effect on information dissemination: The spread of a commercial rumor in Hong Kong. *Social Networks*, 24(1), 49-75.

Woolcock, M. (1998). Social capital and economic development: Toward a theoretical synthesis and policy framework. *Theory and Society*, 27(2), 151-208.

Woolcock, M., & Narayan, D. (2000). Social capital: Implications for development theory, research, and policy. *World Bank Research Observer*, *15*(2), 225-249.

World Bank (1995; 1997). *China Living Standards Survey (CLSS)*, 1995-1997.
Washington DC: Research Centre for Rural Economy/ World Bank. Retrieved from http://siteresources.worldbank.org/INTLSMS/Resources/3358986-1181743055198/3877319-1190295368393/chnbinfo.pdf

World Bank (2003). A manual for planning and implementing the living standards measurement study survey. Washington DC: World Bank. Retrieved from http://documents.worldbank.org/curated/en/1996/05/438573/manual-planningimplementing-living-standards-measurement-study-survey

World Bank Group (1999). Pollution Prevention and Abatement Handbook 1998: Toward Cleaner Production. Washington DC: The International Bank for Reconstruction and Development/The World Bank. Retrieved from http://www.ifc.org/wps/wcm/connect/0f8fe18048855d738f04df6a6515bb18/PP AH.pdf?MOD=AJPERES&CACHEID=0f8fe18048855d738f04df6a6515bb18

World Bank. (1993). *The East Asian Miracle: economic growth and public policy*. New York, NY: Oxford University Press.

World Bank. (1998). The initiative on defining, monitoring and measuring social capital. Social Capital Initiative working paper no.1. Retrieved from /http://siteresources.worldbank.org/INTSOCIALCAPITAL/Resources/Social-Capital-Initiative-Working-Paper-Series/SCI-WPS-01.pdf

World Bank. (2007). Rethinking China's Poverty Reduction Agenda: An Assessment of Poverty and Inequality in China. Washington, DC: World Bank.

World Bank. (2010). *China: Shandong Culture Heritage Sector Review*. Washington, DC: World Bank.

World Coal Institute (2002). Coal and sustainable development. In *Proceedings of the World Summit on Sustainable Development and Its Implications*. New York: Energy/Ge.1.

Wu, S., Walker, D., Devadoss, S., & Lu, Y. (2002). Productivity growth and its components in Chinese agriculture after reforms. *Review of Development Economics*, 5(3), 375-391.

Wu, W., & Choi, W.L. (2004). Transaction cost, social capital and firms; synergy creation in Chinese business network: An integrative approach. Asia Pacific Journal of Management, 21(3), 325-343.

Wu, X., & Meng, X. (1997). The direct impact of the relocation of farm labour on Chinese grain production. *China Economic Review*, 7(2), 105-122.

Wu, Z., Liu, M., & Davis, J. (2005). Land consolidation and productivity in Chinese household crop production. *China Economic Review*, 16, 28-49.

- Wunder, S., Bomer, S., Shively, G., & Wyman, M. (2014). Safety nets, gap filling and forests: a global-comparative perspective. *World Development*, *64*, 29-42.
- Xiao, J. (2001). Determinants of salary growth in Shenzhen, China: an analysis of formal education, on-the-job training, and adult education with a three-level model. *Economics of Education Review*, 21(6), 557-577.
- Xiao, Z., & Tsui, A. (2007). When brokers may not work: The cultural contingency of social capital in Chinese high-tech firms. *Administrative Science Quarterly*, 52(1), 1-31.
- Xie, Y., Yu, M., Tian, G., and Xing, X. (2005). Socio-economic driving forces of arable land conversion: A case study of Wuxian City, China. *Global Environmental Change*, 15(3), 238-252.
- Xinhua. (2004) China urges further protection of arable land. *China Daily News*. Retrieved from http://www.chinadaily.com.cn/english/doc/2004-03/23/content_317346.htm
- Xu, J., Yao, L., & Mo, L. (2011). Simulation of low-carbon tourism in world natural and cultural heritage areas: An application to Shizhong District of Leshan City in China, *Energy Policy 39*(7), 4298-4307.
- Xu, X., Chen, C., Qi, H., He, R., You, C., & Xiang, G. (2000). Development of coal combustion pollution control for SO₂ and NO_x in China. *Fuel Processing Technology*, 62, 153-160.
- Xu, X., Zhao, Y., Hu, Z., Yu, Y., & Shao, F. (2014). Boundary demarcation of the damaged cultivated land caused by coal mining subsidence. *Bulletin of Engineering Geology and the Environment*, 73(2), 621-633.
- Xu, Z., & Zhu, N. (2009). City Size Distribution in China: Are Large Cities Dominant? *Urban Studies*, *46*(10), 2159-2185.
- Xue, Z., Zhang, J., & Wall, G. (2009). The impact of residents' sense of place on their support for the development of regional tourism, *Dili xueban (Acta Geographica Sinica)*, 64, 736-744.
- Yan, Y. (1996). *The flow of Gifts: Reciprocity and Social Networks in a Chinese Village*, Palo Alto, CA: Stanford University Press.
- Yang, D.T. (2005). Determinants of Schooling Returns during Transition: Evidence from Chinese Cities. *Journal of Comparative Economics*, *33*(2), 244-264.
- Yang, D., & Wang, M. (2006). Discussions on potential bias and implications of Lewis Turning Point. *China Economic Journal*, 3(2), 121-136.
- Yang, L., Hu, Z., Li, X., Li, J., & Zhao, Y. (2007). Analysis of land use change and ecosystem service value in the overlapped areas of crop and mineral production in Zoucheng City. *Transactions of the CSAE*, 23(12), 70-75.
- Yang, M. (1994). *Gifts, Favors and Banquets: the Art of Social Relationships in China*. Ithaca, NY: Cornell University Press.
- Yang, W. (2009). Economic Structural Changes and Rural income: Evidence from Chinese provinces during the reform period. *China Economic Review*, 20(4), 742-753.

- Yang, Z., Cai, J., & Sliuzas, R. (2010). Agro-tourism enterprises as a form of multifunctional urban agriculture for peri-urban development in China. *Habitat International*, 34, 374-385.
- YanZhou Mining Group (Dec, 2007 & 2014). Annual working conferences of Yanzhou Ming Group.
- Ye, Z. H., Wong, J. W. C., Wong, M. H., Baker, A. J. M., Shu, W. S. & Land, C. Y. (2010). Revegetation of Pb/Zn Mine Tailings, Guangdong Province, China. *Restoration Ecology*, 8(1), 87-92.
- Yeh, A. G. & Li, X. (1999). Economic development and agricultural land loss in the Pearl River Delta, China. *Habitat International*, 23(3), 373-390.
- Yen, D. A., Barnes, B. R., & Wang, C. L. (2011). The measurement of guanxi: Introducing the GRX scale. *Industrial Marketing Management*, 40(1), 97-108.
- Yeung, I. Y. M., & Tung, R. L. (1996). Achieving business success in Confucian societies: the importance of guanxi. *Organisational Dynamics*, 25, 2 54-65.
- Ying, T., & Zhou, Y. (2007). Community, governments and external capitals in China's rural cultural tourism: a comparative study of two adjacent villages. *Tourism Management*, 28(1), 96-107.
- Yip, W., Subramanian, S. V., Mitchell, A. D., Lee, D. T. S., Wang, J., & Kawachi, I. (2007). Does social capital enhance health and well-being? Evidence from rural China. *Social Science & Medicine*, 64(1), 35-49.
- Yue, L., & Knight, J. (2008). The role of social capital in the labour market in China, *Economics of Transition*, *16*(3), 389-414.
- Yzer, M. C. (2008). The Integrative Model of Behavioral Prediction and messagebased HIV prevention. In P.H. Swanepoel & H. Hoeken (Eds.), Adapting health communication to cultural needs: Optimizing documents in South African health communication on HIV/AIDS (pp. 49-69). Amsterdam, Netherlands: Benjamins.
- Zeng, B. (2008). Tourism development and local poverty: A case study of Qinling Mountain Region, Shaanxi Province, China. Saarbrucken, Germany: VDM Publishing.
- Zeng, B., & Ryan, C. (2012). Assisting the poor in China through tourism development: A review of research. *Tourism Management*, *33*(2), 239-248.
- Zhang, H., & Zhang, X. (2003). A basic study on the tourism image in Turpan. *Journal of Xinjiang Normal University*, 22(3), 86-91.
- Zhang, J. (2003). Urban Xiagang, Unemployment and Social Support Policies: A Literature Review of Labor Market Policies in Transitional China. Report to the World Bank. Retrieved from http://info.worldbank.org/etools/docs/library/152357/(14)%20ZhangJuwei_Socia
- 1%20Support%20systems_%20summary.pdf
 Zhang, J., Zhao, Y., Park, A., & Song, X. (2005). Economic Returns to Schooling in Urban China, 1988 to 2001. *Journal of Comparative Economics*, 33(4), 730-750.
- Zhang, L. W. (2013). The benefits of strawberry planting in Zhongxin, Zoucheng. KMZX, Retrieved Nov 19, 2014 from http://www.kmzx.org/lvyou/ShowArticle.asp?ArticleID=22602

- Zhang, L., Wang, H., Wang, L., & Hsiao, W. (2006). Social capital and farmer's willingness-to-join a newly established community-based health insurance in rural China. *Health Policy*, 76(2), 233-242.
- Zhang, Q. H., Chong, K., & Ap, J. (1999). An analysis of tourism policy development in modern China. *Tourism Management*, 20, 471-485.
- Zhang, X. (2012). Research on the Development Strategies of Rural Tourism in Suzhou Based on SWOT Analysis, *Energy Procedia 16*, 1295-1299.
- Zhang, Y., & Zhang, Z. (2006). Guanxi and organizational dynamics in China: a link between individual and organizational levels. *Journal of Business Ethics*, 67(4), 375-392.
- Zhang, Y., Li, C., Zhou, X., & Moore, B. (2002). A simulation model linking crop growth and soil biogeochemistry for sustainable agriculture. *Ecological Modeling*, 151, 75-108.
- Zhang, Z. Q. (2010). The development of ecotourism industry in Zhongxin, Zoucheng. Zoucheng Government News. Retrieved from http://www.mencius.gov.cn/news01/04/8290.html
- Zhao, J., & Barry, P.J. (2014). Income Diversification of Rural Households in China. *Canadian Journal of Agricultural Economics* 62, 307-324.
- Zhao, W. (2009). The nature and roles of small tourism businesses in poverty alleviation: Evidence from Guangxi, China. *Asia Pacific Journal of Tourism Research*, 14(2), 169-182.
- Zhao, Y. (2002). Measuring the social capital of laid-off Chinese workers. *Current Sociology*, *50*(4), 555-571.
- Zhao, Y. (July, 1999). Labor migration and earnings differences: the case of rural China. *Economic Development and Social Change*, 47(4), 767-782.
- Zheng, Q. M., & Zhong, L. S. (2004). A discussion of developing model of community-involved rural tourism. *Tourism Tribune*, *19*(4), 33-37.
- Zhou, L., & Liu, K. (2008). Community tourism as practiced in the mountainous Qiang region of Sichuan Province, China - a case study in Zheghe village. *Journal of Mountain Science* 5, 140-156.
- Zhou, Z. (2003). China's experience with agricultural cooperatives in the era of economic reforms. In R.Trewin (Ed.) Cooperatives: Issues and trends in developing countries, Report of a workshop in Perth, Australia, March 24–25, 2003. Canberra, ACT: ACIAR.
- Zimmerman, C. C. (1932). Ernst Engel's law of expenditures. *The Quarterly Journal of Economics*, 47(1), 78-101.
- Zong, Y. (2016). The rural tourism development experiences in Zoucheng and Yinan. Jinxiang Cultural Relics and Travel Bureau. Retrieved from http://www.jita.gov.cn/zwwji/new.do?method=getNewsById&uri=nw14564644 54570
- Zou, T. Q., Huang, S., & Ding, P. (2014). Toward A Community-driven Development Model of Rural Tourism: the Chinese Experience. *International Journal of Tourism Research*, 16(3), 261-271.

Zoucheng. (2012). The localization of a village leader in Xiangcheng village, Zoucheng. *www.zoucheng.cc*. Retrieved from http://www.zoucheng.cc/news/local/1626958.html

- Zoucheng Government. (2012). The agricultural development situation in Zoucheng, 2010. Retrieved from
 - http://www.mencius.gov.cn/news02/05/nongyejingji/2012-11-14/17729.html

Zoucheng Government. (2015). The policies about developing rural society and villages' economy in Zoucheng. Retrieved from http://www.jn001.com/news/2015-11/30/content_149165.htm

- Zoucheng government. (March, 2008). The tourism development master plan for Zoucheng city. Beijing: Tourism planning and design institute.
- Zoucheng Statistical Yearbook: 2000-2013 (2001-2014, Aug). (S.Y. Liu, Trans.). Zoucheng: The Bureau of Statistics of Zoucheng City.
- Zoucheng Tourist Administration. (2011). *Number of tourists from 2001 to 2010*. Zoucheng: Zoucheng Tourist Administration.
- Zoucheng Tourist Administration. (2013). The cooperative society in Yishan village, Zoucheng. Retrieved from http://www.sdta.gov.cn/ggxx/ggxx-gskx/newInfo/fb017e8d3f82ed44013fa335b17700ce.html
- Zoucheng Tourist Administration. (2013). *The ownership of tourist attractions and tourism business in Zoucheng-2013*. Zoucheng: Zoucheng Tourist Administration.

Appendices

Appendix A: Simulation model evaluation and testing

While scenarios developed on the system dynamics model for this research were based on a set of parameter assumptions, some uncertainties were embedded throughout the iterative model development processes. Validation is the process by which the model establishes confidence in its usefulness (Schwaninger & Grosser, 2009). The process of validation can determine whether the model is an acceptably accurate representation of reality (Schwaninger & Grosser, 2009; Giannasi et al., 2001). However, there is no single test to assert the validation of a model. Forrester & Senge (1980) pointed out that the level of confidence in the model can increase gradually with ongoing evaluation testing. Therefore, the following discussion focuses on the set of model evaluation tests so as to get an impression of the uncertainties with respect to the issue of model validation.

A.1 Structure verification test

The most relevant structure in the sustainable livelihood framework (DFID, 1999) was considered as the basis of the whole model design. During the model building process, partial models were assessed for behaviour consistent with existing knowledge of the system before being incorporated into the whole model structure. For instance, the agricultural production model test demonstrates model behaviour consistent with existing knowledge and information about stock and flow dynamics in North China, and labour distribution was based on financial incomes relevant to the industries in the model. There was no contradiction of the knowledge about the structure of the real system.

A.2 Dimensional consistency

Dimensional consistency involves specifying the units of measure for each variable in the simulation model, as well as checking for dimensional errors. Vensim provides a tool to check the unit consistency of model equations. The model was found to have no significant dimensional inconsistencies.

A.3 Data used for assessment

Data used for system dynamics model were sourced from first-hand information (i.e., the household survey in Zoucheng city) and secondary data in Zoucheng city such as the 2006-2010 Statistical Yearbooks and tourist data collected by Zoucheng Tourist Administration. Thus, the patterns generated in the simulations are in tune with the patterns in real-life situations. In other words, the generated model behaviour is judged against historical behaviour patterns.

A.4 Boundary adequacy test

In line with the research purposes of this thesis and the general model structure, the simulation model considered all important variables related to the system such as all relevant variables in local three industries (agriculture, mining, and tourism), and relationships between livelihood assets (e.g. social capital, natural capital, and human capital).

A.5 Sensitivity test

To ensure the reliability of recommended policies, sensitivity analysis is necessary to provide insight on how uncertainties in the model outputs respond to the input variables and which input variables tend to affect behaviour changes in the outputs (Guan, Gao, Fukahori, & Watari, 2010; Powell, Fair, LeClaire, Moore, & Thompson, 2005; Moxnes, 2005). With sensitivity analysis of the model, it is easier to choose the methods such as optimization of state variables related to the parameters, direct measurement, literature etc. used to assign values to the parameters (Guan, et al., 2010; Powell, et al., 2005).

The Vensim software provides a tool to examine how the model behaves in response to changes in the values of particular parameters. The first step is to identify relevant input parameters and ranges (Hekimoglu & Barlas, 2010; Taylor et al., 2010) and enter these parameter ranges and distributions in Vensim's Sensitivity simulation module (Ford & Flynn (2005). However, it is not sufficient to use one-variate sensitivity analysis to understand the whole model (Hekimoglu & Barlas, 2010). Thus, the model used Monte Carlo sensitivity simulations (also known as multivariate sensitivity simulation) to perform multivariate sensitivity.

In the simulation model for this thesis, the parameters change of growth rate in Coal Production Subsystem and Agricultural Products Subsystem were used for the system sensitivity analysis. For instance, the parameters fluctuated within the range of historical behaviour patterns. Thus, the study selected these parameters and assigned maximum and minimum values along with a random distribution over which to vary them to examine their impact on model behaviour. The ranges of two parameters are listed as follows: (1). Intrinsic growth rate for coal output (RG_C): (-0.0089, 0.0129); (2). Intrinsic growth rate for agricultural output (RG_A): (-0.0077, 0.021). With the distribution for each parameter specified, the number of simulations was set at 200 to perform the multivariate test, which means this process was repeated 200 times. The probability distribution values for each parameter was drawn from a Random Uniform Distribution (i.e., any number between the maximum and minimum values is equally likely to occur).

Three typical output values (i.e., total coal production (C), agricultural output per household (A), and farmland quality per household (AF)) were summarized because of their important role in the simulation process. The results of the sensitivity test are shown in Figure A1. The shaded areas show the range of results that were found on varying the focal parameters. The sensitivity is expressed as confidence bounds of 50%, 75%, 95% and 100%, which are coloured on the chart. The graphs display the uncertainty in the relevant variables over a period of time. The test result reveals that

the model behaviour does not exhibit implausible changes as a number of key parameters vary.



Figure A.1: Sensitivity analysis on simulation model

Appendix B: LOOKUP



Figure B.1: Coal mining pollution on farmland quality (Input: Total coal output)

Figure B.2: Total coal output and industrial dust discharged (Input: Total coal output)



Appendix C: Summarized data for simulation model

Year	Coal production (Ton)	Employment (Miner)		
2000	28,817,091	-		
2001	37,860,460	99,863		
2002	42,952,692	101,893		
2003	48,542,663	102,024		
2004	44,531,537	99,595		
2005	39,727,014	99,618		
2006	40,520,789	101,808		
2007	37,604,524	102,919		
2008	39,183,914	82,754		
2009	37,160,832	84,584		
2010	41,220,639	95,809		
Average	39,829,287*	97,087		

Table C.1: Coal production and employment

Source: Author's compilation from Zoucheng Statistical Yearbook, 2000-2010

Table C.2: Grain yield (2004-2013)

,	
Year	<i>Grain yield</i> (Kilogram/Mu)
2004	460.71
2005	458.18
2006	447.41
2007	444.88
2008	462.47
2009	459
2010	465
2011	472.8
2012	484.8
2013	490.9

Source: Author's compilation from local Yearbook 2004-2013

Year	Meng Temple	Yi Mountain	Total Arrivals
2003	23,399	99,263	122,662
2004	37,765	79,058	116,823
2005	39,060	107,379	146,439
2006	42,086	92,223	134,309
2007	34,223	54,355	88,578
2008	33,078	72,137	105,215
2009	26,926	138,725	165,651
2010	65,231	132,786	198,017

Table C.3: Number of tourists in Meng Temple and Yi Mountain from 2003 to 2010

Source: Author's compilation from Zoucheng Tourist Administration, 2011

Appendix D: Household survey

SECTION A: GENERAL BACKGROUND QUESTIONS ON PARTICIPANTS									
1. Including yourself, how many people live in your household ?	2. Sex Male1 Female0	3. Relationship to the household head 1Head 2Wife/husband 3Child 4Child in law 5Parents 6Sister/brother 7Grand mother (or father) 8Grand child 9Other relationship	4. How old If age is let equal 05 y month and should be Record 2 digits	A are you? as than or ears, both year recorded Record 2 digits MONTHS	 5. Type of registered permanent residence 1Agricultural household 2Non-agricultural household 	6. Where were you born? (City and Province)	7. Are you a "one child"? 1. No 2. Yes	 8. Where are you living in Zoucheng city? 1. Rural area 2. Urban area 	 9. Which village town are you living? 1.Qianquan.2.Gangshan.3.Fusha n.4.Yishan.5.Kanzhuang.6.Xiang cheng.7.Zhangzhuang.8.Tianhua n.9.Daisu.10.Zhongxin.11.Beisu. 12.Tangcun.13.Pingyangsi.14.Ta iping.15.Guoli.16.Shiqiang.17.Ch engqian. 18. Others (Specify)
10 . How many years have you lived in	11. Marital status	12. How many children do you have?	13. Organisation type that you and your partner working in? 1. Government agencies 2. Public organizations 3. State-owned firms 4. Collective firms 5. Private firms and hybrid firms	14.Is your husband or wife living with you in Zoucheng city?Is your husband is1. Yes2. No	15 . What is the composition of your household? (Filling in numbers)				
Zoucheng 1. Married city? 2. Divorced or separated 3. Widowed 4. Never been married(go to next section)	1. Married 2.	1. Married1. 42.ageDivorced2. 1or3. 3separated4. 43.5. 1Widowedhyt4. Never6. 1beencormarried(go7. 1to next8. 1section)9. 1					Female	Male	
	Divorced or separated				0-4 years				
	3. Widowed				5-9 years				
	4. Never been married(go		6. Foreign company 7. Village	5. Foreign invested company 7. Village		10-14 years			
	to next section)		8. Miner 9. No job, housekeeping or retired		15 years or	older			
			You	Partner (if have)	-				
SECTION B: SOCIAL CAPITAL-POSITION OF SOCIAL CAPITAL									
--	---	---	---	---	--------------------------------------	--	---	--	--
Of your relatives, friends, and acquaintances (If you knew more than one occupant of the position, we asked respondent to think of the first occupant who came to mind.)									
Occupation	 Do you know people in the position? Yes. 2. No 	2. What is his/her relationship to you? 1. Kin. 2. Friends. 3. Acquaintances. 4.	3. Did you know the person when you were looking for your present job? 1. Yes.	4. If you do not know such a person, through whom are you most likely to find him or her? 1. Kin. 2.	5. What is this person's occupation?	6. How long have you known each other?	7. Do you know each other well? 1. Very well. 2. Well. 3. Normal. 4. Not well.		
		Others (specify)	2.100	4. Others			5. Not at all.		
Elementary school teacher									
Journalist									
Administrative personnel of public or private enterprises									
Electrician									
Chief of a section									
Head of public or private enterprises									
University professor									
Worker									
Farmer									
Head of a bureau									
Lawyer									
Housemaid									
Mayor									
Provincial or city party secretary									
Party secretary of a bureau									
Party secretary of a factory									

SECTION C: INFORMATION & LEARNING Information & SOCIAL NETWORKS								
What information or knowledge did yo include information/knowledge about	our household get, and by whom in pasi irrigation and other facilities that can inc	t 24 months? (*Farming techniques crease the farming productivity.	2. Did (Have) these information positively impact (impacted) on your	3. Please explain how if the answer given on Q.2 is yes (if it is possible,				
Business: information/knowledge at	pout earn more incomes from non-farm	business. *Money management:	job or your household livelihood	please recording %changed on your				
helping people using their financial re better livelihood objectives.	sources (cash, bank deposits and othe	r financial investments) to achieve	over the past 2 years? 1Yes	income after gotten information):				
Areas	Information sources (selecting code from right column)	Sources of information access: 1. Relatives, neighbours, & friends; 2.	2No					
1. Farming*		Other individuals (Specify) 3.						
2. Credit and insurance		board; 4. Village/local markets 5.						
3. Government policies		Newspaper 6. Radio, TV, internet; 7. Business or work organisations; 8.						
4. Money management***		Agricultural extension agents; 9. Other						
5. Business**		groups or mass organizations (Specify); 10. None.						
6. Estate								
7. Education & other technics								
8. Jobs								

	SECTION D: HUMAN CAPITAL								
1. The highest edu	1. The highest education you and your family members obtained?					4. Did you attend any of the following courses in the last 24 months?			
0= None, 1=elementary school (first year) 2= elementary school (second year),3=									
elementary scho	ol (third year)	, 4= elementary sch	nool (fourth year) , 5= graduation from					
elementary school (fifth year or sixth year), 6=junior high school (first year), 7=junior high school (second school), 8=graduation from junior high school (third year), 9=senior high									
school (first yea	r), 10=senior high	school (second y	/ear) , 11=graduat	tion from senior high					
school or graduation from Technical secondary school 12=Technical/vocational certificate, 13=Diploma, 14=College/university undergraduate (Bachelors degree), 15=College /universi graduate/postgraduate, 16= PHD,17=Other (Specify).			vocational certificate, , 15=College /university						
Household	Education	Household	Education	2. How old were you when you stopped	Name		Mark X if you did in the past	t 2	Total fees for that training (if its free filling in ())
1. You	level	6	lever	full-time education?	University/College		years		
2. Your		7			Technical/vocational	training school			
partner						e			
Other members	(record	8		3.	Other training course	s			
relationship with	n household			How many years					
head):	1			have you been in paid					
3		9		age at which you	relatives etc.)	s (your friends,			
4		10		stopped full-time educations?	Taught by groups/org (local government, co etc.)	ganisations ommunist party			
5		11			Others (specify)				
5.According to y	your situation, wh	ich one is the most	acceptable	6. The main purpose for	r this training?	7. Did the job-	training programme help you	8. You	ur wage or other income sources e.g.
training fee?						to get your init	ial purpose?	stocks	s were increased by about % per
1. Below 200 y	ruan	2.20	00-400 yuan	1. Job wage	2.	1YES	2NO	month	in your
3.400-800yuar	า	4. 8	00-2000	Professional title test	3. Job			compa	any,organisation/farming or other
5. above 2000				position transition	4.				
				Change job to other of Other (Specify)	organisation 5.				

SECTION E:	ECONOMIC AC	TIVITIES-FARM	ING					
(e).1. CROP PRODU	CTION			(e).2. Fruits				
Code	Name	1.1 Over the last 12 months, did your household grow: Mark X, if the answer is yes	1.2 What was the area has been cultivated?	Code	Name	2.1 Over the last 12 months, did your household grow: Mark X, if the answer is yes	2.2 What was the area has been cultivated?	
1	Wheat			1	Watermelon			
2	Corn			2	Melon			
3	Pachyrhizus			3	Strawberry			
4	Millet			4	Apple			
5	Sorghum			5	Pear			
6	Soy bean			6	Grape			
7	Peanut			7	Peach			
8	Mung bean			8	Apricot			
9	Sesame			9	Red jujube			
10	Cotton			10	Persimmon			
11	Potato			11	Hawthorn			
12	Other root crops (Specify)			12	Kiwi fruit			
				13	Cherry			
				14	Other fruit trees (Specify)			

SECTION E: ECONOMIC ACTIVITIES-FARMING									
(e).3. Vegetable				(e).4. Livestock/aquaculture production					
Code	Name	3.1 Over the last 12 months, did your household grow: Mark X, if the answer is yes	3.2 What was the area has been cultivated?	Code	Name	4.1 Over the last 12 months, did your household grow: Mark X, if the answer is ves	4.2 Units		
1	Spinach			1	Cattle				
2	Celery			2	Cow				
3	Chinese cabbage			3	Horse				
4	Cabbage			4	Donkey				
5	Canola plant			5	Mule				
6	Leek			6	Pig				
7	Cucumber			7	Sheep				
8	Squash			8	Coat				
9	Dongguan			9	Rabbit				
10	Carrot			10	Chicken				
11	Garlic, chili, ginger & spring onion			11	Duck				
12	Eggplant			12	Goose				
13	Tomato			13	Silkworm				
14	French bean			14	Other livestock (specify)				
15	Cowpea								
16	Kidney bean								
17	Lotus root								
18	Mushroom								
19	Other vegetables (specify)								
20				e). 5.total cost from e. (1 Total net profits from e.) to e. (4) per agricultu (1) to e. (4) per agricul	ral season over the last 12 n tural season over the last 12	nonths? months?		
(e).6. Were all of the iter family over the last 12 m	ns together (from e. (1) to e nonths? 1.Yes 2.No	e.(4) the main source of inco	ome to your household or	(e). 7. Were all of the ite sufficient to maintain dat	ms together (from e. (1) to ily/monthly expenses of yo	e. (4), is the income from th ur family? 1Yes 2. N	is type of work or job No		

SECTION E: ECONOMIC ACTIVITIES-NON FARM ACTIVITIES							
Code	Name	1. What are other sources or alternative sources e.g. any business of income to your household budget in the past 12 months? (if yes, mark as "x") (excluding your farm activities, jobs)	2 Was this item the main source of income to your household or family over the last 12 months? 1.Yes 2.No	3 Is the income from this item to maintain daily/monthly expenses of your family? 1. Yes 2.No			
1	Making or selling woven basket or other artifacts						
2	Running a shop or market stall						
3	Home processed food or (repackaged) e.g. Tofu						
4	Running another business (Specify)						

SECTION E: ECONOMI	IC ACTIVITIES-MINING			
1. What is your household members' n	nonthly income?	2. What is the main activity of the	4. Is your job the main source of	5. How did (or will) you deal with job
Relationships to household head:	Monthly wages	company or organization where	income to your household or family?	loss? (circle one only)
		you work?	1. Yes. 2.No	1 Did nothing
		1=underground Mining Operator		2Sold farmland/residential land or
		assistant, 2= underground Mining		Sold other assets (motorbike, livestock,
		Operator, 3=surface operator,		furniture, durable goods)
		4=Superintendent, 5=Officer, 6-Machinist or Mining Tachnician		4. Took shildren out school
		7-transport work 8-sales work		5 Cut down expenditure
		9=logistic work e.g. security.	5. Is the income from this type of work	6Asked for help/support from
		medical staff, chef etc., 10=others	or job sufficient to maintain	relatives, friends
		(specify)	1 Ves 2 No	7Borrowed from relatives/friend/
			1. 165. 2.100	private money lender
				8Borrowed from banks
		4		9Borrowed and received support
				from government and other
				organizations (NGOs)
				a migrant to other cities 12 Borrowed
				from other sources
				13Used savings
				14Got insurance payment
				15Others (specify)

SECTION F: WORKING CONDITIONS & ACTIVITIES OUTSIDE WORKS									
1. Please tell me, using the following scale, are you exposed at work to? (Tick one only on each row)									
Tick one only on each row	All of the time	Almost all of the	Around 3/4 of the time	Around half of the time	Around 1/4 of the time	Almost never	Never		
		time							
Vibrations from hand tools,									
machinery, etc.									
Noise so loud that you would have to									
raise your voice to talk to people									
High temperatures which make you									
perspire even when not working									
Low temperatures whether indoors									
or outdoors									
Breathing in smoke, fumes (such as									
welding or exhaust fumes), powder									
or dust (such as wood dust or									
mineral dust) etc.									
Breathing in vapours such as									
solvents and thinners									
Handing or being in skin contact									
with chemical products or substances									
Radiation such as X rays, radioactive									
radiation, welding light, laser beams									
Tobacco smoke from other people									
Handing or being in direct contact									
with materials which can be									
infectious, such as waste, bodily									
fluids, laboratory materials, etc.									
2. Does your job involve?									
Tick one only on each row	All of the time	Almost all of the	Around 3/4 of the time	Around half of the time	Around 1/4 of the time	Almost never	Never		
		time							
Working at very high speed									
Working to tight deadlines									
3. Does your work affect your health, o	or not? 1. Yes 2. No		4. On the whole, are you	with working conditions in you	ır main paid job? 1. Very satisfi	ed 2. Satisfied 3. Not v	very satisfied		
			4. Not at all satisfied with v	vorking	-				
5. What kind of employment contract of	lo you have? 1. An ir	ndefinite contract. 2. A fix	ted term contract. 3. A tempora	ry employment agency contract	et. 4. An apprenticeship or other	training scheme. 5. No	o contract 6.		
Others (specify)									

SECTION G: HOUSEHOLD EXPENDITURE									
In the past 12 month, on average, what is your household expenditure on the following items per month? (Please just filling in an approximate figure on each item)									
Code	Item	Expenditure per month (RMB)	Code	Item	Expenditure per month (RMB)				
1	Food e.g. Vegetables, rice, meat, wheaten food etc.		7	Water &electricity					
2	Drink e.g. Beer, wine etc.		8	Phone and Internet charge					
3	Cigarettes		9	Housing rental fees (if you have)					
4	Education fees on children		10	Other living cost e.g. toothpaste, soap etc.					
5	Other fees on children e.g. pocket money, toys etc.		11	Other events e.g. wedding					
6	Coal, wood, sawdust, chaff, gas fees								